

# A SCHOOL MANAGEMENT SYSTEM INTEROPERABILITY MATURITY (SMSIM) MODEL FOR SCHOOLS IN SOUTH AFRICA

by

# **KENEILWE MAREMI**

(Student number: 44977883)

submitted in fulfilment of the requirements for the degree of Master of Science in the subject

Computing

at the

# UNIVERSITY OF SOUTH AFRICA

SUPERVISOR: Prof Marlien Herselman

CO SUPERVISOR: Prof Adèle Botha

Date: FEBRUARY 2022

# DEDICATION

I want to take this opportunity to dedicate this research to the following people:

My late twin sister, Dineo Maremi, I really could not have done this without you. Even through your passing, I knew you would want me to finish what I had started.

My Husband, Aden Milanzi, thank you so much my love for every sacrifice you made and for running alongside me until I reached the finish line.

Lastly, my parents, Mom: Catherine Maremi and Dad: Ben Maremi, this one is for you.

## ACKNOWLEDGEMENT

I would love to express my gratitude truly and wholeheartedly by acknowledging the following people and organizations whom, without them, this research would not have been possible:

I want to thank the almighty above, for paving the way for me, opening my mind to understand literature findings and supervisor feedback. For the wisdom and knowledge, he has granted me throughout this journey.

My Supervisor Prof Marlien Herselman, who exceeded expectations of mentorship and guidance. There is no number of words that can express how grateful I am for the blessings of working under her supervision. She became more than a supervisor in times where I felt like giving up on life and my studies. Through the COVID-19 Pandemic she was my pillar of strength to keep going and motivated me to work hard. She was sweet and kind when she needed to be and really gave me tough love when required. I found a comforter in her when I lost my twin sister and a happy place when I fell pregnant and gave birth to my son. Through the sleepless nights of trying to adjust to the new norm of the COVID-19 Pandemic and the hectic schedule of waking up every three (3) hours to feed my baby, Prof Marlien was my support system. This amazing woman's patience and attention to detail is remarkable, and I am thankful for her support. It goes without saying, I have learned a lot from her, and I am absolutely privileged to have started and completed this research under her supervision.

My Co-supervisor Prof Adele Botha, who is the strict and harsh one in the team. In my African language, there is a saying: "Le rojwa le sa le metsi" which means you sharpen and groom a person while they are still young. This saying reflects the values a parent instils in their child so they can grow to become respectable adults. This saying truly resonates with Prof Adele's role in my journey through this research. She was brutally honest and scolded me at the very beginning when I was saying words like "I can't, and I don't know". Although I shed a tear and felt extremely hurt by her scolding, it shaped my entire thinking of how I look at this research and I am extremely grateful to have had the privilege of working with her.

My husband really supported me when I put so many of our goals on hold to complete my Masters. He slept many nights and woke up without my presence, but he still understood. The father he became when I still needed to put in late hours working on the dissertation and he had to look after our son alone.

My parents encouraged me to keep going, who kept reminding me why I was doing this and always showed me how proud they were. They did so without hesitation when I needed them to take the baby for the weekend so I could make good grounds for this research.

I want to thank my friend, colleague, and sister, Tumiso Thulare, who really travelled this journey with me. She was there to listen to all my frustrations and concerns. She shared the happy achievements in this journey and helped me understand some of the comments our supervisors provided. Her support is unmeasurable, and I am thankful.

Lastly, I want to thank my late twin sister, who really was proud of me until the very last day of her life. May her beautiful soul Rest in Peace.

# DECLARATION

I, Keneilwe Maremi, declare that *A School Management Systems Interoperability Maturity (SMSIM) model for Schools in South Africa*, is my own work and that all the sources that I have used or quoted have been indicated and acknowledged by means of complete references.

I further declare that I have not previously submitted this work, or part of it, for examination at UNISA for another qualification or at any other higher education institution.

SIGNATURE

DATE

## ABSTRACT

This qualitative study applied the exploratory Delphi expert reviews as a strategy through the research onion of Saunders, Lewis and Thornhill (2016) to explore School Management Systems (SMS) and how they can be improved through interoperability. The study followed the research methodology as outlined in chapter 5 to conceptualize the School Management Systems Interoperability Maturity (SMSIM) model that provides guidelines for achieving the next level of interoperability in SMS. To conceptualize the model, literature findings, a combination of Organisational Interoperability Maturity Model (ISIMM), and experts with over 10 years of experience in various fields participated in the study. Hermeneutics and thematic analysis were used to analyze and interpret the data. After the Delphi expert review results (in two iterations), it was evident that nothing had to be changed or adjusted from the conceptual literature model therefore, the model that emanated from literature was the final model of the study.

Keywords: South African School Administrative Management System (SA-SAMS), School Management Systems (SMS), and Interoperability Maturity Models (IMM).

# LIST OF PUBLICATIONS RELATED TO THIS STUDY

Conference proceedings

- Maremi, K., Thulare, T. and Herselman, M. 2022. The benefits of digital transformation addressing the hindrances and challenges of e-government services in South Africa: A scoping review. In 2022 IST-Africa Conference (IST-Africa) (pp. 1-8). IEEE. Available at: https://ieeexplore.ieee.org/abstract/document/9845641
- Maremi, K., Herselman, M. and Botha, A. 2020. Scoping the aspects and capabilities of South African School Administration and Management Systems (SA-SAMS). In *2020 Conference on Information Communications Technology and Society (ICTAS)* (pp. 1-6). IEEE. Available at: <u>https://ieeexplore.ieee.org/document/9082454</u>
- Maremi, K., Herselman, M. and Botha, A. 2020. Interoperability Maturity Models for consideration when using School Management Systems (SMS) in South Africa: A Scoping Review. CICTE 2020: International Conference on Information and Communication Technology in Education, 03-04 September. Prague, Czechia. Available at: https://publications.waset.org/abstracts/search?q=%20interoperability
- Herselman, M.E., Botha, A., Maremi, K.J., Dlamini, S.B. and Marais, M.A. 2020. Mobile technology affecting teaching and learning in rural schools. Conference: 16th International Conference on Mobile Learning 2020. Available at: <u>https://doi.org/10.33965/ml2020\_202004L003</u>
- Botha, A., Herselman, M., Rametse, S. and Maremi, K., 2017, May. Barriers in rural technology integration: A case study from the trenches. In 2017 IST-Africa Week Conference (IST-Africa) (pp. 1-10). IEEE. Available at: <u>https://ieeexplore.ieee.org/document/8102349</u>

**TABLE OF CONTENTS** 

# TABLE OF CONTENT

DEDICATIONi
ACKNOWLEDGEMENTii
DECLARATIONiv
ABSTRACTv
LIST OF PUBLICATIONS RELATED TO THIS STUDYvi
TABLE OF CONTENT
LIST OF TABLES
LIST OF FIGURES
LIST OF ACRONYMSxx
CHAPTER 1: SCOPE OF RESEARCH1
1.1 Introduction and Background1
1.2   Problem Statement
1.3 Purpose of the Study
1.4 Main Research Question and Sub-Research Questions (SRQs)
1.5 The Objectives of the Research
1.6 The Scope of the study
1.7 Research Methodology and Design
1.8 Expected Contributions of this Study7
1.9 Limitations and Challenges of this Study7
1.10 Ethical Considerations
1.11 The Organization of the Study
1.12 Summary

СНАР	TER	2:LITERATURE REVIEW
2.1	Intr	roduction11
2.2	The	e use of Information Communication Technology (ICT)11
2.3	The	e Characteristics of Information Technology12
2.4	The	e impact of ICT on Society12
2.4	4.1	Positive Impacts
2.4	4.2	The Negative Effects Relating to the use of ICT14
2.5	The	e Role of ICT in Education14
2.5	5.1	The Role of ICT in Administration and Management of schools15
2.5	5.2	COVID-19 and Lockdown in South Africa17
2.5	5.3	Online Teaching Has Now Become a Necessity Error! Bookmark not defined.
2.5	5.4	The Benefits of Online Learning Error! Bookmark not defined.
2.5 2.5	5.4 5.5	The Benefits of Online Learning
2.5 2.5 2.5	5.4 5.5 5.6	The Benefits of Online Learning Error! Bookmark not defined. 4IR Technologies in Education
2.5 2.5 2.5 2.5 Dt	5.4 5.5 5.6 5.7 uring 1	The Benefits of Online Learning Error! Bookmark not defined. 4IR Technologies in Education
2.5 2.5 2.5 2.5 Du	5.4 5.5 5.6 11 ing 1 5.8 Th	The Benefits of Online Learning
2.5 2.5 2.5 Du 2.5 2.5 2.6	5.4 5.5 5.6 11 ring 1 5.8 Th ICT	The Benefits of Online Learning
2.5 2.5 2.5 Du 2.5 2.6 2.7	5.4 5.5 5.6 11 ring 1 5.8 Th ICT Ass	The Benefits of Online Learning Error! Bookmark not defined. 4IR Technologies in Education
2.5 2.5 2.5 Du 2.5 2.6 2.7 2.8	5.4 5.5 5.6 11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	The Benefits of Online Learning Error! Bookmark not defined. 4IR Technologies in Education
2.5 2.5 2.5 2.5 Du 2.5 2.6 2.7 2.8 2.9	5.4 5.5 5.6 5.7 11111 5.8 Th 1CT Ass Key Sur	The Benefits of Online Learning
2.5 2.5 2.5 2.5 Du 2.5 2.6 2.7 2.8 2.9 CHAP	5.4 5.5 5.6 5.7 111119 5.8 Th ICT Ass Key Sur 7TER	The Benefits of Online Learning Error! Bookmark not defined. 4IR Technologies in Education

3.2 Th	e Advantages of SMS
3.3 Th	e Advantages for Teachers
3.4 Th	e Administrative Advantages
3.5 Th	e Advantages for Learners Error! Bookmark not defined.
3.6 SA	A-SAMS
3.6.1	Background and History of SA-SAMS
3.6.2	Defining SA-SAMS40
3.6.3	The reason for the SA-SAMS design
3.6.4	Typical Setup of SA-SAMS42
3.6.5	Advantages and Disadvantages of SA-SAMS43
3.6.6	The Valistractor Software
3.6.7	Modules in SA-SAMS
3.6.8	Costs Implications of SA-SAMS
3.7 Ke	ey Components of the Chapter
3.8 Su	mmary
CHAPTER	4:INTEROPERABILITY MATURITY MODELS
4.1 Int	roduction
4.2 De	fining Interoperability
4.3 Int	eroperability Models53
4.3.1	Maturity Models53
4.3.2	Levels of Information Systems Interoperability (LISI)54
4.3.3	Organizational Interoperability Maturity Model (OIM)
4.3.4	Levels of conceptual Interoperability

4.3	3.5	Enterprise Interoperability Maturity Model (EIMM)	61
4.3	3.6	Government Interoperability Maturity Matrix (GIMM)	64
4.3	3.7	Information Systems Interoperability Maturity Model (ISIMM)	66
4.4	Co	nceptualizing The Interim School Management Systems Interoperab	ility Maturity
(SM	SIM)	Model	69
4.5	Ch	oosing OIM and ISIMM	70
4.6	Co	mbining OIM and ISIMM	71
4.7	Th 72	e Interim School Management Systems Interoperability Maturity (SM	ISIM) model
4.8	Ke	y ComponeNts of the Chapter	79
4.9	Su	nmary	
CHAF	PTER	5: RESEARCH METH	IODOLOGY
		82	
5.1	Int	roduction	
5.2	Ov	erview of the Research Design	84
5.3	Re	search Philosophy	86
5.3	3.1	Theoretical Lens	
5.4	Th	e Hermeneutic Circle	
5.5	Ap	proach to Theory Development	91
5.6	Th	e Research Approach	91
5.7	Re	search Strategies	93
5.7	7.1	Advantages of the Delphi technique	
5.2	7.2	Disadvantages of the Delphi method	
5.2	7.3	Selecting experts	

5.7	7.4	Different Delphi designs	103
5.7	7.5	Exploratory Delphi	104
5.7	7.6	Scaling techniques	106
5.8	The	e Data Collection Methods	117
5.8	3.1	Questionnaires	117
5.8	3.2	Document Analysis	118
5.8	3.3	Scoping Reviews	120
5.9	Dat	ta Analysis	121
5.9	9.1	Thematic Analysis	121
5.10	Tru	astworthiness	124
5.1	0.1	Credibility	
5.11	Eth	ical Considerations	
5.12	Sur	mmary	126
СНАР	TER	6:ASSESSMENT AND VALIDATION OF THE SMS	M MODEL
		127	
6.1	Inti	roduction	127
6.2	The	e process of conducting the evaluation	128
6.3	Exp	pert review results From iteration 1	129
6.3	3.1	Expert Review Demographic Information	129
6.3	3.2	Results of the Delphi Iteration 1	130
6.4	Usi	ing The SMSIM Model to Assess Ed-Admin	166
6.5	Exp	perts' Results in Iteration 2	167
6.5	5.1	Round 2 of the Delphi Questionnaires	167

6.6	The final School Management Systems Interoperability Maturity (SMSIM) 180	Model
6.7	Summary of the SMSIM Model Guidelines	181
6.8	Summary	184
СНАР	TER 7: CONCLUSION, REFLECTION AND RECOMMENDAT 186	IONS.
7.1	Introduction	186
7.2	Research Overview	187
7.2	2.1 Addressing the Research Questions	187
7.3	The Research Contributions	189
7.3	3.1 Theoretical Contribution	189
7.3	3.2 Practical Contribution	189
7.4	Limitations of the Research	189
7.5	Lesson Learned and Personal Reflections	190
7.6	Recommendations for Future Studies	192
7.7	Conclusion	192
REFE	RENCES	194
APPEN	NDIX A: Ethics clearance certificate	220
APPEN	NDIX B: Informed consent form	222
APPEN	NDIX C: Delphi round one information	226
Round	1 1 Questionnaire	227
Appen	ndix D: Delphi round two information	230
Round	1 2 Questionnaire	236

## LIST OF TABLES

Table 2-2 Technological drivers for the Fourth Industrial Revolution	9
Table 2-3: Summary of recommended solutions with areas of interventions to address 4IR	in
education in South Africa2	21
Table 2-4: Emerging trends in the Education sector with increasing and decreasing roles	in
selected industries	2
Table 2-5: Virtual classes during the lockdown	23
Table 2-6: Use of Social Media Apps and the Introduction of STEM lockdown digital scho	ol
(Sasol Foundation (SF) and African Teen Greeks	24
Table 2-7: Exclusive Online Learning-Remote Learning	25
Table 2-7: Essential components of the chapter	32
Table 3-1: Different SMS and their Capabilities    3	57
Table 3-2: Defining SA-SAMS   4	0
Table 3-3: Advantages and Disadvantages of SA-SAMS    4	3
Table 3-4: Essential components    4	9
Table 4-1: Defining Interoperability	52
Table 4-2: Organizational Interoperability Maturity Model (OIM)    5	;9
Table 4-3: Combining OIM & ISIMM	'1
Table 4-4: Essential components of the chapter	30
Table 5-1: Summary of principles for interpretive field research	0
Table 5-2: Studies that applied the Delphi Technique	)6
Table 5-3: Expert Reviewers    10	)2
Table 5-4: Summary of Delphi Designs (adapted from Fisher, Erasmus & Viljoen, 2020)10	)3
Table 5-5: Primary Delphi Scaling Techniques    10	)7

Table 5-6: Other Comparative Scaling Techniques	
Table 5-7: Processes of data analysis in thematic analysis (Braun & Cl	arke, 2019; Kiger &
Varpio, 2020)	
Table 5-8: Themes	
Table 6-1: Iteration 1: Descriptive statistics	
Table 6-2: Iteration 2: Descriptive statistics	

## LIST OF FIGURES

Figure 1-1: Adopted Research Onion ((Saunders et al., 2016))7
Figure 1-2: Organisation of the dissertation
Figure 2-1: Research process of the study highlighting the phases used to develop the SMSIM model
Figure 2-2: Outline of Dissertation
Figure 3-1: The research process highlighting Chapter 3 as part of Phase 1 of the study34
Figure 3-2: The architecture of SA-SAMS (Adapted from Muriithi and Masinde, 2016)42
Figure 3-3: SA-SAMS Modules46
Figure 3-4: The research process that highlights how Chapter 3 contributes to the theoretical framework
Figure 4-1: The research process of the study highlighting that Chapter 4 is part of Phase 1 of the study
Figure 4-2: The LISI Model (Ostadzadeh, Shams & Badie, 2015; Gürdür & Asplund, 2018)
Figure 4-3: Enterprise Interoperability Maturity Model (EIMM) (Ostadzadeh et al., 2015)61
Figure 4-4: Government Interoperability Maturity Model (GIMM)64
Figure 4-5: Information Systems Interoperability Maturity Model (ISIMM)(Rezaei et al., 2014)
Figure 4-6: information systems' interoperability maturity transition (Campos et al., 2013, Van Staden and Mbale, 2012)
Figure 4-7: Conceptualizing the School Management Systems Interoperability Maturity (SMSIM) model
Figure 4-8: School Management Systems Interoperability Maturity (SMSIM) model74
Figure 4-9: Refined Interim SMSIM model

Figure 4-10: Chapter overview
Figure 5-1: Research Process to conduct the research
Figure 5-2:Research Process to conduct the research
Figure 5-3: Adapted Research Onion applicable to this study ((Saunders et al., 2016))86
Figure 5-4: A Slider Scale
Figure 5-5: The Five-point Likert scale as an ordinal scale114
Figure 5-6: The Four-point ordinal Likert scale with undecided as a separate option114
Figure 5-7: Delphi Method 1st iteration (John-Matthews et al., 2017; Mukherjee et al., 2014)
Figure 5-8: Delphi Method 2nd iteration (John-Matthews et al., 2017; Mukherjee et al., 2014)
Figure 5-9: Most frequently used words across the selected literature
Figure 6-1: The research process and how Chapter 6 covers phases 2 and 3
Figure 6-2: Experts who participated in the study130
Figure 6-3: Q1 Delphi Round 1 Questionnaire131
Figure 6-4: Q2 Delphi Round 1 Questionnaire132
Figure 6-5: Q3 Delphi Round 1 Questionnaire133
Figure 6-6: Q4 Delphi Round 1 Questionnaire134
Figure 6-7: Q5 Delphi Round 1 Questionnaire
Figure 6-8: Q6 Delphi Round 1 Questionnaire136
Figure 6-9: Q7 Delphi Round 1 Questionnaire
Figure 6-10: Q8 Delphi Round 1 Questionnaire139
Figure 6-11: Q9 Delphi Round 1 Questionnaire140

Figure 6-12: Q10 Delphi Round 1 Questionnaire	41
Figure 6-13: Q11 Delphi Round 1 Questionnaire14	42
Figure 6-14: Q12 Delphi Round 1 Questionnaire14	43
Figure 6-15: Q13 Delphi Round 1 Questionnaire14	44
Figure 6-16: Q14 Delphi Round 1 Questionnaire14	45
Figure 6-17: Q15 Delphi Round 1 Questionnaire14	46
Figure 6-18: Q16 Delphi Round 1 Questionnaire14	47
Figure 6-19: Q17 Delphi Round 1 Questionnaire14	48
Figure 6-20: Q18 Delphi Round 1 Questionnaire	49
Figure 6-21: Q19 Delphi Round 1 Questionnaire	50
Figure 6-22: Q20 Delphi Round 1 Questionnaire	51
Figure 6-23: Q21 Delphi Round 1 Questionnaire	52
Figure 6-24: Q22 Delphi Round 1 Questionnaire	53
Figure 6-25: Q23 Delphi Round 1 Questionnaire	54
Figure 6-26: Q24 Delphi Round 1 Questionnaire	55
Figure 6-27: Q25 Delphi Round 1 Questionnaire	56
Figure 6-28: Modules of Ed-Admin16	51
Figure 6-29: Why choose Ed-Admin	52
Figure 6-29: Ed-Admin represents other systems and applications – therefore it was selected	ed
	52
Figure 6-30: System Integrating with Ed-Admin10	53
Figure 6-31: Systems Integrating with Ed-Admin16	54
Figure 6-32: Integration	55

Figure 6-33: Level of interoperability in Ed-Admin167
After the first round of Exploratory Delphi questionnaires a report was prepared and anonymously provided to the experts to reflect and change or adjust their initial responses if required and from this report, all experts did not need to change their initial responses, instead
that round 2 that evoluate the SMSIM model is required. The report is attached in appendix E
that found 2 that evaluate the SMISHVI model is required. The report is attached in appendix E.
Figure 6-34: Q1 Delphi Round 2 Questionnaire
Figure 6-35: Q2 Delphi Round 2 Questionnaire169
Figure 6-36: Q3 Delphi Round 2 Questionnaire
Figure 6-37: Q4 Delphi Round 2 Questionnaire171
Figure 6-38: Q5 Delphi Round 2 Questionnaire
Figure 6-39: Q6 Delphi Round 2 Questionnaire
Figure 6-40: Q7 Delphi Round 2 Questionnaire
Figure 6-41: Q8 Delphi Round 2 Questionnaire
Figure 6-42: Q9 Delphi Round 2 Questionnaire
Figure 6-43: Q10 Delphi Round 2 Questionnaire
Figure 6-44: School Management Systems Interoperability Maturity (SMSIM) model 180
Figure 7-1: The research process and how this chapter supports Phase 3 of the study

# LIST OF ACRONYMS

Acronym	Description
ANA	Annual National Assessment
CD	Compact Disc
C4ISR	Command, Control, Communications, Computers, Intelligence, Surveillance, and
	Reconnaissance
COVID-19	Coronavirus pandemic
DBE	Department of Basic Education
DCDT	Department of Communication and Digital Technologies
EIMM	Enterprise Interoperability Maturity Model
FOSS	Free and Open Source Software
4IR	Fourth Industrial Revolution
GIMM	Government Interoperability Maturity Matrix
ICT	Information Comunication Technologies
IMM	Interoperability Maturity Model
ISIMM	Information Systems Interoperability Maturity Model
IQMS	Integrated Quality Management System
LISI	Levels of Information Systems Interoperability Model
MOE	Ministry of Education
OIM	Organizational Interoperability Maturity Model
SA-SAMS	South African School and Administration Management System
SGB	School Governing Body
SITA	State Information Technology Agency
SMS	School Management Systems
SMSIM	School Management Systems Interoperability Maturity
STEAMIE	Science, Technology, Engineering, Arts, Mathematics, Innovation, and
	Entrepreneurship
STEM	Science, Technology, Engineering and Mathematics
TV	Television
DDD	Data Driven Districts
ATM	Automatic Teller Machine

## CHAPTER 1: SCOPE OF RESEARCH

#### 1.1 INTRODUCTION AND BACKGROUND

School Management Systems (SMS) are large databases that schools utilize to automate their day-to-day administrative operations (Nureni, 2014; Sarker, 2016; Akter, 2018; Ssempala, Kabuunga, Taremwa & Magero, 2018). Multiple users widely utilise SMS, allowing them to be configured independently to address the needs of each school (Sarker, 2016; Akter, 2018). Using SMS assist schools with having to deal with tedious paperwork, they can be easily used, and they work accurately and seamlessly in various scenarios. As a result, these systems save time, increase school management efficiency, and reduce workloads (Sarker, 2016; Akter, 2018).

SMS have gained prominence in recent years as a powerful tool for e-leadership and datadriven decision-making. Teachers, learners, and parents use SMS to communicate and exchange information (Blau & Presser, 2013). As a result, these systems allow schools to easily manage their information, which reduces teachers' workload and creates a centralized database that can be accessed and used by all schools (Wei, Piaw, Kannan & Moulod, 2016). Specifically, the system automates two key functions: academic administration and school administration (Wei *et al.*, 2016).

There are many different SMS that are used in many schools around the world, to mention a few (Arslan, 2013; Al-Hayek, 2016; Kunda & Chama, 2016; Saquin & Marcial, 2016):

- Fedena is used by more than 200 countries, including India, Somalia, Malaysia, Egypt, Saudi Arabia, Nigeria, and many others, and has been implemented in more than 40 000 schools, of which 15 000 are under the Department of Basic Education (DBE) of the Government of Kerala
- South Carolina uses the OPENSIS School Management System to maintain learners' reports, attendance, health records, and practice reports.
- OpenAdmin is a Free and Open-Source Software (FOSS) school management system used in Canada that provides sites for teachers, parents, and administrators.
- E-School, a virtually independent platform that allows any user to access the system from any location through the Internet, was developed in India (Al-Hayek, 2016) and others (for example, SchoolTool, Fekara, OPUS and My School).

SA-SAMS (South African School and Administration Management System) is used by • most schools in South Africa as mandated by the DBE (Gxwati, 2011; Irene, 2013). The smooth operation of a school depends on effective administration and management. To make the administration of schools more effective, the DBE developed SA-SAMS to solve management and administration challenges and streamline data in schools (Molepo, 2015; Muriithi & Masinde, 2016). The DBE has made SA-SAMS freely available to all schools in order to ensure that data is submitted in the correct format (Kuriakose, 2014). There are several forms of financial documentation that schools need to keep on a regular basis: general ledgers, trial balances, cash books, petty cash journals, and bank reconciliations (Hlongoane, 2019). Public schools should align their financial records with SA-SAMS requirements. For this reason, all schools must use SA-SAMS to ensure consistently accurate and complete accounting records (Hlongoane, 2019). SA-SAMS was chosen as the single reporting source in 2008 by the Council of Education Ministers (MEC) representing all provinces (Irene, 2013; Kuriakose, 2014). This decision did not enforce the use of SA-SAMS in schools but outlined how the data's reporting format or structure should be presented to the DBE (Malan, 2014; Van Der Berg, Wills, Selkirk, Adams & Van Wyk, 2019). Each SMS used by schools should easily integrate with SA-SAMS to reduce duplication (Maponya, 2015; Hinds, 2016).

Unfortunately, there has been a lack of technical support and training in some provinces, which has led to SA-SAMS not being adopted as quickly as expected by the public schools (Muriithi & Masinde, 2016). Support requests for schools are often ignored for months due to a lack of support personnel. As a result of these frustrations, schools often abandon SA-SAMS and revert to manual operations (Muriithi & Masinde, 2016). Another challenge is the interoperability of various other types of SMS with SA-SAMS. Through interoperability, data exchange is significantly automated, reducing errors and delivering information to the point of use, enabling more timely actions and a flow of information that can be used effectively in the decision making processes (Cooper, 2014).

The implementation of interoperability can have a great deal of benefits for SMS. An interoperable system allows for continuous information exchange, for example, exchanging data with other systems for performance improvements, creating services, and controlling the processing of information systems and operations (Nsaghurwe, Dwivedi, Ndesanjo, Bamsi,

Busiga, Nyella, Massawe, Smith, Onyejekwe, Metzger & Taylor, 2021). Interoperability can be measured using Interoperability Maturity Models (Awasthy, Flint, Jones & Sankaranarayana, 2018). The maturity model assesses the extent to which key activities and processes are identified, practised, managed, and implemented effectively (Awasthy *et al.*, 2018). Several types of Interoperability Maturity Models (IMM) exist such as the Levels of Information Systems Interoperability Model (LISI), Organizational Interoperability Maturity Model (OIM), Enterprise Interoperability Maturity Model, Information Systems Interoperability Model (ISIMM), and Government Interoperability Maturity Matrix Model (GIMM). From these models, a combination of OIM and ISIMM will be used to conceptualize a School Management Systems Interoperability Maturity (SMSIM) model to address the purpose of this study, which will be used to measure the level of Interoperability in SMS and provide a guideline that can be used to improve SMS. Section 4.5 discuss the reasons these two models were chosen.

An education system's effectiveness depends on how well its information system supports sharing, storing, categorizing, and using information (Jakimoski, 2016). To determine how interoperability can best benefit SMS, it was necessary to evaluate the interoperability level in SMS and ultimately enhance the system by providing guidelines for advancing to the next level on interoperability in SMS, with the aim of potentially addressing the disadvantages of SMS and ensuring that all SMS can share relevant information according to SA-SAMS requirements.

#### **1.2 PROBLEM STATEMENT**

The problem statement of this study lies in the disadvantages of SMS, although SMS offer many advantages to schools, it also comes with many disadvantages such as (Al-Hayek, 2016; Muriithi & Masinde, 2016):

- increased workloads for teachers,
- duplication of work,
- redundancy and errors,
- for schools without internet access there is often delays for the submission of reports to the DBE by schools as the submission is done manually through a CD or mailing system,

- most schools use a free SMS without features like discipline, custom reporting, registration, or inventory. There are some that are narrowed due to space capacity, the number of learners, and storage space,
- often, SMS are installed on a standalone computer, limiting access to the system, which
  results in tension and stress for administrators working alone during peak demand
  times, such as when schools must process learner reports at the end of the term, causing
  delays,
- despite limited access to the system, teachers spend much of their time manually validating learner's marks on paper that is never-ending, which increases the possibility of mistakes,
- the lack of skilled Information Communication Technology (ICT) personnel who can utilize the system. On the government's side, there is an insufficient supply of relevant software, a lack of maintenance culture, as well as an insufficient level of security and purity regarding data,
- can only be accessed within the school's intranet,
- limit learners and parents or guardians from accessing the learners' information.

Despite the DBE's efforts to introduce SMS to simplify the administration in all schools, SMS still lacks in some way or the other as schools without internet access still cannot submit data to the DBE through a click of a button. Furthermore, SMSs are also not communicating nor sharing information, thus lacking interoperability. As a result, schools work in silos, necessitating a SMSIM model that can be used as a guideline that assesses the level of interoperability to improve and support schools with their SMS. The sharing of information must be incompliance to the PoPI Act discussed in chapter 3 of this study. Interoperability has educational benefits that can help address the above-mentioned disadvantages of SMS. As these advantages include, according to Cooper (2014):

- Reducing the burden on staff members of the school to enter data: instead of registering a learner multiple times for the curriculum, the learner can be entered once into the system. This educational benefit can benefit teachers who waste teaching time by repeatedly enter data into the system.
- **Improving data quality:** interoperability improves data quality, as information flows efficiently, and there is a timely action caused by the delivery of information to the point of use, significantly reducing duplication and human errors.

- **Promoting adaptability and promote data sharing:** this means, schools can easily transfer data to the DBE.
- **Supporting data-driven decision making**: accurate and timely data is essential for good decisions.
- Enhancing efficiency and timeliness: Information can be delivered to the point of use more rapidly, resulting in more efficient and effective information flow.

These advantages of interoperability can enhance SMS to ensure an improved overall information flow for effective decision making and management of schools.

The SMSIM model will provide guidelines to assess the level of interoperability in SMS that may help improve the disadvantages of SMS. Refer to chapter 4 on how the disadvantages of SMS may be addressed using the SMSIM model.

## **1.3 PURPOSE OF THE STUDY**

The way interoperability can enhance SMS was provided above; therefore, the purpose of the study was to conceptualize a SMSIM model for schools in South Africa. This conceptualized SMSIM model applied a combination of OIM and ISIMM model to assess the level of interoperability in SMS and provided clear guidelines on how SMS can advance to the next level of interoperability to improve. This model may enhance the level of interoperability maturity in SMS and highlight the value of such a model for existing SMS in schools. The SMSIM model is envisaged to enable schools in SA to first measure their maturity levels. Secondly, to acknowledge the growth gained in their maturity levels to realize the potential future of how the systems can improve and evolve and link with SA-SAMS.

## 1.4 MAIN RESEARCH QUESTION AND SUB-RESEARCH QUESTIONS (SRQS)

The following research questions address the purpose of this study:

**Main Research Question:** How can a SMSIM model assist schools in improving the interoperability and maturity levels of their current SMS?

- **SRQ1:** What constitutes the components of a model of interoperability maturity for Schools Management Systems?
- **SRQ 2**: What existing models of interoperability maturity can be used to conceptualize a SMSIM model?

• SRQ 3: What value will a SMSIM model have for schools using Schools Management Systems?

## 1.5 THE OBJECTIVES OF THE RESEARCH

To address the purpose of this study (the conceptualization of the SMISM model), the following objectives apply to:

- investigate the existing models of Interoperability Maturity (IM) to determine the levels and components of a model to support interoperability in Schools Management Systems (SMS)
- identify guidelines for achieving interoperability and adapt these for Schools Management Systems
- determine what components the SMISM model should consider and
- identify the value that the SMSIM model can have to support schools with their own SMS.

## **1.6 THE SCOPE OF THE STUDY**

This will focus on interoperability maturity models and how these can benefit SMS to become interoperable with other SMSs to ensure better data integrity. Better data integrity can influence decision making as accurate data can be applied by decision-makers.

## 1.7 RESEARCH METHODOLOGY AND DESIGN

As this study aimed to conceptualize a SMSIM model, the methodology applied the explorative qualitative design.

The study is grounded in interpretivism as a research philosophy that will follow the research onion process as suggested by Saunders, Lewis and Thornhill (2016:p207). Furthermore, hermeneutic principles was used during the analysis of the data to translate and interpret the text, verbal, and nonverbal communication (Hiller, 2016:p22; Saunders *et al.*, 2016:p182). During the qualitative data collection phase, inductive reasoning was also used to identify essential themes and patterns in the collected data (Leedy & Ormrod, 2021). Various experts with several years of experience in various fields of expertise were consulted to collect data through an explorative Delphi method. The exploratory Delphi design allows data to be collected, the model to be assessed and validated without physical contact, which is more

appropriate for this study since participants have been almost impossible to reach following the COVID-19 pandemic. The following adapted research onion is applied in this study as displayed below:



Figure 1-1: Adopted Research Onion ((Saunders et al., 2016))

The research methodology is discussed in more detail in Chapter 5.

#### 1.8 EXPECTED CONTRIBUTIONS OF THIS STUDY

The study is expected to contribute to School Management Systems (SMS) by conceptualizing an Interoperability Maturity Model that can assist in improving interoperable SMSs in schools. A SMSIM model can contribute to literature as there is currently no literature focusing on the interoperability of SMS in schools. SMS is currently not linked with Interoperability Maturity Models. This results in a gap between the two fields of study, and this will open a platform of interoperability in schools and its management systems to simplify management of schools. Section 7.3 provides the detail of the final contributions of the study.

## 1.9 LIMITATIONS AND CHALLENGES OF THIS STUDY

The limitations of this study are as follows:

- This is an exploratory qualitative study with exploratory Delphi expert reviews that are limited to only experts with over 10 years of expertise in the field of ICT in education, Interoperability, School Management Systems (SMS), ICT4D and Interoperability.
- Furthermore, this study is also limited to SMS that are different from SA-SAMS. The importance of this is to get a clear view of how SA-SAMS and other open and free versions of SMS can be improved from the learnings of the paid versions of SMS. Only schools utilizing a paid version of SMS will be considered for the study because they have a longer period of using and engaging with customized paid-for internationally obtained SMS before the SA-SAMS was developed. They also have an improved knowledge of their own SMS (in many more facets apart from only data), and they do not rely on a free version of SMS.

## 1.10 ETHICAL CONSIDERATIONS

University of South Africa (UNISA) ethics guidelines were followed for this study to protect the rights of participants and ensure the integrity of the research. Appendix A provides the ethical clearance certificate.

The research adhered to the following:

- The confidentiality and privacy of participants.
- Informed consent.
- Good Research.
- Accountability and voluntary participation.
- Detailed information about ethics can be found in Section 5.11 of Chapter 5.

#### 1.11 THE ORGANIZATION OF THE STUDY

Figure 1-2 shows the organization of the dissertation.

An introduction to the study is provided in Chapter 1 as well as the problem statement, purpose, research questions, scope of the study, and expected contributions and limitations.

Chapter 2 covers the literature review of the study, and it conceptualizes the literature review into two separate chapters, namely Chapters 3, and 4. Chapter 2 discusses ICT and the role of ICT in education, Chapter 3 discusses SMS, and Chapter 4 discusses interoperability as well as interoperability maturity models.

Chapter 5 covers the study's research methodology, which consists of research design and design data collection protocol.

Chapter 6 presents the results of the Delphi questionnaire and the SMSIM model is verified and validated, after analyzing Delphi expert review results.

Chapter 7 draws a conclusion and provides recommendations for future studies. It also answers the MRQ and SRQs and provides a personal reflection.

The following figure summarizes the research process and the structure of the chapters with the associated outcomes:



Figure 1-2: Organisation of the dissertation

#### 1.12 SUMMARY

The foundation of the research is laid in this chapter. This chapter introduced the concept of School Management systems (SMS) and discussed the problem statement of this study. The scope was defined, and the research was justified. A general research methodology was outlined (see Figure 1-1). The research methodology and process followed have been provided in Figure 1-1 and the organization of the dissertation has been depicted in Figure 1-2. Chapter 2 will discuss the literature review; this chapter provides insight into ICT and the impact it has

not only on society but also education. Although it will provide a high-level overview of ICT and its impact on society, this chapter will form the basis of other literature studies to follow.

## CHAPTER 2: LITERATURE REVIEW

#### 2.1 INTRODUCTION

As per the research process below, this chapter forms part of phase 1 and this chapter aims to highlight the importance of ICT in the education sector and the importance of interoperability when using these ICTs. Firstly, this chapter will provide an overview of what ICT is, then scope it to it's purpose in society and in general. The role of ICT in education will be provided, and then how ICT interoperability can be perceived and the positive and negative effects of ICT interoperability when one assesses it, will be highlighted. It further discusses the COVID-19 pandemic and how online learning assisted society using 4IR technologies. This is very important for the study as it is vital to know the overall impact of ICT in society and the education sector to conceptualize a model that considers all the impacted areas worldwide.



Figure 2-1: Research process of the study highlighting the phases used to develop the SMSIM model

#### 2.2 THE USE OF INFORMATION COMMUNICATION TECHNOLOGY (ICT)

Information and communications technology, also known as ICT, is a common extension of the term "information technology" (IT). ICT integrated communication (wireless and mobile signals), security, audio-visual systems, computers, and middleware allowing business users to

store, send, access, and use information (Ratheeswari, 2018; Hatakka, Thapa & Sæbø, 2020; Suleiman, Zakari, Sani & Ukashatu, 2020). An ICT system enables audio-visual and computerbased communication networks to be linked via a single cable or channel (Nureni, 2014; Ratheeswari, 2020; Suleiman *et al.*, 2020; Adarkwah, 2021).

Based on the description provided above, ICT enables communication and allows people to stay connected. Organisations can do a lot with the power of technology; hence, this study focused on improving the system that is implemented in schools for the daily management and administration of the school. The section below provides the characteristics of information technology.

## 2.3 THE CHARACTERISTICS OF INFORMATION TECHNOLOGY

The characteristics of Information Technology, according to Ratheeswari (2020), involve:

- Acquiring, storing, distributing, managing, transporting, or obtaining data or information.
- Access to real-time information.
- Easy availability of updated data.
- Providing different types of communication media.

Based on the above characteristics, it is evident why the integration of ICT into education has been considered the primary key to human progress (Veerasamy & De Souza-Daw, 2012) as all these characteristics can support teaching and learning. However, before discussing ICT in education, it is vital to first understand the positive and negative impacts relating to the use of ICT in society. This knowledge assisted as considerations when conceptualizing the School Management Systems Interoperability Maturity (SMSIM) model. These will be discussed in the following two subsections below.

#### 2.4 THE IMPACT OF ICT ON SOCIETY

ICT has an impact on modern society, both positively and negatively. Following are some of these (Nureni, 2014; Fraillon, Ainley, Schulz, Friedman & Duckworth, 2020:5; Imam & Itodo, 2020):

#### 2.4.1 **Positive Impacts**

The following positive impacts of ICT in society are provided by Nureni (2014); Bhatti, Iqbal, Noreen, Mukhtar and Javed (2021):

- **Faster communication speed:** Prior to the Internet, news used to be published by news providers through newspapers and presented through the Television and Radio only. Anyone can now receive information or news efficiently via the Internet, including business partners, friends, and family members. Due to advancements in Internet connection speed, communication has become fast and inexpensive.
- Lower communication cost: It is cheaper and more efficient to communicate using the Internet over other methods, such as courier services, mailing through the post office, or phone calls. Internet access remains a challenge in developing countries as compared to developed countries.
- **Reliable mode of communication:** A computer is a reliable tool. With the Internet, information can be retrieved and accessed from any location. As a result, it is a reliable method of communication.
- Effective sharing of information: The advancement of ICT has made it possible for people worldwide to exchange news and information on mailing lists, Internet forums and discussion groups.
- **Paperless environment:** A paper-free environment has been made possible by ICT. Rather than using paper, digital mediums can store and retrieve information. A paperless environment can also be created through online communication, such as emails, online chats, and instant messaging.
- **Borderless communication:** Data can be retrieved instantly, connected, accessible, and a variety of functions can be performed on the Internet. Services and information are now available without regard for geographical boundaries. Through the Internet, communications and information can be exchanged without borders.
- **Create employment:** Even though many employment fields have lost jobs (especially now with COVID-19 spreading), other employment sectors (such as the Healthcare sectors) experienced growth and could create jobs. In businesses, it is necessary to maintain all computers: fix hardware, install software, etc.

Nureni (2014) noted that ICT has proved to be a positive influence on society in reducing manual work and human error, thus schools have implemented ICTs (hardware and software) also reduce manual work and human error thus improving the quality of information. The following section presents the negative effect of using ICT.

#### 2.4.2 The Negative Effects Relating to the use of ICT

The following adverse effects of using ICT are also recorded, according to Nureni (2014):

- **Individualistic and introverted**: Today, people generally prefer online communication over in-person interactions. People are more likely to become introverts and individualistic.
- Health problems: Using computers for prolonged periods can harm users as using computers could lead to mental stress, eyestrain, physical stress. Health problems can be solved with ergonomic environments. An ergonomic chair can reduce back pain and screen filters can reduce eye strain.
- Unemployment: Some jobs are lost because computers replaced humans, such as in manufacturing, where fully automated fields are now found in many factories. Rather than employing human labour, computer-controlled robots build things. Although robots are expensive to purchase, they save the industry money over time. In addition, robots can operate 24 hours a day without taking a break. Bank Tellers: At first banks used to have a lot of Tellers for depositing and withdrawing money, now tellers are reduced to a few due to the introduction of Automatic Teller Machines (ATMs).

Therefore, it is evident from the above positive and negative impacts that these should be considered when conceptualizing the SMSIM model. The next section will further outline the role ICT plays in education as this study's focussed domain.

#### 2.5 THE ROLE OF ICT IN EDUCATION

ICT has become integral to education worldwide (Bindu, 2016; Arshad, 2020). Educational institutions have increasingly adopted ICT as a part of their environments. The use of ICT in education will become a regular aspect of society, along with technology tools (Arshad, 2020; Ratheeswari, 2020). An increase in the number of students, teachers, and educational institutions influences teaching and learning (Hernandez, 2017). Therefore, ICT is essential in distributing education (Bindu, 2016). The following are the roles that ICT plays in education,

ICT (Sarkar, 2012; Olaore, 2014; Amua-Sekyi & Asare, 2016; Bhattacharjee & Deb, 2016; Ratheeswari, 2018; Agrawal & Alharbe, 2019; Das, 2019; Del Carmen Trillo-Luque, Vilches-Vilela, Quintero-Ordoñez, Fuentes-Gómez & Dauder, 2020:p251; Hatakka *et al.*, 2020; Imam & Itodo, 2020; Khan, 2020; Rana & Rana, 2020; Bhatti *et al.*, 2021):

- Provides educational opportunities such a tablet that can be used to for teaching and learning.
- Assists universities with their education, research, and public service missions.
- Provides a new communication channel between students, teachers, academic staff, and the community.
- Provides access to the syllabus, A student can access the learning program at any time, and they can do so from anywhere.
- Can be utilized to leverage distance learning to bring education to the doorsteps of children living in remote rural areas.
- It helps to stay updated on the latest developments by using various technologies.
- The ICT curriculum enables students to understand the nature of technology, how they can use and manipulate technology, and how ICT impacts individuals and communities.

It is evident that ICT positively impacts the education sector and there are many benefits in utilizing it. Section 2.5.1 below focuses on the role of ICT in administration and management which is the focus of the study. It is essential to understand the role of ICT for administration and management purposes in a school, to conceptualize a model that will be useful and add significant value to the use of SMS in the South African schools.

#### 2.5.1 The Role of ICT in Administration and Management of schools

Malan (2014) posits that the use of the Internet, landlines, and computers in education management and administration have increased in recent years to support school administration's sustainable development. School administrators use ICT administration and management software for various administrative tasks, from data storage to information management (Malan, 2014).
Three main administration groups use ICT daily for administration and management namely (Lin, Xie, Jeng & Wang, 2011; Mwalongo, 2011; Seyal, 2012; Ghavifekr, Afshari, Siraj & Seger, 2013; Marinov & Tsankova, 2015; Qureshi & Qazi Abro, 2016):

- Administration heads: Seyal (2012) states that, as administration heads, principals must have the necessary skills to use ICT to perform management and administrative tasks. In both administrative and managerial activities, principals can act as role models regarding information technology because they provide an example that others can follow and are available when needed for help.
- Administration teachers: Have administrative responsibilities in addition to their teaching responsibilities such as overseeing all departments and offices.
- Administration staff: Administrative employees use ICT to manage their daily tasks. Organizing finances, communicating, maintaining records, processing paperwork, and collecting information for SMS are some of the tasks that administrators do in schools.

When applying ICT to perform administrative tasks, an improvement is detected in the efficiency of the administration and management at every level of the school, from the classroom to the library and throughout the school (Das, 2019). ICT plays a vital role in making administration in a school less burdensome. ICT also offers possibilities that improve and develop administration and strengthen the capability to handle administrative work more effectively (Das, 2019).

The support that ICT can provide to education, in the end, should not be underestimated, as it influences the administration and management of the school. Access to information from various sources is now a possibility. The new opportunities that the fourth industrial revolution (4IR) can provide will also play a role in improving the education sector. The digital technologies which were used in section 2.5.7 during the COVID-19 pandemic are a pure demonstration of the impact technology has to ensuring educational continuity. Teachers and learners accessed new information online during this time than ever before, as this was now the *new normal* for both teachers and learners (Mhlanga & Moloi, 2020).

The following section describes the COVID-19 pandemic and lockdown in South Africa and other parts of the world. Discussing the pandemic and lockdown is essential as this will provide a clear view of the role digital technologies played in ensuring continuity in education.

### 2.5.2 COVID-19 and Lockdown in South Africa

Sansa (2020) claims that COVID-19 is a novel coronavirus that infects people with severe respiratory illness (SARS-COV-2). The virus first surfaced in China in late 2019 in Wuhan City, and researchers have connected it to a condition known as COVID-19 (Adnan & Anwar, 2020; Arshad, 2020; Sansa, 2020; Durizzo, Asiedu, Van Der Merwe, Van Niekerk & Günther, 2021; Gwenzi & Rzymski, 2021).

COVID-19 emerged at a time when there was a global learning crisis; the High Levels of Learning Poverty were intensified by the Pandemic (Worldbank, 2020). More than 160 countries had to close their schools due to the COVID-19 pandemic. According to the World Bank, 1.6 billion children and youth worldwide were excluded from school due to school closures (Arshad, 2020; Mhlanga & Moloi, 2020; Worldbank, 2020; Durizzo *et al.*, 2021).

The government of South Africa enacted a national lockdown (26 of March 2020), which meant that all educational institutions would be closed. This had an adverse effect on learning. In addition, school closings may have resulted in a loss of skilled human resources and education, which could affect economic growth (Mhlanga & Moloi, 2020; Durizzo *et al.*, 2021).

Due to the COVID-19 pandemic, many companies closed or reconsidered their use of digital technologies (Arshad, 2020). There was no preparation for the lockdown, so universities and public schools without 3IR technology were forced to close. On the continent of Africa, this was a common occurrence due to challenges in infrastructure, protocols, and data costs. The lockdown affected many South African public schools, especially in townships and rural areas, as they were under-resourced and could not provide their learners with ICT access to teaching content. This resulted from the challenging infrastructure in rural areas with no bandwidth or stable Internet access to support learning (Dube, 2020; Mhlanga & Moloi, 2020).

To reduce the size of the academic barrier, some educational institutions transferred some courses to their online platforms (Adnan & Anwar, 2020; Alabbadi & Al-Masaeed, 2020). Primary, secondary, and tertiary education in South Africa had to face the reality of providing only online teaching to their learners during the outbreak of COVID-19 (Mhlanga & Moloi, 2020; Ożadowicz, 2020; Zhong, 2020).

According to Mhlanga and Moloi (2020), the education sector will find it difficult to return to old-school teaching methods in the aftermath of the pandemic, as social distancing will

continue to be an issue to prevent the spread of this virus. When the reality of lockdown struck, education sector corridors were occupied with discussions about online learning and using television and radio for revision. This pandemic motivated the education sector to embrace digital transformation during the lockdown (Alabbadi & Al-Masaeed, 2020; König, Jäger-Biela & Glutsch, 2020; Mhlanga & Moloi, 2020).

It is evident from this section that digital technologies (see section 2.5.5) played a considerable role in the country's functionality during the lockdown. As a result, online teaching and learning became necessary to ensure that learners do not lose out on a whole year of schooling. Sadly, pandemics are natural disasters that occurs without notice, and their effects may not be known until it is too late; however, the use of digital technologies mitigated the impact of the COVID-19 pandemic. In the aftermath of COVID-19, digital technologies were used to maintain school functionality, as many schools had to close and reopen using these technologies (Television, Radio, Social Media platforms such as YouTube). The next section provides an overview of what 4IR is and how it will benefits society. It also explains the technologies that were used in South African Schools to ensure educational continuity during the COVID 19 pandemic.

### 2.5.3 4IR Technologies in Education

4IR enables individuals to move and connect digital domains through technology to enhance and manage their lives (Kayembe & Nel, 2019). According to Schwab (2016a), 4IR includes genomics, nanotechnology, design and quantum computing, therefore the new revolution goes well beyond the use of computers and technology. The 4IR combines physical, digital, and viral interactions that make it different from previous models (Schwab, 2016a).

Davis (2016) defines 4IR as the advent of cyber-physical systems that allow humans and machines to interact in new ways. Despite these capabilities being grounded in the infrastructure and science of the third industrial revolution (3IR), Davis (2016) further argues that the fourth industrial revolution represents a completely new method of integrating social science with society and even the human body (Davis, 2016; Shava & Hofisi, 2017).

The term "4IR" refers to the fusion of technologies that causes the distinctions between the physical, digital, and biological worlds to become hazier (Davis, 2016; Global, 2018). 4IR impacts almost every aspect of our daily lives, changing how people relate to technology and how and where they work (Global, 2018; Schwab, 2019).

All the definitions of 4IR provided by the four authors Das (2019), Shava and Hofisi (2017), Schwab (2016a), and Kayembe and Nel (2019) are appropriate for this study as they all broaden the definition of 4IR from the technology to the people using it. Section 2.5.5 discuss the technologies that were used to ensure educational continuity and combat the spread of the COVID 19 pandemic during lockdown. The authors identify these technologies as the 4IR technologies, however, literature findings has identified the confusion regarding the understanding of 4IR. Li, Hou and Wu (2017) argue that, the understanding of Cyber Physical Systems (CPSs) causes some confusion about the essence of the 4IR. Furthermore, he states that the developments of digital, physical, and biological technologies are three fundamental technological drivers of the Fourth Industrial Revolution. These three technological drivers can be summarized in Table 2-2.

Technology drivers	Fields
Digital	The Internet of Things (IoT)
	Artificial intelligence and machine learning
	Big data and cloud computing
	Digital platforms
Physical	Autonomous Cars
	3D Printing
Biological	Generic Engineering
	Neurotechnology

Table 2-2 Technological drivers for the Fourth Industrial Revolution

As observed from the table above, digital technology is one of the technological drivers for 4IR, and for the purpose of this study, these technologies are the focus as they were used during the COVID-19 pandemic, specifically digital platforms. SMS systems can thus be seen as a digital technology which the model emanating from this study will aim to improve. According

to Schwab (2016b), digital technology is the fundamental driving force for 4IR. Digital power is responsible for nearly all the innovations and advances of the Fourth Industrial Revolution.

The following section discusses the DBE's plan, presented by the Presidential Commission regarding 4IR in South Africa (DCDT, 2020). This is essential information as it indicates the gaps in the education system and how 4IR could help resolve these. A summary of recommendations in the education sector is provided in Table 2-3.

### 2.5.4 The Future Vision of 4IR for Education in South Africa

Public education in South Africa fails to equip learners with the skills they need to become productive members of society (DCDT, 2020). Private schools' rapid growth results from this underperformance since private schools already have advanced 3IR technologies, such as computers, the Internet, and proper SMS. Although the formal education system faces numerous difficulties and difficult realities, some positive steps have been taken to adapt the current basic education system to the demands of the fourth industrial revolution (DCDT, 2020). A process began in 2015 by the DBE to prepare for 4IR, including a review of the current curriculum (DCDT, 2020).

The four competency categories utilized as a benchmark for 21st-century curricula—critical thinking and problem solving, creativity and innovation, cooperation, teamwork, communication, and digital literacy—were determined to be well addressed by the curriculum. However, these competencies are not fully applied in everyday teaching and learning activities and we do not see the results as a country (DCDT, 2020). As a result, the public education system must begin focusing on developing students' creativity, critical thinking, and problemsolving skills. Science, Technology, Engineering, Arts, Mathematics, Innovation, and Entrepreneurship, or STEAMIE, disciplines are more important today to educate students on 21st-century skills, where the arts and humanities play a crucial part in strengthening STEM learning. When STEM is being introduced to South Africa, there is the opportunity to use STEAMIE approaches from the beginning Table 2-6 discusses how STEM was used to mitigate the COVID-19 pandemic (DCDT, 2020).

There is a universal and urgent need for education and awareness of what 4IR represents at many levels in our society. The DCDT (2020) report recommended the creation of a national online portal to showcase the already-existing 4IR centers of excellence and to educate, inform,

and offer updates on training and other opportunities in the 4IR context. Table 2-3 below summarises recommended solutions to address 4IR in education in South Africa.

High-level recommendation (solutions)	Actionable projects / possible areas of intervention
Catalyze Structural Change in the Education System: Efforts must be made to initiate structural changes, as the education system will undoubtedly evolve structurally to reflect the architecture of the 4th industrial revolution over time. The delivery of skills should be enhanced through flexibility, agility, speed of accreditation, integration of learning streams, mobility of learners, remote delivery of content, cognitive flexibility, and technology.	<ul> <li>Redesigning and aligning the skills ecosystem to facilitate agility for 4IR learning.</li> <li>Coordination of the various systems and components within the skill ecosystem must be prioritized to ensure it fits the new skills needs of the 4IR era.</li> <li>People can enter and leave the learning system multiple times during the lifelong learning process with industry-aligned and stackable micro-credentials.</li> <li>Introduce relevant technology and devices (comprehension skills, digital literacy. The Human Resource Development Council should facilitate the systemic change process.</li> <li>Ensure the deliverable is subject to a timeframe, assisted by the 4IR Commission and driven by the Digital and Future Skills Forum. This ecosystem can be linked to a nodal network based on Al that coordinates and streamlines ecosystem components.</li> <li>Align skills according to the national Digital Skills Strategy, which acts as a framework.</li> </ul>

Table 2-3: Summary of recommended solutions with areas of interventions to address 4IR in education in South Africa

Source: Adapted from (DCDT, 2020)

It is evident from Table 2-3 that the alignment of the skills ecosystem with 4IR is prioritised through various bodies (for example, Digital, and Future skills forum and using the Digital Skills strategy of 2020). The importance of digital literacy skills to be able to operate 4IR technology was also mentioned in the Strategic plan 2020-2025 of the State Information Technology Agency (SITA) (SITA, 2022).

Table 2-4 below provides emerging trends, increasing, and decreasing roles in selected industries. In the DCDT report (2020), the following education sector changes in industry and society are provided in the table below (DCDT, 2020):

Table 2	2-4:	Emerging	trends in the	Education	sector w	vith in	creasing	and a	decreasing	roles i	n selected
i	ndus	stries.									

Sectors	Trends Driving Industry Change	Envisioned	Increasing	Decreasing Roles
		Future	Roles	
The	Globalization: Increasingly more	Education is the	Designers of	Registration clerk
Educatio	people will be in the middle class	key to economic	learning offer	Accounting &
n Sector	within the next ten years, increasing	and social	mobile devices	bookkeeping
	the pressure on the education	prosperity in an	Curated	Mediocre teachers
	system; It is critical to recognize	increasingly	knowledge	Librarians
	that inequality of opportunity leads	globalized and	specialists	Proof-readers
	to disparities in well-being, causing	connected world.	Learning	
	political and social unrest.		progress	
			analysts	

Source: Adapted (DCDT, 2020)

It is thus evident from the DCDT report (2020), published after COVID-19, the focus is on globalizing technology to reach more people and increase the use of mobile devices in teaching. But, unfortunately, before the COVID-19 pandemic, there were limitations in the use of these technologies in South African schools (Ibrahim, 2020). Section 2.5.5 discusses digital transformation towards 4IR in the Education Sector in South Africa.

# 2.5.5 Digital Transformation Towards 4IR in the Education Sector in South Africa During the Lockdown

This section will provide an overview of the digital technologies used during the COVID-19 pandemic in South Africa. These technologies ensured continuity and showed the impact of technology during the natural state of disaster. Mhlanga and Moloi (2020) identifies these digital technologies as 4IR technologies, however, as seen from the section 2.5.3, there is a confusion regarding 4IR in South Africa.

### 2.5.5.1 Virtual Learning during the Lockdown

Table 2-5 below shows the digital platforms (Ibrahim, 2020), used by the DBE in South Africa to provide virtual classes during the lockdown that mainly consisted of television and radio. This was a response to the disruption that COVID-19 caused throughout the nation. Lessons were delivered to students via radio, DSTV, and E.TV stations, as stated in the table (König *et al.*, 2020; Mhlanga & Moloi, 2020; Ożadowicz, 2020).

Technology	Description	Connectivity	Platform	Conditions of	Target Group
Used				Use	
Television	Teachers	Offline	Television	Free	Primary
(SABC,	delivering		Desktop	(lockdown)	Secondary
DSTV, E.tv)	lessons live to				
	learners on TV				
Radio (SABC)	Teachers deliver	Offline	Radio/desktop	Free	Primary
	lessons live to			(lockdown)	Secondary
	learners				

Table 2-5: Virtual classes during the lockdown

Source: Adapted from (Mhlanga & Moloi, 2020).

The South African government offered certain online learning options, and several media outlets there backed it. The Department of Communication and Digital Technologies (DCDT) led the initiative. The minister announced that the DCDT and DBE joined forces to ensure this reality through the media (Mhlanga & Moloi, 2020). Schools were closed during lockdown to decrease the impact of COVID-19 (Carvalho, Rossiter, Angrist, Hares & Silverman, 2020). Moreover, the nation's broadcaster built two studios to broadcast virtual classrooms. The teachers presented lessons live in these studios, and E.tv developed a dedicated channel for three months to be displayed on learners' open view (Carvalho *et al.*, 2020). Additionally, courses were provided on several local radio stations across the nation. Thus South Africa can implement virtual classrooms through radio and television, which were during the pandemic, to move the educational system towards the 3IR (Mhlanga & Moloi, 2020).

### 2.5.5.2 Social media and a digital school used during lockdown

However, the DBE was slow in implementing digital technology during the lockdown. Immediately following the lockdown, the DBE, collaborating with the non-profit coding organisation Africa Teen Geeks, launched a STEM lockdown digital school (Mhlanga & Moloi, 2020). A free science, technology, engineering, and maths (STEM) Lockdown Digital School has been established by the Sasol Foundation through the artificial intelligence-based educational platform Ms Zora (Mhlanga & Moloi, 2020). 34 educators from both public and private schools participated in the program, teaching sessions via DBE's social media pages, including Facebook, Twitter, and Ms. Zora online (Mhlanga & Moloi, 2020). Students could access the recordings on the DBE website at any time and these programs were provided free of charge (Mhlanga & Moloi, 2020). The STEM lockdown digital classroom inspired

authorities to digitise education. In the following table, the technology used, the connectivity type and the target groups are provided:

Technology Used	Description	Connectivity	Platform	Conditions of Use	Target Group
Internet	Teachers in public	Online	Desktop	Free	Primary
(website), (Ms	and private schools		laptop	(Lockdown)	Secondary
Zora, SF,	offer classes		mobile		
Siyavula, DBE)	through a live				
	stream				
Facebook (Ms	Teachers in public	Online	Desktop	Free	Primary
Zora)	and private schools		laptop	(Lockdown)	Secondary
	offer classes		mobile		
	through a live				
	stream				
Twitter (Ms.	Teachers in public	Online	Desktop	Free	Primary
Zora)	and private schools		laptop	(Lockdown)	Secondary
	offer classes		mobile		
	through a live				
	stream				

Table 2-6: Use of Social Media Apps and the Introduction of STEM lockdown digital school (SasolFoundation (SF) and African Teen Greeks

Source: Adapted from (Mhlanga & Moloi, 2020).

The maintenance of the system following the lockdown is crucial. This could be done after a thorough performance review of these initiatives (Mhlanga & Moloi, 2020).

Universities and other institutions of higher learning were obliged to only provide online courses because of the lockdown extensions in South Africa. Some of the technology used for these institutions are described in Table 2-7.

Technology Used	Description	Connectivity	Platform	Conditions of Use	Target Group
Internet	Learners learn on their	Online	Desktop laptop	All rights	Tertiary/
(website)	own at home		mobile	reserved	Primary/
YouTube (Most					Secondary
universities)					
Microsoft	Used mainly by staff	Online	Desktop laptop	Freemium	Tertiary
Teams	and learners in tertiary		mobile		
	institutions to hold				
	discussions				
Skype	Used mainly by staff	Online	Desktop laptop	Freemium/ All	Tertiary
	and learners in tertiary		mobile	rights reserved	
	institutions to hold				
	discussions				
WhatsApp	Used mainly by staff	Online	Desktop laptop	All rights	Tertiary
groups	and learners in tertiary		mobile	reserved	
	institutions to hold				
	discussions				
Zoom	Group discussions	Online	Desktop laptop	Freemium	Tertiary

Table 2-7: Exclusive Online Learning-Remote Learning

### Source: Adapted from (Mhlanga & Moloi, 2020)

Several universities in South Africa have shifted to online learning, even though some institutions did not make the change public. For instance, in order to maintain social isolation, students at the Universities of Johannesburg, Cape Town, and Pretoria were informed that their second semester classes would be conducted online (Mhlanga & Moloi, 2020; Ożadowicz, 2020).

Although schools incorporated the above-mentioned digital technologies to ensure learning continuity, not all schools benefited from it to the fullest extent, particularly schools in rural areas as well as marginalised communities where the Internet is not always accessible (Adnan & Anwar, 2020). However, as discussed in Table 2-5, the government ensured there were educational television (TV) programs and radio station access to information in the country, this shows the critical role of applying digital technologies during the pandemic.

There is still a long way for the education system to use 4IR technologies as these will require a lot of investments, skilled personal and time to eventually evolve to the 4IR era, however the use of digital technologies really benefited the education system as can be observed from the section above. Although there is confusion about the Industrial Revolution era in the education sector and how far it has evolved, digital technologies still contributed to successfully combating the negative effect of the pandemic.

Furthermore, the COVID-19 pandemic impacted the administration of the school, and the following section discusses some of the challenge encountered during this time.

# 2.5.6 The Impact on School Related Administration in the Light of COVID-19 Pandemic

The COVID-19 pandemic had a significant impact on every societal sector, including the schools and the school administrations. According to Aytaç (2020) study, half of the 32 school administrators who took part in his study did not have an emergency action plan for the pandemic process and instead followed Ministry of National Education (MoNE) guidelines. Administrators at the schools said that during the pandemic phase, having strong technology leadership and crisis management abilities is crucial (Aytaç, 2020). Another half of the group said that the administrators and their teams developed a comprehensive, long-term plan for handling distance learning during the pandemic, and therefore transparency was essential for making collective decisions (Aytaç, 2020). In Akbaba Altun and Bulut (2023) study, It was found that school administrators' plans were updated due to pandemic conditions that closed educational institutions an extended time. Furthermore, because crisis plans were too formal and prescriptive, they had to be modified to the circumstances. As a result, school administrators said they updated their crisis plans according to the new pandemic circumstances caused by the COVID-19 outbreak. School administrators also felt that the people selected for the crisis response team were not suitable and certain parts of the crisis plan did not work.

The following are the lessons learned during the COVID-19 pandemic as stated by Akbaba Altun and Bulut (2021); and Akbaba Altun and Bulut (2023):

Lessons learned: The Things That Should Be Done Before the Crisis

- 1. Knowing How to Manage Crises: Before a crisis hits educational organizations, being knowledgeable about the ways to cope with crises beforehand is utmost important.
- Having a Foresight Regarding the Crisis: School administrators should have a foresight about the crisis before the crisis, should be more open to communication, and anticipate some situations to be experienced.

- 3. Getting Prepared for Unexpected Situations: School administrators expressed their views that they should be ready for unexpected events and ready for possible problems they may encounter.
- Preparing a Crisis Plan: The research indicated that development of a crisis plans prior to experiencing a crisis was important. Additionally, plans should contain multiple levels and attempt to cover all contingencies.
- 5. Building Crisis-Appropriate Schools: It was also found that, besides preparing a crisis plan, school buildings and related infrastructure need to be suitable for unexpected circumstances.
- Appointing / Being a Competent Administrator: The research also indicated that before a crisis arises, leaders and other people in charge should be capable of administrative tasks and efficient.

Lessons learned: The Things That Should Be Done During the Crisis

- 1. Preparing and Implementing a Crisis Intervention Program: Effective planning is indispensable during challenging times. School administrators stated that normally there should be a crisis plan, but that the plan should be reviewed in cases of a crisis, revised according to the situation and some planning should be made according to the current conditions.
- Taking Action: Taking the necessary actions is one of the valuable findings of the study. In other words, being alert and do what is required is crucial during crises. School administrators expressed their views that as soon as the crisis is felt, action should be taken immediately.
- Supporting: The study also highlighted the importance of supporting the stakeholders. It was stated that they primarily tried to calmly support students and parents.
- 4. Informing: In the following sentences, school administrators mentioned the importance of immediately informing students, teachers and parents about the situation. They said that to control the situation and to produce the solutions to the problems, the informing process should be accelerated. They also believe that quick information gave chance to analyze the situation and take the necessary actions:
- 5. Consulting: It was clear from the findings of the study that benefiting from other colleagues' experiences is useful. It was said that administrators should consult other educators to make the right decision in a crisis.

- 6. Motivating: Motivation became important according to the findings. They should try to maintain their educational activities by keeping their morale high and taking necessary precautions. Administrators should motivate teachers and students positively.
- 7. Emotion Management: They said that administrators should manage their emotions in case of crisis. It has been stated that school administrators should be calm, pioneering, rational, patient, gathering, prudent, selfless, understanding, empathic, proactive and cool.
- 8. Process Management: They stated that in times of crisis, school administrators should plan the process well and manage it together with the team in cooperation.
- 9. Also, it was stated that school administrators should manage the process without departing from reasoning, logic and science in the crisis management process, and do research for better quality process management:
- 10. Leadership. School administrators emphasized the importance of leadership in times of crisis. They stated that school administrators should be calm, visionary, solution-oriented, responsible and conscious in times of crisis. In addition, it was stated that administrators should know media literacy well and use social media effectively.
- 11. Organizing. Organizing constitutes a large part of the crisis management and it is necessary to make the organization perfect.
- 12. Coordination. School administrators have stated that when there is a crisis, school administrators establish a good crisis team, coordinate the teams, and ensure coordinated work at all levels. In this regard, school administrators expressed their opinions.
- 13. Communication. Communication has been the most emphasized by school administrators in crisis management. It was emphasized that in times of crisis, school administrators should be accessible to all stakeholders, enrich their communication channels and keep these channels open. In addition, it was emphasized that school administrators need to frequently inform parents, teachers and students about the process and establish and maintain an effective communication network.
- 14. Taking Joint Decisions. The study indicated that involving other stakeholders during decision making process is one of the right things to do while managing the process. School administrators expressed that they make predictive, rational, practical and participatory decisions in times of crisis.

- 15. Monitoring: School administrators said that they also monitored how remote teaching was carried out
- 16. Solving the Adaptation Problems: Due to Covid-19 pandemic; several emotional, social, environmental and physical adaptations were immediately necessary. After school administrators started remote teaching, they supported teachers and students who had difficulties adapting to the process.

Apart from explaining the role of ICT in education and the opportunities of the 4IR technologies in education, the next section focuses on ICT and interoperability as it is important to understand how it supports teaching and learning, the flow of information in these and between these systems are still necessary to ensure useful quality information. Therefore, the next section covers the importance of ICT and interoperability.

### 2.6 ICT AND INTEROPERABILITY

Interoperability in ICT refers to the transmission and distribution of accurate data, information, and other components, such as collaborations, to systems (including organizations or other entities), operations, and components that are often viewed positively as part of ICT programs (Gasser & Palfrey, 2007; Ohwo, 2020). Additionally, interoperability leads to other positive outcomes such as choice, ease of use, and competition (Gasser & Palfrey, 2007). Innovative technology improves the efficiency of systems, reduces user hassles, and provides consumers with a broader range of options (Gasser & Palfrey, 2007; Esmailzadeh, Noori, Aliahmadi, Nouralizadeh & Bogers, 2020).

Information can flow smoothly between education systems, and the data is useful across all systems relevant to education, which is one advantage of interoperability. It is evident from the previous sections that ICT ensures fewer burdens in the school administration and makes teaching and learning easier. It is also apparent that ICT has negative effects on society and the world, such as security and privacy, but integrating ICT and interoperability ensures the overall information flow from one point to another. This section ties ICT and interoperability, which is important for the study. Without interoperability in SMS, information cannot flow seamlessly, and data errors might occur because of inaccurate data, which can hamper effective decision-making. Data interoperability is also crucial for all schools and the education departments of a specific country

Online learning sites promote an additional learning experience where students can interact, collaborate, and take their lessons faster and with their time, as mentioned in the section above (Ali, 2020).

The next section will further outline the benefits and drawbacks of access to ICT interoperability when assessing it. This is important in the context of the purpose of this study because if one can understand the importance of assessing interoperability, it can assist in making interoperability between SMS more meaningful and worth the effort for schools to use when applying the model from this study. Therefore, assessing the drawbacks and benefits of interoperability in ICT is vital for conceptualizing the model. Furthermore, it provides a clear understanding of the possible impact that can be obtained from ICT Interoperability.

### 2.7 BENEFITS OF ICT INTEROPERABILITY

When assessing ICT interoperability, one must consider its potential benefits and drawbacks as these can assist in conceptualizing the model to ensure it is reliable, secure, and diverse, as indicated by Gasser and Palfrey (2007), and Novakouski and Lewis (2012).

The following benefits of ICT interoperability are highlighted by Gasser and Palfrey (2007), Novakouski and Lewis (2012), and Kerber and Schweitzer (2017):

- **Innovation:** two examples are presented below to illustrate how interoperable systems can and have led to innovation in the ICT space. The Internet is the first interoperable design to which many non-interoperable networks and systems converge. As one of the first applications of the Internet, email proves its importance in this second example. Email protocols or the concept of email were not exclusive, and their designs were generally interoperable.
- **Competition:** competitive markets result in lower prices and provide incentives to design better products and services, which benefit consumers. In general, ICT collaboration promotes innovation by increasing competition.
- Autonomy, flexibility, and choice: users' autonomy and flexibility are enhanced when interoperability levels are higher, and all conditions are carefully considered.
- **Openness:** any ICT environment that fosters creativity and growth relies on interoperability if properly developed and maintained.

Potential Drawbacks, according to Gasser and Palfrey (2007), Novakouski and Lewis (2012), Berryman, Yost, Dunn and Edwards (2013), and Van Alsenoy, Coudert, Jasmontaite and Verdoodt (2015), include the following:

- Security: Interoperability encourages openness and allows data to flow from one end to another. However, open data access can leave a gap for malicious programs. Furthermore, when policies and infrastructure security are incompatible, the benefits of cooperation disappear quickly.
- **Privacy:** a common question regarding cooperation involves a concern that cooperation might reduce privacy. An increased number of interactions can result in more players gaining access to personal information that is exchanged with the operating system.
- **Reliability:** as ICT systems become more complex, they become less reliable, the more complex processes become, the more difficult they are to rectify quickly.
- Accountability: there has been an accountability framework for data protection for more than 30 years. It has been shown that when everyone is responsible, no one is responsible, so when an error is committed, no one takes responsibility. Due to interoperability, transparency of data flow can be a risk to accountability.
- Accessibility: given that the use of ICTs entails inherent risks beyond those of reliability and security, there is a chance that such incidents may induce different players to withdraw from the online environment generally.

All SMS should consider the above adverse effects of ICT interoperability as these can reduce the effectiveness of SMS and, if not considered, can hamper the use and trust in the system. The section below presents the key components of the chapter to consider when conceptualizing the SMSIM model.

### 2.8 KEY COMPONENTS OF THE CHAPTER

Components refer to the essential parts that make up something (Bucherer & Uckelmann, 2011). The table below lists the essential components from this chapter towards conceptualizing the SMSIM model.

Components	Section	Key Focus Area	Author
The impact of ICT on society	2.4	Provides an overview of the greater impact on the daily living of individuals for work, communication, and socialization	(Nureni, 2014), (Fraillon et al., 2020), (Imam & Itodo, 2020)
The role of ICT in education	2.5	This adds to the model by explaining the role of ICT in education. This section cut across the learning, teaching, and administration, which are key to conceptualizing a valid SMSIM model	(Bindu, 2016), (Arshad, 2020), (Del Carmen Trillo- Luque et al., 2020:251), (Hatakka et al., 2020), (Khan, 2020b).

Table 2-7: Essential components of the chapter

### 2.9 SUMMARY

This chapter looked at the overall impact of ICT on society and education and the importance of interoperability and integration. Furthermore, this chapter provided support for the conceptualization of the model by showing both the benefits and drawbacks of ICT and how digital technologies were used during the lockdown caused by the COVID-19 pandemic. The SMSIM model will be conceptualized with the consideration of every item in this chapter. Of specific concern from this chapter for the SMSIM model, is to consider the impact of ICT on society. This chapter contributes towards phase 1 (highlighted with a blue block) of creating the model as per the research process below.



### Figure 2-2: Outline of Dissertation

Chapter 3 discusses and outlines SMS; this chapter will discuss in detail what SMS is, the advantages and disadvantages of SMS, and the purpose. The following Chapter 3 is very important for the study as it discusses the main system the model aims to improve.

## CHAPTER 3: SCHOOL MANAGEMENT SYSTEMS

### 3.1 INTRODUCTION

Chapter 2 provided an overview of ICT and its role in the education sector, as well as its role in interoperability. Furthermore, it discussed the impact of COVID-19 in South Africa, concentrating on how the educational sector was affected, and digital technologies that were used in the education sector to mitigate this crisis and ensure school continuity. This chapter provides an overview of School Management Systems (SMS) for South African schools. First, the chapter explains what SMS is, then describes a few SMSs as per literature findings, including SA-SAMS. From Figure 3.1 it is evident that this chapter forms part of Phase one of this study.



Figure 3-1: The research process highlighting Chapter 3 as part of Phase 1 of the study

In Section 1.1, SMS was introduced, and its history, purpose, and capabilities were discussed. As a continuation of Chapter 1, this chapter explains the advantages SMS offer teachers, administrators, and learners. It also describes the various versions of SMS used by schools, including free and paid versions, and the current state of schools in South Africa. This is vital for this study as some schools already use SMS that differ from SA-SAMS.

Using SMS, schools can innovatively keep track of their records. SMS is a form of Information Management System (MIS) important for schools' smooth operation (Hassan et al., 2014).

Many SMS are typically purchased and owned by individual schools and are located within their premises. Education vendors generally customize SMS to meet the needs of different schools. Some schools purchase servers, desktop computers, and software, such as SMS, software for programming, applications, etc. (Mwangi, 2014). Section 3.2 below discusses advantages of SMS.

## 3.2 THE ADVANTAGES OF SMS

The advantages of SMS are the following, according to Sarker (2016):

- The school's information and activities can be accessed instantly
- School academic processes and administrative duties will be better managed using SMS
- It runs on desktops and laptops
- School budgets can easily accommodate it, making it affordable.

Apart from these general advantages, there are advantages geared explicitly towards teachers in Section 3.3, and administrators in Section 3.4.

## 3.3 THE ADVANTAGES FOR TEACHERS

For teachers, SMS can also assist with (Awadallah, 2016):

- Managing the grades of learners.
- Keep their marks up to date.
- Managing leave applications online.
- Registration of learners and teachers.
- Keep track of learners' attendance.
- Assist in the writing of class notes.
- Staff members, learners, and parents can communicate via e-mail with detailed reports and SMS.

Additionally, SMS offers advantages for administrators, as discussed below:

### **3.4 THE ADMINISTRATIVE ADVANTAGES**

The administrative advantages according to Sarker (2016):

• SMS reduces workload of adding, removing, or registering teachers for the administrator. Teachers can add, remove, and register themselves on the system.

The advantages, according to various authors, are thus obvious. Understanding these advantages will help conceptualise the model, as only the problem areas are addressed, and benefits may be enhanced.

Chapter 1 of the study listed a few types of SMS in the introduction section. From the literature findings, only Fedena, MySchool, Ascend SMS, SchoolTool, and Feraka will be discussed in more detail as per the table below. The table defines and provides capabilities of different SMS systems used by schools worldwide (Arslan, 2013; Adebayo, Kanyinsola & David, 2014; Almgren, 2014; Al-Hayek, 2016; Awadallah, 2016; Mythili, Mishtra & Singh, 2018):

SMS	Definition	Capabilities
Fedena	It is a flexible school management system that serves all management, administrative, and educational needs in thousands of schools throughout the world.	<ul> <li>Teachers have complete control over how they manage their schools' databases, and it provides various tools for managing educational activities.</li> <li>In addition, it can be customized so that institutions can make the necessary modifications to it, thereby making it more effective.</li> <li>There are high-security servers with SSL certificates to provide high-quality data encryption.</li> <li>By using it, institutions avoid the possibility of their data being accessed by malicious individuals.</li> </ul>
My School	In this web-based application, PHP is used to build the web interface and MySQL is used to handle the database backend.	<ul> <li>The ability to manage multiple campuses simultaneously.</li> <li>Provide schools with an Internet front-end.</li> <li>Ensure effective communication with parents and other stakeholders</li> <li>Assist parents, teachers, and students in building and maintaining a community</li> <li>Zero redundancy in managing administrative records</li> <li>Optimize resources as much as possible</li> <li>Management of fees, attendance, and exams are automated.</li> <li>Dynamic substitutions are managed automatically when creating and regenerating timetables.</li> </ul>
Ascend SMS	For Catholic and Independent schools, this is a free, full-featured school administration software program	<ul> <li>It provides parents with smartphone apps, a straightforward discipline reporting system, and a health management system for the school nurse.</li> <li>Among other features, administrators will find the school vitality, learner performance, and reporting feature particularly attractive.</li> </ul>
SchoolTool	This is an open-source cloud-based school administration system designed for schools in developing nations	<ul> <li>Teachers will find it appealing because of its many features.</li> <li>Teachers may manage their classrooms with daily participation journals, skill assessment documents, grade books, and attendance sheets. They can also easily make report cards and interface with well-known calendar programs like Google Calendar.</li> </ul>

SMS	Definition	Capabilities
Fekara	It is an all-around school administration powerhouse with a student population of 50 or less	<ul> <li>It provides all kinds of services, including management and administration of schools, budgets, tests, and assignments.</li> <li>An app is included for convenient access on the go, which can be used on tablets and smartphones.</li> </ul>

Source: Adapted from (Arslan, 2013; Adebayo et al., 2014; Almgren, 2014; Al-Hayek, 2016; Awadallah, 2016; Mythili et al., 2018).

Fedena is designed to be accessible to anyone who needs the SMS. An Internet connection and desktops are all that schools need to implement My School; schools do not need expensive hardware and software. Teachers, students, and parents can all access **MySchool's** system. This keeps learners informed of school activities and grades, and allows parents to contact teachers with complaints, recommendations, or any other issue related to their children's education. Using the system, teachers can communicate with learners and post grades or anything else they want learners to see or receive. The system is under the control of the principals (Awadallah, 2016). According to **Ascend SMS**, schools shouldn't be required to pay outrageous fees for items they require for their success (Awadallah, 2016). The software **SchoolTool** is an excellent starter tool for smaller schools, but it does not provide all the functionality you might need. For example, other applications will have to handle human resources, reporting, and financial management. In addition, the free version of **Fereka** has limits regarding the number of learners, and storage it can handle.

It is evident that different schools use different SMS systems. Depending on affordability and the school's needs, some schools use a free SMS and others use a paid SMS. In the following section, SA-SAMS is discussed in detail.

### 3.5 SA-SAMS

Having briefly introduced SA-SAMS in Chapter 1, this section aims to provide an in-depth explanation of SA-SAMS, its advantages and disadvantages, the modules it contains and how it is typically set up in schools. A key outcome of this assessment will be to conceptualize a model that assesses the level of interoperability in SMS and then provides guidelines to achieve the next level of interoperability. It is also essential to understand why schools are not using SA-SAMS and rather opting for paid versions, this after the exploratory Delphi expert review findings. The following section provides the background and history of SA-SAMS.

### 3.5.1 Background and History of SA-SAMS

The South African education sector has a well-established management information systems (Irene, 2013; Shaikjee, 2020). In 2005, the DBE developed and implemented the South African School Administration and Management Systems (SA-SAMS) to gather statistics on the number of students enrolled in schools (Gxwati, 2011; Kuriakose, 2014; Malan, 2014; Pholotho, 2017; Moyo, 2018; Ramazhamba, Mashiane & Dlamini, 2018). Initially, the Free

State Department of Basic Education took the lead on implementing SA-SAMS through advocacy to all stakeholders in 2007; as a result, all public schools in the province are now using SA-SAMS, a fully integrated computer solution encompassing all aspects of school administration (Gxwati, 2011; Shaikjee, 2020). SA-SAMS has been successfully deployed to all public schools where at least one computer has been available for administrative purposes since 2007 (Gxwati, 2011; Shaikjee, 2020). SA-SAMS has significantly improved data quality in schools with electricity. As a part of SA-SAMS implementation, the department employed trainers on contract, one for each district. To ensure that schools are adequately trained to use SA-SAMS, the trainers trained the principals and administrative staff on how to use the system (Gxwati, 2011).

After understanding its history, it is important to explain what SA-SAMS is, and its capabilities, especially from literature found in the Free State Province. Section 3.6.2 below defines SA-SAMS.

## 3.5.2 Defining SA-SAMS

The table below defines SA-SAMS as per literature findings according to different authors.

	Definition	Author
	The term "South African School and Administration	(Molepo, 2015)
	Management System" is abbreviated as "SA-SAMS." It is a	
	cost-effective, user-friendly, and electronic application	
	designed to support and benefit schools in various ways.	
	Sibiya (2019) describes South African school administration	(Sibiya, 2019)
	and management systems as computer programs that assist in	
	school administration, management, and governance.	
SA-SAMS	Designed, updated, and improved by the DBE, this software	(Van Wyk, 2021)
	solution provides comprehensive school administration and	
	management functions	
	It is a desktop (off-line) system that has been widely used and	(Van Wyk, 2021)
	tested across all provinces (Department of Basic Education,	
	2013)	
	A comprehensive program that collects information about	(Kuriakose, 2014)
	students, parents, and staff in a school.	
	(Van Wyk,	
	system maintained, enhanced, and created by the National	Gondwe & De
	Department of Business Education.	Villiers, 2017)

Table 3-2: Defining SA-SAMS

To show 11: Constant in the first of the second	$(M_{\rm e})$
It simplifies administrative and management tasks and	(Malan, 2014)
streamlines principals' governance and management	
responsibilities in most South African schools.	
This computer application enables public schools in South	(Adam, Butcher,
Africa to manage, administer, and govern themselves.	Tusubira,
	Sibthorpe &
	Souter, 2011)

Sources: Adapted from (Adam et al., 2011; Kuriakose, 2014; Malan, 2014; Molepo, 2015; Van Wyk et al., 2017; Sibiya, 2019; Van Wyk, 2021)

According to all authors in Table 3-2 above, SA-SAMS is an administrative system used by schools, which is a reasonable definition for this study. The purpose of SA-SAMS is described in the following section. SA-SAMS was introduced to address several problems, as outlined in this section.

# 3.5.3 The reason for the SA-SAMS design

There are many different reasons why SA-SAMS was designed, and these includes (Irene, 2013; Malan, 2014; Molepo, 2015; Muriithi & Masinde, 2016; Moyo, 2018; Ncanywa & Sibiya, 2020):

- Disadvantaged schools used several applications to accomplish different tasks; for example, financial obligations were handled by a separate application, another handled assessment, and human resources and timetabling were handled by a third application. Consequently, the same data had to be entered into many different applications several times and updated individually whenever changes were made. To meet this need, SA-SAMS was developed as a comprehensive system for school management that includes all aspects of school administration.
- A primary goal of SA-SAMS was to assist school administrators with access to storing information and simplify school life while establishing district, provincial, and national information databases.
- Schools can use SA-SAMS to manage, govern, and administer their operations.

It is evident that SA-SAMS was designed to address problems in the education sector, and to some extent, the system covers all the necessary aspects. Although this system addressed several problems, it still had some shortcomings or disadvantages, which could be improved with a School Management Systems Interoperability Maturity (SMSIM) model that this study

aims to conceptualize. A typical SA-SAMS setup is described in the next section. This is important for the purpose of the study as it adds to the disadvantages of SA-SAMS as per Table 3-3.

### 3.5.4 Typical Setup of SA-SAMS

The figure below illustrates the architecture of SA-SAMS (Muriithi & Masinde, 2016):



Figure 3-2: The architecture of SA-SAMS (Adapted from Muriithi and Masinde, 2016)

A school administrative clerk or a designated teacher operates SA-SAMS on an independent computer at the school. The school administrator or designated teacher takes care of all data collection tasks, such as processing admissions, recording fees, or capturing learners' performance data. Apart from supporting operational activities at the school, SA-SAMS also serves as the DBE's primary source of school information (Muriithi & Masinde, 2016).

SA-SAMS is typically installed on standalone computers. This causes a lot of problems for school operations as it delays data submissions to the DBE as well as frustrations for the administrators who mostly have access to it, the system limits access; Table 3-3 of this chapter, explains this further under disadvantages. To conceptualize a model that assess the level of interoperability in SMS and provide guidelines to achieve the next level of interoperability in SMS, it is essential to understand these disadvantages and the typical setup of SA-SAMS.

The typical setup of SA-SAMS as per figure 3-2 is only applicable for schools without internet access. Unfortunately, schools without internet access still need to submit data to the DBE through manual means (Post office, CD or USB). Schools with connectivity can submit data to

the DBE electronically through a software product called the Valistractor that is provided by the DBE. Section 3.6.6 of this study discusses the Valistractor in detail. The following section explains the advantages and disadvantages of SA-SAMS.

# 3.5.5 Advantages and Disadvantages of SA-SAMS

SA-SAMS can provide the following advantages and disadvantages (Reddy, Prinsloo, Netshitangani, Moletsane, Juan & Janse Van Rensburg, 2010; Gxwati, 2011; Irene, 2013; Kuriakose, 2014; Malan, 2014; Molepo, 2015; Muriithi & Masinde, 2016; Pholotho, 2017; Moyo, 2018; Mbuqe, 2020; Ncanywa & Sibiya, 2020; Shaikjee, 2020; Strydom, 2020).

Type of School	Advantages of SA-SAMS	Disadvantages of SA-SAMS
Offline Schools (no internet connection)	It ensures that the most up- to-date information is always available to users.	Due to poor connectivity to high schools with limited resources, the SA- SAMS is not connected to other systems at the DBE.
Offline Schools (no internet connection) And Online Schools (internet access)	By using the system, principals can precisely estimate the impact of adding a staff member on the school's budget. The principal can use this information to make an informed decision about the school's management.	It Fixes bugs that develop over time with patches. Schools must download and install these patches on their own computers, and this can sometimes be a hassle for a busy principal. After a year of using the latest version, principals are required to report any bugs and limitations, and must wait for a year before a compact disc (CD) with the updated and corrected system is available. Generally, it takes about one year for an additional feature to be implemented and fixed.
Offline Schools (no internet connection) & Online Schools (internet access)	Can be used for an inventory of stationery at the school	When a fault has been created at the school level, the school is unable to correct it until the Provincial Office is contacted because the staff members are unskilled to solve the fault.
Offline Schools (no internet connection)	Assists with staff training, leave administration, and absenteeism.	Because SA-SAMS is not connected to schools' systems, submitting data to the DBE is difficult and time-consuming. A school extracts requested data from SA-SAMS, copies it to a CD or flash disk, and physically sends it to the

Table 3-3: Advantages and Disadvantages of SA-SAMS

Type of School	Advantages of SA-SAMS	Disadvantages of SA-SAMS	
		DBE. After receiving data from schools, the MOE uploads it to their data warehouse in another lengthy process. School statistics are delayed significantly due to manual submissions.	
Offline Schools (no internet connection)	With the use of SA-SAMS, provincial and national	Despite automating the production of hard-copy records and reports that must	
	departments can quickly and easily access data for annual surveys that will enable effective educational planning.	go to the district office or be kept at school, the system does not enable online data transfers or file transfers.	
Offline Schools (no	SA-SAMS helps schools	If a learner is archived during the year	
internet connection)	put together an annual school survey to send to	and subsequently moved to another school, all the data for that learner from	
And Online Schools (internet access)	the DBE.	January to June will be lost.	
Offline Schools (no	The academic profiles,	SA-SAMS only assists in school	
internet connection)	behavioral histories, exam scores, parent-guardian	administration, which means the information it collects can only be	
And Online Schools (internet access)	information, and other details of students who are enrolled in a class are all kept in the system	accessed by the school's principal and higher authorities.	
Offline Schools (no	Designed to automatically	SA-SAMS fails to share the data it	
internet connection)	handle public and school holidays while	gathers about a student with the educators and parents of the pupils,	
And Online Schools	administering and	despite its broad purpose of supporting	
(internet access)	managing leave. The system offers on-demand details about credits for each person and type of leave.	a principal with administering a school by obtaining information about all students at a school and their activities. The only people who can access records in SA-SAMS are the main and higher authorities.	
Offline Schools (no	The system helps schools	There is a failure to address all	
internet connection)	to comply with all standards, such as those for	stakeholders' needs in the school community, including parents and	
And Online Schools	Snap Survey or school	teachers.	
(internet access)	surveys, and updates can be made periodically in		
	compliance with the most recent policy requirements.		

Type of School	Advantages of SA-SAMS	Disadvantages of SA-SAMS
Offline Schools (no	Ensures learners are placed	SA-SAMS suffers from problems
internet connection)	in the appropriate classes	common to many software systems.
	and promoted	For example, features are only updated
And Online Schools	appropriately.	annually, and software patches for
(internet access)		already existing bugs must be
Offline Schools (no	Develops timetables, relief	downloaded and installed.
internet connection)	registers, and consolidated	
	leave reports monthly and	
And Online Schools	quarterly.	
(internet access)		

From the tabled advantages of SA-SAMS, it is evident that schools that use it have benefited significantly from it. However, as shown in the above table, the system has several shortfalls that needs to be addressed. Some of these disadvantages are a result of lack of internet access in schools, and others are actual disadvantage of SA-SAMS. The section below presents the software product called Valistractor that is provided by the DBE to enables online schools (with internet access) to electronically submit data to the DBE negating the diagram in figure 3-2.

### 3.5.6 The Valistractor Software

The New Leaders Foundation, in partnership with the Department of Education, developed the Valistractor – a tool used to authenticate the quality of data submitted by schools to the Data Driven Districts (DDD) Dashboard (Daniel, 2017). This tool validates and extracts school data to display on the DBE Dashboard, it is a key tool in improving the quality of data collected for 21<sup>st</sup> Century data management (DBE, 2021). The tool is made available to all provinces and to date, more than 7,000 schools are uploading their SA-SAMS data via the Valistractor tool every week, with more than 4,000 users repeatedly accessing the dashboard to augment their decisions. The use of the system drastically improves data accessibility. Importantly, the tool is designed to help district and circuit staff use data to make decisions. Further, the tool assists school management in planning and helps educators identify learners' task-level struggles Kearney (2019).

The above section really provides a solid perspective on the improvements of SA-SAMS. Schools with internet access can realise the full benefits of SA-SAMS using the Valistractor tool. As seen in Kearney (2019) report, over 7000 schools are using this tool and this was documented in 2019, from that time to date, more schools should be using this tool that really

allows SA-SAMS schools to electronically submit data to the DBE. Despite this, there are still some disadvantages of SA-SAMS in which the SMSIM model seeks to address.

SA-SAMS' modules are described in the following section, which illustrates the system's full functionality. Various modules support the management and administration of the school by performing specific tasks.

### 3.5.7 Modules in SA-SAMS

The following figure provides an outline of the different modules found in SA-SAMS (Strydom, 2020):



Figure 3-3: SA-SAMS Modules

The modules will now be discussed (Mathevula & Uwizeyimana, 2014; Molepo, 2015):

• General school information: This module contains details about the school, including names of supporting schools, school cycle information, school terms, classes, demerit codes, subjects offered at the school, and the days of the year that the school is teaching and not teaching. Generally, this module contains the parameters of most managerial functions related to school activities and needs, which must be updated yearly.

- Human resource information: This module includes information about teachers, staff members, performance appraisal records, and weekly attendance records. In addition to helping schools record and compare Integrated Quality Management System (IQMS) results, this module helps schools to track their progress.
- Learner and parent information: In this section, administrators can store information about current and future learners' enrolment. This module can also include information about parents. The significance of this module lies in its ability to report the number of students per week and the number of students promoted to the next grade at the end of the year. Additionally, it contains grades, schedules, and registration forms for learners. The marks that are captured are automatically converted into percentages and levels. Students' marks are also analyzed according to their subject, gender, and language.
- Governance information: This module contains all records related to the school governing body (SGB), such as policies, training records, information about members and training records. The school can use this module to monitor learners' misbehaviour statistics and fees, as these areas require the SGB to make decisions.
- **Financial assistant:** This module provides functions for managing daily finances, including income, petty cash, banking, and cheques. Besides maintaining budgets, the module also helps with bank reconciliations.
- **Standard letters and forms module**: Through this module, a selected group of people can receive new or existing letters. This module allows parents, teachers, and staff to print bank application forms.
- Export data: This module is seen as being extremely crucial since the main advantages of utilizing the SA-SAMS program are thought to be the exporting of data from the 10th day and the data from the yearly school survey. All the necessary information about the school, its human resources, and its students are automatically imported into SA-SAMS using the import function. In addition, the survey is filled out automatically, thus reducing the amount of time that schools must spend on the survey and the number of errors they make.
- Annual national assessments (ANA): This module includes learning outcomes of Annual National Assessment (ANA), parents' information, human resource information, general school information ANA forms, and governance information.
- The curriculum module includes, but is not limited to, recording and reporting student progress. Assigning subjects and placing students in classes is done here. The

evaluation cycle is set up at the end of each semester, which includes tasks required for reporting purposes. A schedule can be printed by subject, class, and year after marks have been captured.

- The timetabling module solves planning problems for both high and primary schools.
- The physical resource module provides teachers with access to student support materials for school retrieval. Students can print reports showing their grades and class and how many books they have for each topic. A list of fixed school supplies can be created based on specifications and incorporated into the asset management system. Reports can be generated for the school's physical resources that have been captured.

Thus, SA-SAMS serves a wide variety of functions that improve the overall functionality of schools; these functions were considered when conceptualizing the SMSIM model. The cost of SA-SAMS' is described in the following section.

### 3.5.8 Costs Implications of SA-SAMS

The deployment of SA-SAMS is currently ineffective as schools need to install the system individually and maintain a copy of the updates on their premises. These duplications lead to longer upgrade cycles which can stretch to a year, and higher implementation costs (Kuriakose, 2014). However, this package is provided voluntarily, and schools are not required to use this tool but are allowed to use the School Administration Management System (SMS) of their choice. In addition, SA-SAMS is provided to schools free of charge to give all schools equal access to the computer system (Shaikjee, 2020). Other school management systems used by smaller schools are Microscope and Edupac, which schools have to pay for, while SA-SAMS is free for all public schools (Strydom, 2020).

It is evident from the above that SA-SAMS is offered to schools for free but has cost implications tied to its maintenance. Table 3-1 discussed some different SMS used worldwide and among them is MySchool SMS, an inexpensive web-based SMS. Schools need the Internet only to implement this SMS and therefore some schools are not using SA-SAMS. This is also why private schools have opted for private software to assist with their administrative activities. Further reasons why the usage of SA-SAMS is low will be obtained during data collection and the results are presented in Chapter 6 of this study. The section below provides the key components of the chapter.

# 3.6 KEY COMPONENTS OF THE CHAPTER

The essential components of this chapter are very important in conceptualizing the SMSIM model.

Components	Section	Key Focus Area	Author
School Management	3.1-3.5	The focus is on the free and open	(Awadallah, 2016),
Systems (SMS)		versions of SMS. These are key as	(Sarker, 2016), (Mythili et
		the data collection will be done on	al., 2018), (Al-Hayek,
		the different versions that will	2016)
		serve as a comparison to improve	
		SMS and conceptualize a SMSIM	
		model	
SA-SAMS	3.6	This SMS is key for the	(Molepo, 2015), (Sibiya,
		conceptualization of the model as	2019), (Van Wyk, 2021),
		it is a free and open SMS that was	(Strydom, 2020),
		developed, introduced, and	(Shaikjee, 2020),
		deployed to various schools in	(Kuriakose, 2014),
		South Africa by the DBE,	(Muriithi & Masinde,
		comparing the results of the	2016).
		Delphi method against the	
		capabilities of SA-SAMS can help	
		improve this SMS.	

### 3.7 SUMMARY

This chapter discussed SMS and provided a table that summarized some different SMS which exist worldwide. It also provided an in-depth explanation of SA-SAMS. This SMS is essential as it is the focus of the study is the SMSIM model that aims to improve SA-SAMS. The importance of this chapter is that it provides a base for the study that will be conducted in schools in South Africa to conceptualize the model. Of specific concern from this chapter for the model is to consider the SA-SAMS, Cost implications of SA-SAMS, why the system was designed, and why schools are using their own SMS. Like chapter 2, this chapter also contributed towards conceptualizing the model as per the picture below.



*Figure 3-4: The research process that highlights how Chapter 3 contributes to the theoretical framework.* 

Chapter 4 outlines and discusses the different Interoperability Maturity Models, which were used to conceptualize the SMSIM model.

# CHAPTER 4: INTEROPERABILITY MATURITY MODELS

### 4.1 INTRODUCTION

As per literature findings, Chapter 3 described SMS and listed the different types of SMS available worldwide; however, only a few SMS were discussed in detail. This is the last literature chapter of phase 1 as displayed in the research process below:



*Figure 4-1: The research process of the study highlighting that Chapter 4 is part of Phase 1 of the study.* 

In this chapter (Chapter 4), the concept of interoperability is introduced, and different Interoperability Maturity Models (IMMs) are described according to literature. Each model is outlined and discussed in more detail. The School Management Systems Interoperability Maturity (SMSIM) model is conceptualized based on two (2) models that were be selected in this chapter.

It might be beneficial, but achieving interoperability is certainly not simple (Brailer, 2005; Zeng, 2019). A clear definition of interoperability does not exist, although it is regarded as a vital success factor (Wozak, Ammenwerth, Hörbst, Sögner, Mair & Schabetsberger, 2008). Without interoperability, utilising technology poses a significant risk (Weber-Jahnke, Peyton & Topaloglou, 2012).
According to literature findings, any organization can benefit from interoperability, but achieving it is not an easy task. Therefore, conceptualizing the SMSIM model was not easy, but it will be beneficial in the end. Therefore, defining interoperability in the following section was essential to understanding the use of interoperability and its potential when SMSIM model was conceptualized.

## 4.2 DEFINING INTEROPERABILITY

Defining interoperability is key to gaining an insight into interoperability, and how it can benefit the outcome of the study. Table 4-1 defines interoperability.

Term	Definition	Author
Interoperability	The concept of interoperability can be viewed as providing	(O'connor, 2017; Sjarov,
	a seamless means of exchanging information, for example,	Kißkalt, Lechler, Selmaier &
	automatically personalizing services, so that other systems	Franke, 2021)
	can use it to improve performance, enable and create	
	services, control operations and process information.	
	Wikipedia defines interoperability as the ability to make	(Elkhodr, Shahrestani &
	systems and organizations work together. In IEEE terms,	Cheung, 2016)
	interoperability is the capability of two or more systems to	
	exchange and use information.	
	In broad terms, interoperability is the ability of various	(Iroju, Soriyan, Gambo &
	data exchange systems and software applications to work	Olaleke, 2013; Zeid,
	together, exchange accurate, effective, and consistent data,	Sundaram, Moghaddam,
	and use the information they exchange.	Kamarthi & Marion, 2019;
		Sjarov <i>et al.</i> , 2021)
	The ability to exchange data and use information between	(Patel, Patel & Scholar, 2016;
	two or more systems or components is called	Kirpes, Danner, Basmadjian,
	interoperability. As a result of this definition, several	Meer & Becker, 2019; Zeid et
	challenges arise, such as:	al., 2019; Jaleel, Mahmood,
	Get the information,	Hassan, Bano & Khurshid,
	Exchange data, and	2020; Sjarov et al., 2021).
	Use the information to understand it and process it	

Table 4-1: Defining Interoperability.

The definition in the last row of the table from Patel, Patel and Scholar (2016), Zeid, Sundaram, Moghaddam, Kamarthi and Marion (2019), Kirpes, Danner, Basmadjian, Meer and Becker (2019), Jaleel, Mahmood, Hassan, Bano and Khurshid (2020), and Sjarov, Kißkalt, Lechler, Selmaier and Franke (2021) concord appropriate for this study as it emphasizes the importance of systems communication and seamless information exchange. The following section outlines and discusses different types of Interoperability Maturity Models.

### 4.3 INTEROPERABILITY MODELS

There have been a lot of study done on interoperability evaluation models. A review of all existing interoperability evaluation models published since 1980 is presented in this section. The Web of Science was searched to identify interoperability evaluation models published between 1980 and 2021. Additionally, Google Scholar was used to supplement the search. Some models for evaluating interoperability are further described considering their significance (Rezaei et al., 2014, Gürdür and Asplund, 2018).

### 4.3.1 Maturity Models

The maturity model shows a path toward a more organized and systematic manner of conducting business. Maturity models are frequently used to evaluate business processes or specific parts of organizations. With maturity assessments, stakeholders can meaningfully evaluate the existing maturity level of an organization's many elements, highlighting areas for improvement and determining what must be done first to move toward a higher degree of maturity (Proença & Borbinha, 2016).

It is generally understood that "maturity" refers to a state of perfection, readiness, or completion, as well as a progression in the development of a system (Schumacher, Erol & Sihn, 2016). Organisations and processes use maturity models to conceptualize and measure their maturity to achieve specific objectives (Schumacher *et al.*, 2016). As-is situations are assessed, improvement measures are derived and prioritised through maturity models, and progress is tracked. An anticipated, intended, or logical progression from a starting state to maturity is represented by a set of stages in a maturity model (Pöppelbuß & Röglinger, 2011). In maturity models, a certain domain's maturity is evaluated using a wide range of criteria, including competency, capability, and level of sophistication (De Bruin, Rosemann, Freeze & Kaulkarni, 2005). Thus, maturity models are tools for evaluating organisational elements and selecting actions that enhance these elements' maturity (Xavier, Reyes, Aoussat, Luiz & Souza, 2020).

The Maturity Model's ability to identify hurdles, promote initial self-assessment, facilitate best practices for doing so, and plan for ongoing improvement is one of its main goals (Mcleod, Babb & Barlow, 2020). A maturity model is a tool that aids in cooperation planning, decision-making, systematizing transformation programs, and analyzing the efficacy of the actions taken (Silva, Ribeiro, Pinto & Monteiro, 2021).

This study uses all the definitions and purposes of the maturity model defined by all four (4) authors (Schumacher *et al.*, 2016; Mcleod *et al.*, 2020; Xavier *et al.*, 2020; Silva *et al.*, 2021). According to them, maturity is the state of being complete, perfect, and ready, used to evaluate the system as-is and assess its maturity. To ensure that the SMSIM model is complete, perfect, and ready, the 'as-is' process of the organisation is noted and improved, and the maturity of the organisation is assessed, it is essential to ensure the completion of the model is complete, perfect, and ready. The section below also answers the SRQ of the study.

SQR 2: What existing models of interoperability maturity can be used to conceptualize a SMSIM model?

Below are Sections 4.3.2; 4.3.3; 4.3.4; 4.3.5; 4.3.6, and 4.3.7, which discuss various Interoperability Maturity Models (IMM). This section is critical as two (2) IMM models were chosen to help conceptualize the SMSIM model for schools in South Africa. As presented below, the first IMM is the Levels of Information Systems Interoperability (LISI) interoperability maturity model.

### 4.3.2 Levels of Information Systems Interoperability (LISI)

The Levels of Information Systems Interoperability (LISI) was created in 1998 by the United States (US) Department of Defence C4ISR Working Group. The US Department of Defense uses US intelligence agencies and the defence community's C4ISR (Command, Control, Communications, Computers, Intelligence, Surveillance, and Reconnaissance) systems (Mingqian & Shuai, 2020). An information system's interoperability can be assessed by using the LISI reference model. This technique establishes, gauges, evaluates, and verifies the level of interoperability that organizations or systems should strive towards (Hendrawan, 2018; Espadinha-Cruz, Grilo, Gonçalves-Coelho & Mourão, 2019; Sjarov *et al.*, 2021). LISI measures the degree of interoperability between systems. Figure 4-2 illustrates the levels of the LISI model. In the LISI model, complexity levels of interoperability are enhanced within the system. Isolated, Connected, Functional, Domain, and Enterprise are among the five degrees of interoperability (0–4), each of which resides inside a specific environment (Guédria, Naudet & Chen, 2008; Rezaei, Chiew, Lee & Aliee, 2014; Jabin, Dimyadi & Amor, 2019b; Luo, Zhang, Mao, Kou & Zhang, 2019; Sjarov *et al.*, 2021).

- Level 0 Isolated interoperability this level has numerous standalone or isolated systems in a manual environment. It is impossible to directly connect to these systems, which have a manual interface. There is an integration and extraction of data manually between multiple systems at this level of interoperability.
- Level 1 Connected interoperability within a peer-to-peer setting: Interoperability between connected systems is based on straightforward electronic data transfer.
- Level 2 Functional interoperability in a distributed environment: This level permits data exchange between systems that are connected by a local network area. Systems share logical data models and provide increasingly sophisticated media exchanges. Functions and systems can share information by combining heterogeneous data in a simple information format.
- Level 3 Domain-based interoperability in a coordinated setting: The domain-based interoperability level allows for the sharing of data across numerous users via wide area networks (WANs). At this level, domain data models are used for information interchange between separate applications. Domain-based interoperability systems facilitate database-to-database interaction through business rules and processes.
- Level 4 Enterprise-based interoperability in a general environment: Through enterprise-based interoperability, systems can access global data across several domains. At this level, multiple users can access the data concurrently, and both the programs and the data are equally shared and distributed. This level also allows for advanced collaboration. All common data formats are interpreted in the same way across the entire enterprise.

The visualization of the LISI model is provided below.

## KENEILWE MAREMI: 44977883

LEVEL		Interoperability Attributes				
(Environment)		Procedures	Applications	Infrastructure	Data	
Enterprise Level (Universal)	4	с Ь	Multi-National Enterprises Cross Government	Interactive (cross applications)	Multi- Dimentional	Cross- Enterprise Models
		a	Enterprise DoD Enterprise	Full Object Cut & Paste	Topologies	Enterprise Model
Domain Level (Integrated)		с	Domain Service/Agency	Shared Data (Situation Displays Direct DB Exchanges)	WAN	DBMS
	3	ь	Doctrine, Procedures, Training, etc.	Group Collaboration (White Boards, VTC)		Domain Models
		a		Full Text Cut & Paste		
		с	Common	Web Browser		
Functional Level (Distributed)	2	ь	Operating Environment (DII-COE Level 5) Compliance	Basic Operations (Documents, Maps,Briefings, Pictures Spreadsheets, Data)	LAN	Program Models & Advanced Data Formats
		а	Program Standard Procedures, Training, etc.	Adv. Messaging (Parsers, E-Mail+)	Network	
Connected		d	Standards Complaint (JTA, IEEE)	Basic Messaging (Plain Text, E-mail w/oattachments)	Two Way	Basic
Level (Page to Pages)	1	с		Data File Transfer		Data
(Peer-to-Peer)		ь	Security Profile	(Text Chatter, Voice, Fax, Remote,		Formats
		а		Access.Telemetry)	One Way	
lsolated Level (Manual)		d	Media Exchange Procedures		Removable Media	Media Formats
		с Ь	Manual Access Controls	N/A	Manual Re-entry	Private Data
		а 0	NATO Level 1 NO	KNOWN INTEROPE	RABILITY	

Figure 4-2: The LISI Model (Ostadzadeh, Shams & Badie, 2015; Gürdür & Asplund, 2018)

The LISI process is built on the LISI Reference Model. The attributes of the LISI Reference Model can be seen in the rows and columns of the five LISI interoperability levels as Procedures, Applications, Infrastructure, and Data (PAID). The essential specialized capabilities are made possible by the broad categorisation of level/attribute intersections. As a

result, the LISI divides interoperability aspects into four uniform properties (Rezaei *et al.*, 2014; Gürdür & Asplund, 2018; Hendrawan, 2018; Luo *et al.*, 2019):

- **Procedure attributes** include various forms of operational controls and documented guidance that impact the integration, development, and operation of systems. Further, standards, policies, and procedures are addressed in the procedure attributes to facilitate the exchange of information between systems.
- Application attributes include the system mission, the primary reason for creating the system, and functional process attributes—which include a variety of operational controls and documentation—have an impact on all facets of system development, integration, and operation. The procedural qualities contain architectural rules, standards, policies, and procedures that permit information interchange between systems. The primary objective of system development, the system mission, and the functionality requirements are examples of application qualities. These characteristics point to programs that provide processing, exchange, and manipulation.
- **Infrastructure attributes** establish and maintain a connection between applications and systems. Among these attributes are the systems, networks, and hardware that enable the interaction.
- **Data attributes** describe the system's information processes, encompassing both its format (syntax) and content (semantics). Among the data attributes of interoperability are protocols and formats that enable the exchange of data and information.

The LISI Model offers the valuable feature of expressing the results in interoperability metrics to assess interoperability. The level of system interoperability is measured by the LISI metric. The Interoperability Questionnaire and the LISI Capabilities Model act as data aggregators and measurement templates to assess the level of interoperability (Rezaei *et al.*, 2014; Gürdür & Asplund, 2018).

Based on the Interoperability Maturity Model above, it is evident that LISI provides a process for assessing an information system's interoperability, focusing on enhancing its interoperability levels of complexity. However, this model will not be considered for the study as it was extended by the Organisational Interoperability Maturity Model (OIM) (discussed below) that assesses the non-technical and human activity level to interoperate with other organisations (Maremi, Herselman & Botha, 2020). The following section describes OIM.

### 4.3.3 Organizational Interoperability Maturity Model (OIM)

This concept extends the LISI by adding a more amorphous layer of command-and-control functionality. This model defines five levels of organisational maturity to describe interoperability. Table 4-2 provides a detailed illustration of the organizational interoperability maturity model (Haghighinasab & Khosravi, 2011; Rezaei *et al.*, 2014; Gürdür & Asplund, 2018; Jabin *et al.*, 2019b; Sjarov *et al.*, 2021).

The Organizational Interoperability Maturity (OIM) model was created in 1998 by the Australian Defence Science and Technology Organization to evaluate the interoperability of organizations (OIM). The objective of OIM is to assess an organization's ability to interoperate with another by looking at nontechnical or human-activity aspects. In addition to four organizational interoperability qualities, it has levels equivalent to LISI (independent, cooperative, collaborative, combined, and unified) (preparation, understanding, command style, and ethos) (Guédria *et al.*, 2008; Haghighinasab & Khosravi, 2011; Rezaei *et al.*, 2014; Gürdür & Asplund, 2018; Mcdaniel, 2020). The four enabling traits in OIM are preparedness, understanding, command style, and ethos and are characterized as follows (Mcdaniel, 2020):

- **Preparedness:** This attribute indicates how prepared an organization is to interoperate. Furthermore, it focuses on training, experience, and doctrine that enables different organizations to work together.
- The understanding attribute measures understanding of how information and knowledge are shared within the organization and how they are used. Necessary levels of information and knowledge-sharing are considered in the attribute.
- **Command style:** This attribute describes the way decisions are made, how roles and responsibilities are assigned, and how the organisation is managed. As part of this process, roles and responsibilities are examined both delegated and shared.
- Ethos: This attribute refers to the objectives, values, culture, and trust of an organization. Since its introduction, the organisational interoperability maturity model has been reviewed by several researchers.

Then, on a scale of 0 to 4, these four characteristics are evaluated across five stages of increasing interoperability: independent, cooperative, collaborative, combined, and unified.

	Preparedness	Understanding	Command Style	Ethos
Level 4 Unified	Complete-normal day-to-day working	Shared	Homogenous	Uniform
Level 3 Combined	Detailed doctrine and experience in using it	Shared comms and shared knowledge	One chain of command and interaction with home org	Shared ethos but with influence from home org
Level 2 Collaborative	General doctrine in place and some experience	Shared comms and shared knowledge about specific topics	Separate reporting lines of responsibility overlaid with a single command chain	Shared purpose: goals, value system significantly influenced by home org
Level 1 Ad hoc	General guidelines	Electronic comms and shared information	Separate reporting lines of responsibility	Shared purpose
Level 0 Independent	No preparedness	Communication via phone etc	No interaction	Limited shared purpose

Table 4-2: Organizational Interoperability Maturity Model (OIM)

Source adapted from: (Lane & Valerdi, 2011)

# 4.3.4 Levels of conceptual Interoperability

To address the need for conceptual interoperability that extends across technical models like LISI, conceptual interoperability model tiers were proposed. The paradigm views interoperability as a conceptual challenge rather than a technical one, and it seeks to integrate conceptual and technical design. The levels are (Gürdür & Asplund, 2018; Jabin *et al.*, 2019b; Leal, Guédria & Panetto, 2019):

• Level 0 - Independent: At this level of interoperability, interactions between autonomous organizations are detailed. Organizations operate at this degree of interoperability with only personal interactions. Organizations on this level may not share common goals or purposes but may have to interoperate in scenarios with no precedent. Interoperability at this level leads to unexpected and unplanned arrangements. Despite this level's absence of formal frameworks, communication occurs through meetings, faxes, and telephone calls.

- Level 1 Adhoc: Few organizational frameworks can support ad hoc arrangements at this level of interoperability. Guidelines explain interoperability, but specific arrangements have not been finalized. Overarching goals are shared at this level of organizational interoperability. Despite this, organizations continue to exist as independent entities because of individual, organizational goals.
- Level 2 Collaborative: To achieve this level of interoperability, recognized frameworks must be prepared, and common objectives must be determined. Moreover, roles and responsibilities are assigned based on ongoing responsibilities, although the organizations remain distinct. The level of organizational interoperability is where knowledge exchange and training take place as the last stage.
- Level 3 Integrated: There is a common understanding, shared goals, and readiness to interoperate at this level of interoperability. Using it as a detailed doctrine has a significant amount of experience. The frameworks at the integrated level of organizational interoperability have been prepared and tested, despite any remaining ties to a home organization.
- Level 4 Unified: At this degree of interoperability, organizational objectives, value systems, command structures, and knowledge bases are shared. Organizational interoperability is continuous at the unified level. There are no impediments to complete full interoperability in the organizational framework at this level. However, it is likely that only very homogeneous organizations will achieve a unified level of interoperability.

OIM evaluates humans' ability to interact with the system and other organizations. The LISI model is extended by this model. Thus, OIM was chosen to conceptualize the SMSIM model for schools in South Africa. It is important to understand how SMS users interact with SMS and their shortfalls; this model (OIM) does that. However, this model is unable to look at the technical elements of SMS to ensure interoperability and thus, this model was combined with Information Systems Interoperability Maturity Model (ISIMM), which focuses on the system itself as per Section 4.3.7 (Maremi *et al.*, 2020). The following section lists and outlines the Enterprise Interoperability Maturity Model (EIMM).

### 4.3.5 Enterprise Interoperability Maturity Model (EIMM)

The enterprise interoperability maturity model was created by the European Commission through the ATHENA IP (Advanced Technologies for interoperability of Heterogeneous Enterprise Networks and their Applications Integrated Project). There are identified maturity levels and areas of concern in the enterprise interoperability maturity model (See Figure. 4-3). Interoperability and collaboration issues would be defined by specific objectives and goals for each area of concern. The interoperability and collaboration maturity levels are defined depending on the existence or absence of maturity indicators. Each indicator must meet the threshold conditions or reach the level required for each maturity level (Guédria *et al.*, 2008; Rezaei *et al.*, 2014; Ostadzadeh *et al.*, 2015).



Figure 4-3: Enterprise Interoperability Maturity Model (EIMM) (Ostadzadeh et al., 2015)

The following section lists and explains the maturity levels in EIMM:

Business strategy and processes: The area of business strategy and processes is concerned with defining, identifying, and executing company strategy and procedures. It pursues and necessitates strengthening the cooperation procedures for various organizational units in addition to the external entities to attain interoperability in this area of concern.

- **Organization and competencies:** This area of concern deals with improving, enacting, specifying, and identifying organizational structures containing the skills and knowledge of the identified players. Identifying the external entities to collaborate with, specifying the networked organization topology, and improving and deploying it to achieve interoperability in this area is necessary.
- **Products and services:** This area focuses on the design, specification, identification, and lifecycle strategy of an organization's products and services, as well as its quality characteristics. Finding new opportunities and specifying the same components for products and e-services that use networked technologies to achieve interoperability in this area is necessary.
- **Systems and technology:** This area of concern includes improving, maintaining, operating, acquiring/constructing, designing, specifying, and identifying enterprise systems. Ideally, this involves developing links to enterprise models and tracing them back. Innovative technologies that foster interoperability must be researched and evolved to achieve interoperability in this area of concern.
- Legal environment, security, and trust: This area of concern focuses on recognizing the requirements for trust, security, and the law, working with outside parties to manage these issues, and offering solutions to do so because they are essential to interoperability.
- Enterprise modelling: An all-encompassing sixth area of concern directly affects all the previously identified areas. Enterprise Modelling refers to improvements, applications, constructions, and specifications for enterprise models. For enterprise modelling, the specification identifies the proper languages, meta-models, infrastructure, techniques, organization (skills and people), and related activities. Additionally, it emphasizes the compatibility of corporate models.

The following are the five stages of the enterprise interoperability maturity model (Rezaei *et al.*, 2014):

• **Performed:** Enterprise modelling and collaboration are performed ad hoc at the Performed maturity level. Organizations collaborate with external entities (customers, administration, suppliers), but the relationships are not planned carefully. It is common for collaborative processes and tasks to exceed budgets and schedules at this level.

Aside from that, they cannot repeat their past success and are not utilizing the technology to its full potential.

- **Modelled:** Enterprise modelling and collaboration are carried out similarly at the Modelled maturity level, and the method has proved to be useful. The application of the described techniques and meta-models occurs at this level. Furthermore, roles are clearly defined, and everyone is aware of the business model's implementation. To promote collaboration, network technologies are also used.
- **Integrated:** Enterprise modelling has been formally documented, communicated, and applied consistently at the Integrated maturity level. This level involves organizations using a defined framework and methodology for enterprise modelling, integrating the various dimensions, and tracing the model back to the enterprise IT infrastructure. Models are improved using a knowledge base. As part of the enterprise model, interoperability technologies, standards, and externalisation facilitate business collaboration.
- **Interoperable:** At the Interoperable maturity level, enterprise models offer dynamic interoperability, changeability, and external entity evolution. At this level, the enterprise model smoothly incorporates the workplace of the employees. Results and process metrics are the foundation of continuous improvement for the individuals and organizations involved.
- **Optimizing:** To achieve the Optimizing maturity level, an organization must be able to adapt and react to changes in the business environment quickly, efficiently, and with agility. This level aims to integrate enterprise systems with enterprise models and improve interoperability by researching and applying innovative technologies. Enterprise modelling could be utilized at the Optimizing maturity level to fulfil the overall goals of the associated business, people, or unit.

This model will not be considered for the study as it focuses on the enterprise's interoperability maturity model. This model will be used when the study's main focus starts assessing SMS as an enterprise (Maremi *et al.*, 2020). The following section discusses the Government Interoperability Maturity Matrix (GIMM).

#### 4.3.6 Government Interoperability Maturity Matrix (GIMM)

Figure 4-4 illustrates the government interoperability maturity matrix (GIMM) that was presented by Sarantis, Charalabidis and Psarras (2008).



Figure 4-4: Government Interoperability Maturity Model (GIMM)

With the model, administrations can perform a simple self-evaluation to assess their status in relation to e-Government interoperability as well as what steps need to be taken to improve their positioning when it comes to implementing e-Government systems and providing services to citizens and businesses. The European Interoperability Framework's (EIF) interoperability types are expanded by this maturity model, which also identifies several interoperability attributes that must be considered when assessing each organization's situation regarding e-Government interoperability. The GIMM is depicted in Figure 4-4 as being composed of a number of levels, each of which corresponds to a distinct interoperability level for a collection of interoperability qualities (IA) (Guédria *et al.*, 2008; Rezaei *et al.*, 2014; Cestari, Loures & Santos, 2018; Fernandes, Neto & Santos, 2018; Jabin *et al.*, 2019b; Jabin, Dimyadi & Amor, 2019a; Almeida Prado Cestari, Loures, Santos & Panetto, 2020).

### 4.3.6.1 Five levels of maturity are defined in the GIMM

The GIMM have the following five levels (Guédria et al., 2008; Rezaei et al., 2014; Cestari et al., 2018; Fernandes et al., 2018; Gürdür & Asplund, 2018; Jabin et al., 2019a, 2019b; Almeida Prado Cestari et al., 2020):

- Level 1 Independent: The connection between independent groups is discussed at the independent level.
- Level 2 Adhoc: Ad hoc arrangements can be supported only with minimal organizational frameworks.
- Level 3 Collaborative: There are recognized mechanisms in place to promote interoperability at this level. Even when shared goals are determined and roles and tasks are allocated, organizations maintain their individuality.
- Level 4 Integrated: This degree of interoperability calls for a common concept of interoperability with other organizations, as well as common goals and value systems.
- Level 5 Unified: At this degree of interoperability, organizations share knowledge bases, organizational goals, organizational structure, and values.

## 4.3.6.2 Three main interoperability dimensions in the GIMM

The following three interoperability dimensions are defined in the GIMM (Rezaei et al., 2014):

- Organisational interoperability: The organizational interoperability component entails creating collaborations between organizations that share information and may have various internal structures and processes, as well as formulating business objectives and structuring business processes. Organizational interoperability also attempts to satisfy the needs of the user community by making certain that services are user-friendly, accessible, clearly identified, and quickly available.
- Semantic interoperability: This dimension ensures that any application can understand the precise meaning of information exchanged, regardless of its original purpose. Combined with other information resources, this dimension facilitates systems' ability to process information meaningfully. A multilingual front-end delivery of services is, therefore, a prerequisite of semantic interoperability.
- **Technical interoperability:** Technical interoperability connects computer systems and services. Security, accessibility, data display, data interchange, middleware and data integration, and connectivity services are among the services included in this dimension.

GIMM defines several Interoperability Attributes classified into three interoperability dimensions using a parametric score matrix. Hence, a vector containing the GIMM Interoperability Attributes is a good representation of the interoperability state. Lastly, the

Government Interoperability Maturity Matrix provides mathematical formulations and patterns to consider when transitioning a government organization to an interoperability state (Rezaei *et al.*, 2014; Gürdür & Asplund, 2018).

In addition to assessing the state of the administrations in terms of e-Government interoperability, this model also identifies the steps that need to be taken to improve system implementation. Due to the advanced nature of this model, which focuses specifically on e-governance, it was not considered when the SMSIM model of the study was conceptualized. The SMSs in the education sector is used by both private and public institutions that must submit data to the DBE, therefore the GIMM model narrows the focus to e-government and does not consider private sectors. Moreover, the study focused on schools in South Africa that uses different SMS to conceptualize the SMSIM model (Maremi *et al.*, 2020).

## 4.3.7 Information Systems Interoperability Maturity Model (ISIMM)

To evaluate the interoperability maturity of information systems, the Information Systems Interoperability Maturity Model (ISIMM) was created. The LISI and GIMM theories served as the foundation for the ISIMM, which primarily focuses on the technical components of interoperability that would enable data sharing and exchange in an information systems environment (Van Staden & Mbale, 2012; Campos, Chalmeta, Grangel & Poler, 2013; Anggoro, Hubeis & Sailah, 2018; Almeida Prado Cestari *et al.*, 2020).

ISIMM in Figure 4-5 specifically outlines the stages and levels of interoperability sophistication that an organization's information systems will go through. Information system interoperability maturity (ISIMM) provides a standardized and systematic approach to analyzing and measuring. The construction of an interoperable systems environment within the government will be made easier by ISIMM's provision of the means to gain a deeper understanding of interoperability in information systems, which goes beyond investigating interoperability's intricacies (Van Staden & Mbale, 2012; Ostadzadeh *et al.*, 2015; Anggoro *et al.*, 2018; Almeida Prado Cestari *et al.*, 2020).



Figure 4-5: Information Systems Interoperability Maturity Model (ISIMM)(Rezaei et al., 2014).

The following areas are where the ISIMM focuses on the technical interoperability of information systems (Van Staden & Mbale, 2012; Campos *et al.*, 2013; Ostadzadeh *et al.*, 2015):

- Data interoperability: The ability of various software applications from heterogeneous systems to comprehend the syntactical and semantic meaning of data obtained from various data models by using common data models, mappings, and structures is referred to as data interoperability.
- **Software interoperability:** This is described as the ability of diverse software organizations to exchange and share data by solving their differences.
- **Communication interoperability:** User systems can communicate and connect through standard protocols to provide interoperability.
- **Physical Interoperability:** This refers to the ability of computer hardware, network devices, and peripherals to interact with one another.

Figure 4-6 illustrates the maturity interoperability computing environment tiers of ISIMM (Van Staden & Mbale, 2012; Campos *et al.*, 2013; Ostadzadeh *et al.*, 2015):

- Level 1 Manual: There are no connections between information systems, and data is shared manually between systems.
- Level 2 Adhoc: The basic sharing of non-standardized data takes place electronically with other organizations. Databases and applications are separated, and data is not shared between organizations. Instead, data is exchanged between systems in a point-to-point manner on an ad-hoc basis.
- Level 3 Collaborative: This level enables broader connections with legacy systems. In a distributed setting, basic collaboration is achieved between independent

applications. Logical data models are shared and used in the data exchange process. However, minimal standard functions exist, applications and databases are separated, and data is not shared.

- Level 4 Integrated: Data in the integrated stage are shared to some extent, and data is exchanged between independent applications using shared domain-based data models. Collaboration is at an advanced domain level. Integration of services or systems is being implemented between organisations.
- Level 5 Unified: At this stage, data and applications can be fully shared and distributed between organizations. Organizations are interoperating continuously through high-quality services at this level of enterprise collaboration. There is a common interpretation of data, and it is based on a standard exchange model. There is full interoperability between front-end and back-end systems. The processes at this level are also automated.



Figure 4-6: information systems' interoperability maturity transition (Campos et al., 2013, Van Staden and Mbale, 2012)

The transition of an interoperable environment from a very disparate information systems environment to a highly common integrated and shared information systems environment, as indicated in Figure 4-6, is defined by the interoperability maturity stages of ISIMM. This classifies a change in the interoperability of the information systems environment from a low level to a high level (Van Staden & Mbale, 2012; Campos *et al.*, 2013; Ostadzadeh *et al.*, 2015).

It is evident from the section above that ISIMM focuses on the technical aspect of interoperability of the system; this makes it a good model to combine with OIM, which focuses on the human aspect to ensure a valuable model is conceptualized. For an organization to

operate effectively, it is important to evaluate how the organization's system supports its dayto-day operations and how the people interact with the systems put in place. In other words, the users of SMS and SMS are equally important in the school's operation (Maremi *et al.*, 2020).

# 4.4 CONCEPTUALIZING THE INTERIM SCHOOL MANAGEMENT SYSTEMS INTEROPERABILITY MATURITY (SMSIM) MODEL

Chapters 2, 3, & 4 of the literature review assisted in the conceptualization of the School Management Systems Interoperability Maturity (SMSIM) model. Figure 4-7 below outlines how each chapter contributed to conceptualizing the SMSIM model across different literature review sections.



Figure 4-7: Conceptualizing the School Management Systems Interoperability Maturity (SMSIM) model

It is evident from this figure that Chapter 2 focussed on ICT aspects, Chapter 3 on school management systems and Chapter 4 on interoperability maturity models. Concepts from these literature chapters will be applied to conceptualize the model.

### 4.5 CHOOSING OIM AND ISIMM

To conceptualize a model that is effective, reliable, and efficient, Interoperability Maturity Models (IMM) were outlined and discussed in the following sections: 4.3.2 - 4.3.7. From those IMMs, Organization Interoperability Maturity Model (OIM) and Information System Interoperability Maturity (ISIMM) Models were chosen for conceptualizing the initial SMSIM model before the data collection process.

These models were chosen because:

- The OIM concept builds on the LISI architecture and emphasizes interoperability. This model focusses on the ability organisation to interoperate. The model also focus on employees in the organisations and how they interact with the changes in the organisation. See Section 4.3.3 for more information on how this model evaluates non-technical or human activity and an organization's capacity for collaboration.
- The ISIMM assesses the level of interoperability between information systems; this is important as it will give a clear picture of interoperability in the SMS. As information exchange is the main goal of interoperability, ISIMM also concentrates on the more intricate technological elements that enable data to be shared and exchanged within an information systems environment. With this maturity model, a deeper comprehension of information systems will be gained, which is essential for SMS, see Section 4.3.7.

Both models were created as extensions of the LISI model, and they are quite complementary to one another. It is crucial to comprehend the SMS's level of interoperability so that it may be enhanced, and it is only marginally significant to predict how users will interact with the system. From system to organizational adaptation, these two theories offer a comprehensive perspective.

Since the LISI Model resembles the ISIMM more, it is not considered in this context. The LISI model extends the ISMM and OIM model and measures the degree of interoperability between systems (see sections 4.3.2, 4.3.3 and 4.3.7). When interoperability has been incorporated into SMS, GIMM can be used to conduct a self-evaluation that evaluates the administrations' current position regarding e-government interoperability and the steps necessary to improve their posture with relation to system implementation. Future uses of the GIMM model can be considered (see Section 4.3.6). Because EIMM is a high-level maturity model that focuses on

the enterprise, see Section 4.3.5, it is not considered. The two (2) models were combined in the next part are (OIM & ISIMM).

# 4.6 COMBINING OIM AND ISIMM

Section 4.3 discussed different Interoperability Maturity Models (IMM) which exist. Section 4.5 of this study chose two (2) IMM from the list that were discussed and justified why only two (ISIMM & OIM) models were chosen to conceptualize the interim SMSIM model. In this section, Table 4-3 below, combines the two models chosen to conceptualize the interim SMSIM model for schools in South Africa. The interim SMSIM model will be given to experts for evaluation based on determining the worth, significance, validity, and value of the model for SMS. These results will be documented in Chapter 6.

Organizationa	l Interoperability	v Model (OIM)	Information Systems Interoperability Maturity Model (ISIMM)		
Level 4 unified	Complete normal day-to- day working	Shared understanding	information is interpreted the same. Systems are fully interoperable.	Complete data sharing	Level 5 Unified
Level 3 Integrated	The same level of understanding and preparedness to interoperate	common goals and value systems	There is a higher level of collaboration and services integration	There is data that is shared to a certain extent in the integrated stage	Level 4 Integrated
Level 2 Collaborative	frameworks and common goals	Shared Comms and shared knowledge	Least shared functions exist	There is a broader connection to legacy systems	Level 3 Collaborative
Level 1 Ad hoc	Guidelines	Shared Purposes	No data that is shared among organizations	There is a separation of applications and databases	Level 2 Ad-Hoc
Level 0 Independent	No Preparedness	No Interaction	No sharing of data	No connection of Information Systems	Level 1 Manual

This table combined ISMIM (4.3.7) and OIM (4.3.3) models, the components and levels in these models are discussed in their respective sections (4.3.7, 4.3.3). These models will serve as the main guideless for achieving interoperability in SMS. These models cover two crucial areas: how users interact with the system and the interoperability of SMS as a digital technology. The expert review Delphi questionnaires are mostly derived from this combined table to conceptualize the SMSIM model that cuts across SMS from the system's perspective to the user. The output of the findings during data collection was used to further refine the conceptualized SMSIM model of this study.

A combination of OIM and ISMIM also answers the sub-research question 1 of the research questions.

SQR 1: What constitutes the components of a model of interoperability maturity for Schools Management Systems?

The following section was derived from Table 4-3 above and was verified and validated through the expert review results. In Chapter 6, this did not change but ed-admin was assessed using this model. This section presents the interim SMSIM model conceptualized based on the current state of SA-SAMS and different SMS according to literature and the two combined maturity models in Table 4-3.

# 4.7 THE INTERIM SCHOOL MANAGEMENT SYSTEMS INTEROPERABILITY MATURITY (SMSIM) MODEL

The School Management Systems Interoperability Maturity (SMSIM) model is a feeder model that assesses the level of interoperability in the SMS and continuously provides guidelines to improve the current level of interoperability in SMS. This model consists of five phases: the foundation, primary, intermediate, secondary phase, and tertiary phases. The SMSIM model uses the tested, verified, and approved Interoperability Maturity Models (IMM) which are a combination of Information System Interoperability Maturity Model (ISIMM) and Organizational Interoperability Maturity Model (OIM) to assess and provide guidelines for improving SMS. A feeder at each phase goes into each level of ISIMM and OIM to assess interoperability at each level and provide guidelines for the next level.

From the model, it is evident that on the left side, one finds the ISIMM with its interoperability levels with the same number of levels with different names as the OIM on the right side of the

### KENEILWE MAREMI: 44977883

model that focuses more on the organizational maturity to provide interoperability. Both has five levels of interoperability maturity. OIM has levels 0-4 and ISIMM has levels 1-5. Only the names of the first level in each differ (ISIMM has a manual level because at this low level, information systems are not connected, and data is shared manually). The OIM refers to this level as independent because there is no preparedness in the organisation to have interoperability at this level.

The second level in both models uses the phrase Adhoc to refer to a level where only basic data is shared (ISIMM) and in the OIM the organization's preparedness has guidelines and the understanding is shared with a specific purpose.

Moving to the third level in both one sees the use of the same name (Collaboration) as at this level in ISIMM there is collaboration regarding a broader connectedness to legacy systems with basic collaborations. In the OIM there is collaboration regarding the sharing of knowledge and experience.

At the fourth level, the ISIMM focuses on integrating data interoperability in information systems and OIM a combination level in the OIM where there is more detail and depth in the understanding and preparedness of the organization.

Level five of both (ISIMM & OIM) models is called the Unified level, in ISIMM, this level is reached when data is fully distributed between organizations and collaboration is at an advanced enterprise level.

The interim model that is conceptualized from literature in Figure 4-8, will go through a Delphi expert review and the consensus will determine how many rounds will be required (see Chapter 6 for the number of rounds that occurred in the study).

### KENEILWE MAREMI: 44977883



Figure 4-8: School Management Systems Interoperability Maturity (SMSIM) model

The icons in the centre represents a typical SMS, this can differ from one SMS to another depending on the icons each SMS has to represent different types of features of the system.

The arrow indicates that guidelines will be provided for the entire SMS.

The red dotted bidirectional arrow represents the process of feeding into the OIM & ISIMM to assess the level of interoperability of SMS.

The following section describes each level and further provides guidelines to move to the next level.

**Foundation phase (bottom middle of Figure 4-8):** At this level, a feeder is sent into Information Systems Interoperability Maturity Model (ISIMM) and Organizational Interoperability Maturity Model (OIM) manual or independent level in parallel to assess the required level of interoperability. The outcome of the level serves as a guideline for the next level: Systems are not connected in this interoperability level, and data is shared manually through the post office. Schools are normally unprepared for interoperability to occur, they only communicate via phones among staff members and other schools, there is no interaction with other schools except through communicating via manual means (digital platforms are not used), and there is a limited shared purpose. For schools to get to the adhoc level, they need basic data sharing and general guidelines for interoperability and some electronic means of communicating and sharing information. This can be as simple as using email to communicate with other schools and each other.

At this phase, teachers have increased workloads. There is duplication of work as marks and administrative tasks are done manually (no digital technologies), there is a lot of redundancy and errors because of fatigue and human error, and submission of reports to the DoE is always delayed. However, according to literature, most schools, especially in the private sector, have passed this stage, specifically regarding manual communication and data sharing. Interoperability can reduce the burden on staff members of the schools by introducing systems that will improve data quality and enhance efficiency.

**Primary phase (second phase in the middle of** Figure 4-8): At this level, a feeder is sent into **ISIMM and OIM Adhoc level** in parallel to assess the required level of interoperability, the outcome of the level serves as a guideline for the next level. In this level of interoperability, the school has separate reporting lines of responsibility, general guidelines to achieve

### KENEILWE MAREMI: 44977883

interoperability, and a shared purpose with other schools, there is basic data sharing, applications and databases are separate, data is not shared with other schools, and data is exchanged through systems in a point-to-point manner. For schools to get to the collaborative level, they need to have a shared logical data model, basic collaboration with other schools, and general doctrine to support interoperability and some experience in interoperability.

Schools in this level of interoperability have improved decimally compared to the schools in the foundation phase of interoperability. However, the improvement is minimal, and a lot of work must be implemented to realise the full benefits of interoperability. It is observed that, at this level, school administrator is still burdened with entering data into the systems of the schools as databases and applications are separate, and data is shared in a point-to-point manner. There is limited access to system and only the system administrator can enter data into the system, this opens room for errors, duplicates, redundancy and compromising the integrity of the data between the back-and-forth communication of the systems. By using a SMSIM model as a guideline model to help improve SMS, the requirements of the next level will be obtained and with slow progress, this type of school will run smoothly.

**Intermediary phase:** At this level, a feeder is sent into **ISIMM and OIM collaborative level** parallelly to assess the required level of interoperability, the outcome of the level serves as a guideline for the next level: In the collaborative level of interoperability. There are broader connections to legacy systems. Although minimal standard functions exist, applications and databases are still separate at this level. There is basic data collaboration with other schools, and data is still not shared. Schools can share communications and knowledge about specific topics with other schools. The general doctrine is put in place to support interoperability and some experience, reporting lines are still separate, and the school shares goals, purpose, and value systems with other schools. Schools must exchange data between independent applications for getting to the integrative level.

Unfortunately, legacy systems are outdated computing software and hardware that are still being used. These systems meet the needs they were originally designed for but don't allow for growth. Although these systems present a major investment and, for the most part, they cannot be abandoned. Interoperability connects these types of legacy systems with the new systems. With this advantage of interoperability, information is never lost and there is room for growth. However, most of the disadvantages of SMS are still not entirely solved in this phase as data is still not shared and there are still silos in the schools even though some communication between schools can occur. Schools in this phase must work on getting to the next phase to fully realise the benefits of interoperability. Many schools in South Africa are in this phase of interoperability, and they know the benefit of interoperability and measures are put in place to support interoperability. However, data is still not shared, and applications and databases are still separate. Unfortunately, there is still too much strain on the staff members of the schools to enter data into different systems.

When interoperability is applied correctly in the school, there should be seamless information exchange between the systems, by a click of a button, other departments, such as the finance department and administrative departments, must be able to retrieve information without the need to re-enter what was already entered by another staff member from another department. Paid SMSs can do this to some degree. Ed-admin, which was mentioned by one of the experts during the round 1 questionnaire of the Delphi study, has 40 different modules which can be accessed from anywhere at any time. To some degree, thesis SMS is interoperable, it is also able to integrate with other systems. However, it, unfortunately, still works in silos with other schools and other applications not included in the system. Also, there is no evidence of such an SMS integrating with legacy systems.

Secondary phase: At this level, a feeder is sent into ISIMM and OIM Integrative or combined level in parallel to assess the required level of interoperability. The outcome of the level serves as a guideline for the next level. At this level of interoperability, schools need to have detailed policies & experience in using digital technologies as well as the ability to communicate, sharing knowledge, and interact with other schools through the use of digital platforms and SMS. On a technical level, data is shared and exchanged between independent applications, and systems are integrated between organizations. To get to the unified level, they need to have interoperability completely implemented in the school, both on a technical level and the human activity level.

Schools that qualify to be on this level have implemented interoperability effectively, meaning there are systems integrations from one organization to another. Schools can communicate and seamlessly share information without compromising security and privacy. However, most schools have not reached this stage of interoperability. To get to the next level of

interoperability, schools must be completely interoperable with other schools on a day-to-day basis.

**Tertiary phase:** At this level, a feeder is sent into **ISIMM and OIM's unified level** parallelly to assess the required level of interoperability. The outcome of the level serves as a guideline for the next level. At this level, schools have a shared understanding with other schools, and are completely interoperable with other schools on a day-to-day basis. In addition, the school's data is fully distributed between organizations with a common interpretation on a technical level. At this level, schools are continuously interoperating, data and applications are fully shared, processes are completely automated, and front and back-office systems are fully interoperable. This is the last level of interoperability, and schools would have reached the ultimate level of interoperability by getting to this level. This level usually is easy to achieve for schools of the same kind, for example, a school that is branched out into different arears.

Schools needs to be interoperable to promote data sharing, however there must be compliance with the PoPi Act. When a learner changes schools, they must provide a report card or transfer letter to be admitted. If schools were interoperable, the report card could be verified by clicking a button through their SMS, and to transfer a learner's history from one school to another, the learner or parent (if the learner is underage) would need to provide consent. Also application to a new school requires documentation such as birth certificates, Immunisation etc., this information can be accessed through the SMS as it is required learner admission (Education, 2021). Schools are responsible for protecting the privacy of children's information (Elizabeth, 2021). POPIA provides that the personal information of children may only be processed in limited circumstances and even then, should never be processed unless sufficient guarantees are provided for to ensure that the processing does not adversely affect the individual privacy of the child (Rakhee & Era, 2021). People do not need to be interoperable however they will be affected by the interoperable school and or system therefore, they need to be skilled on how to utilise these systems and how to protect the school, parents and the child's personal information by ensuring compliance with the PoPi act.

The figure below provides a more refined interim SMSIM model to show the simplicity of how the model feeds into each level to assess interoperability and provide guidelines required for the next level of interoperability. This figure also shows what is assessed at each level:

# KENEILWE MAREMI: 44977883



Figure 4-9: Refined Interim SMSIM model

# 4.8 KEY COMPONENTS OF THE CHAPTER

The following section shows the key components from this chapter that were essential to consider for the model, the essential components in the table below are the two chosen models that were combined in Section 4.6 to conceptualize a SMSIM model:

Components	Section	Key Focus Area	Author
Interoperability	4.2	By defining and explaining interoperability, this component allows a broad understanding when conceptualizing SMSIM model	O'Connor (2017), Sjarov et al. (2021), Elkhodr et al. (2016), Zeid et al. (2019), Kirpes et al. (2019), Jaleel et al. (2020).
Organizational Interoperability Maturity Model (OIM)	4.3.3	This model focuses on the organization and the human aspect in the organization, this is essential as it balances the conceptualization of SMSIM model.	Gürdür and Asplund (2018), Sjarov et al. (2021), Jabin et al. (2019), McDaniel (2020).
Information System Interoperability Maturity Model (ISIMM)	4.3.6	This model is key as it focuses on the technology aspect of the organization, this completes the SMSIM model	Anggoro et al. (2018), Almeida Prado Cestari et al. (2020), Ostadzadeh et al. (2015).
Combining OIM & ISIMM	4.6	By combining OIM & ISIMM model, a clear picture can be observed to conceptualize the model	

## 4.9 SUMMARY

This chapter briefly introduced interoperability and described Maturity Models. It then listed and discussed different existing Interoperability Maturity Models. Two (2) models were chosen from the identified models to help conceptualize the theoretical model of this study, namely OIM and ISIMM. These were the most important models to consider. This chapter further combines the chosen models to ensure they are ready for data collection and analysis that will form Chapter 6. Finally, this chapter conceptualizes a SMSIM model, which will be improved and refined after data collection and data analysis.

# KENEILWE MAREMI: 44977883



Figure 4-10: Chapter overview

From the above figure, it is evident that this chapter concluded Phase 1 of the study.

# CHAPTER 5: RESEARCH METHODOLOGY

### 5.1 INTRODUCTION

This chapter describes the methodology that underpins the study. In order to study a phenomenon of interest, researchers gather, examine and deduce evidence (Leedy & Ormrod, 2021). For a problem to be solved, appropriate data must be collected (Leedy & Ormrod, 2021). The research methodology guides the research process in terms of designing, determining effort, and achieving results. In many cases, the research statement determines the research questions as well as the subject discipline (Herselman, 2011). Research methodology refers to how research should be done. It contains an understanding of the research question and a set of assumptions that guide the methods of investigation (Melnikovas, 2018). As an integral part of a dissertation or thesis, a methodology ensures consistency between selected tools, techniques, and underlying philosophy (Melnikovas, 2018).

For the purpose of this study, Leedy and Ormrod (2021) definition will be used as they emphasize that the research methodology is an essential part of the research that guides how the research will be undertaken and conducted.

This is a qualitative exploratory research study with exploratory Delphi expert reviews, and the research design and methodology used in this study are outlined in this chapter. This chapter details the research process adopted for this study and explains the data collection process undertaken by the researcher as well as reasons for choosing each method. The purpose of this study is achieved by following the research process as illustrated in Figure 5-1.

### KENEILWE MAREMI: 44977883



Figure 5-1: Research Process to conduct the research

The completion of each phase allows for the next phase to start.

- Phase 1: The literature review aimed to answer the second SRQ by exploring existing Interoperability Maturity Models (Chapter 4, Section 4.3). The identified Interoperability Maturity Models (IMM) assisted in creating a combined Interoperability Maturity Model (OIM & ISMIM) in Section 4.6 that can be used to assess the level of interoperability of SMS in schools of South Africa, thus achieving the outcome of the first phase.
- Phase 2: Delphi Method consists of a series of questionnaires distributed to selected experts under a survey coordinator's supervision (Hirschhorn, 2019). The survey coordinator gives participants anonymous feedback on the panellists' responses after the first round of questions and before any new ones are asked. Upon reviewing this feedback, participants may reconsider their opinions when responding to subsequent questionnaires (Hirschhorn, 2019). This technique, which incorporates surveys and monitored opinion responses in a safe, anonymous setting, is a potent tool for exposing and articulating differing viewpoints and advancing knowledge (Hirschhorn, 2019). The literature study in phase 1 (Chapters 2 and 3) provided a combined Interoperability Maturity Model (OIM & ISMIM) that can be used to assess and conceptualize the School Management Systems Interoperability Maturity (SMSIM) model for schools in South Africa. Experts will go through iterations of questions to validate the model until

a consensus is reached. This will be done to ensure that the final output is credible, accurate and relevant. At the end of round 1, the results reaffirmed the problem statement of the study and the need to conceptualize a SMSIM model that can improve SMS in South Africa as it can assist schools to measure the interoperability maturity level of their SMS and to use the guidelines to improve their SMS.

• Phase 3: The result of the Delphi Method in phase 2, using the combined model in phase 1, will assist in verifying if the interim SMSIM model (Chapter 2, 3 and 4) is validated for the purpose of the study, thus conceptualizing a final SMSIM model for schools to use in SA (Chapter 4). Thus, answering the main research question. This model will be validated through expert reviews as discussed in Chapter 6, and conclusions, as well as recommendations, will be provided in Chapter 7.

Phase 2 of Figure 5-1 is discussed in more detail below and forms the basis of this chapter.

# 5.2 OVERVIEW OF THE RESEARCH DESIGN

Saunders and Lewis (2017) research onion is shown in Figures 5-2 and 5-3. Layers are used to depict the research design. The path of the investigation is determined by the outer layers, which in turn influences the interior levels (Saunders & Lewis, 2017).

The research onion below illustrates the different research designs which exist. For this study, all the research designs/ techniques will be discussed in detail and only the design which will be chosen by the researcher will be justified.



Figure 5-2:Research Process to conduct the research

According to Melnikovas (2018), the research onion has six main layers::

- **Research philosophy in a study** describes the reality, knowledge, and values of the study by defining ontology what exists, epistemology how we know what exists, and axiology what is true.
- Approach to theory development usually includes: deduction which is implied by the research philosophy at the previous level an existing theory is used as a starting point, followed by a question or hypothesis and data collection for confirmation or rejection; induction observation and data collection are followed by description and analysis to develop a theory; abduction observation of empirical phenomena is followed by research that draws a best guess or conclusion from the available data. Inductive approaches are used to develop theories in fields with little research, whereas deductive approaches are commonly used to analyze existing theories. A surprising fact is usually used as the starting point for the abductive approach, followed by induction and deduction to determine the most likely explanation.
- **Methodological choice** determines whether quantitative and qualitative methods will be used or a combination of both.

- A strategy involves planning how data will be collected and analyzed: case studies, experiments, action research, surveys, narrative inquiries, archival research, ethnographies, grounded theories, etc.
- **Time horizons** involve planning how data will be collected and analyzed: experiments, surveys, archival research, case studies, ethnographies, action research, grounded theories, narrative inquiries, etc.

The Figure below summarizes the research onion with the identified components at each level that were applied in this study.



Figure 5-3: Adapted Research Onion applicable to this study ((Saunders et al., 2016))

The specific chosen aspects applicable to this study, in each layer will now be further discussed.

## 5.3 **RESEARCH PHILOSOPHY**

The job of a researcher is to understand and articulate beliefs about the nature of reality, what can be known about it, and how to gain that knowledge and this is also known as research paradigms (Rehman & Alharthi, 2016; Saunders *et al.*, 2016:p130). Paradigms are theoretical frameworks containing assumptions about the methodology, ontology, epistemology, and

methods. Basically, it is a way to understand and study the world as it is. Four types of research paradigm exist and include (Guba & Lincoln, 1994; Wahyuni, 2012; Hussain, Elyas & Nasseef, 2013; Rehman & Alharthi, 2016; Saunders *et al.*, 2016:p177; 178; Creswell & Creswell, 2018:p54; 58):

- **Positivism.** Because of the writings of French philosopher Auguste Comte, the area of philosophy known as positivism rose to popularity in the early nineteenth century. A Positivist believes that reality is independent of humans. The positivist strives to understand the social world in the same way as the natural world. A large part of positivist thought is based on experimentation.
- **Critical realism.** A group of twentieth-century authors affiliated with the Institute of Social Research at the University of Frankfurt gave rise to critical theory. Critical theorists hold a historical realist ontological position. The nature of reality cannot be denied, but it has been shaped by cultural, political, and religious factors. Research using the critical method consists of dialogic and dialectical interactions; the researcher works with the subjects in order to change their attitudes towards social systems that are preventing them from meeting their intellectual and social requirements.
- **Interpretivism.** Interpretivism responds to positivism's dominance. Interpretivism rejects the idea that there is a single, verifiable reality. To understand their subjects, researchers put themselves in their subjects' shoes and look through their eyes. The constructivist theory asserts that knowledge and understanding are derived from interpretation rather than assumptions regarding facts.
- **Pragmatism.** A pragmatic approach evolved from using what worked best in each situation. A pragmatic researcher uses a variety of methods to understand existence by focusing on the research problem.

The research philosophy applicable to this study is the interpretive research philosophy. This approach will be used to better understand people and their experiences when they evaluate and validate the conceptualized model and its components. Interpretivism is a philosophy that emphasizes the researchers' need to understand humans' role as social actors (Saunders *et al.*, 2016:p181). Interpretivism takes a particular context as the focus of understanding, and the central belief of this paradigm is that reality is socially constructed (Hiller, 2016:p22; Saunders *et al.*, 2016:p181). As part of the interpretation process, qualitative researchers compare the results with the literature, present the overall results, discuss a personal viewpoint, and state
limitations and future directions (Creswell & Creswell, 2018:p313). Furthermore, it is important to understand the theoretical lens through which the research is conducted, and the section below elaborates on that.

### 5.3.1 Theoretical Lens

This section discusses the theoretical framework that guides the research.

### 5.3.1.1 Institutional Theory

In studying the role that SMS and interoperability can play in supporting the administrative burden for schools, it is vital to comprehend the broader educational context and how this influences its functioning. For this reason, this research makes use of the Institutional Theory. According to Raynard, Johnson and Greenwood (2015), organizations are a collection of controlled activities that operate within a complex network. However, there is growing recognition of the parallels between formal organizational structures and institutions, which has influenced how they are operated. To ensure that the model is in line with the broader educational context, this research uses Institutional theory.

Organizations embed institutional norms to establish industry standards for guiding operations. Furthermore, Institutional Theory believes that external and internal pressures influence how organizations operate (Alkalbani, Deng, Kam & Zhang, 2016). This refers to the notion that organizational structures are designed to reflect and adhere to established norms, while their needs are deferred (Othman, 2016).

Through the Institutional Theory, the researcher could unpack the varying factors contributing to the education ecosystem where SMS needs to function. This theory also assisted the researcher in confronting the issues related to the fragmentation of SMS and how a conceptual model that applies interoperability and maturity levels can be designed to better assist schools in using SMS.

#### 5.3.1.2 Theoretical Grounding

The researcher will further use hermeneutics to translate and interpret the text of the experts, verbal and nonverbal communication. Hermeneutics relates to vocal and non-vocal communication and refers to the interpretation and conversion of the script (typically ancient scriptures) (Friston & Frith, 2015). It is mainly used during data analysis. When deciding when

an interpretation is correct, the main issue of hermeneutics is to focus on evolving measures. This issue is not limited to clarifying ancient scripts (Jardine, 2015). Although it does not offer a method, hermeneutics gives academics a sense of how human understanding functions in the real world (Jardine, 2015).

Following the hermeneutic circle (Klein & Myers, 1999) as explained in Section 5.4, understanding the meaning and importance of individual texts depends on understanding the whole body of relevant literature (built up through understanding individual texts). Therefore, the hermeneutic circle can assist in surveying relevant literature (Boell & Cecez-Kecmanovic, 2010). The hermeneutic circle serves as a tool for evaluating our understanding of a particular text (Warnke, 2011).

The following section discusses the hermeneutic circle, which will be used during data analysis.

## 5.4 THE HERMENEUTIC CIRCLE

Researchers and participants interpret phenomena as part of an iterative process in the continuing hermeneutic circle. The hermeneutic circle helps interpret data and fosters communication between the researcher and participants (Craig, 2017). Additionally, the hermeneutic circle method of analysis enables the researcher to examine the data through a continuous process of assessing the whole and specifics of the information gathered (Harper, 2018). It was crucial to understand the findings from the expert evaluations and how they contributed to this research after the data was collected through the Delphi expert review. hermeneutics was utilized to analyze and interpret the consensus of rounds 1 and 2. With the goal of comprehending multiple components that contribute to the functions of a whole, hermeneutics is grounded in the interpretive paradigm (Herselman and Botha, 2016). Hermeneutics functioned as a valuable analysis approach to gain an in-depth understanding of problems in SMS.

Table 5-1 summarizes the principle for interpretive field research and how it was applied for this study.

Fundamental	Summary of Dringinlas for Interpretive Field Desearch
Principle	Summary of Frinciples for interpretive Field Research
Hermeneutic Circle	This principle was applied throughout the research and mostly in this chapter (6) to gain human understanding by iterating between the interdependent meaning of parts and the whole they form.
Contextualization	The context of this principle focused on the SMSIM model that was conceptualized to address the problems in School Management Systems (SMS).
Interaction Between the Researchers and the Subjects	The interaction of these principles occurred in two (2) phases, the first phase is when the researcher interacted with the literature review and then developed questionnaires to pose to experts, and the second phase is the interaction between experts and the researcher. First, interactions between the experts and the researcher occurred when expert reviews were administered. The results are reflected in section 6.4 of chapter 6.
Abstraction and Generalization	The conceptualized SMSIM model presented in section 6.5.2.2 can be used by Schools in South Africa and can potentially solve problems in SA-SAMS.
Dialogical Reasoning	The theoretical foundation developed by the literature findings in Chapters 3 to 5 and the contributions from the expert reviewers served as the basis for the interpretation in this study (results presented in section 6.5).
Multiple Interpretations	The supervisors' helpful criticism was considered and interpreted to properly document the results that were obtained. The researcher was also sensitive in interpreting the data collected from experts' reviews of round 1 to round 2 Delphi questionnaires.
Suspicion	Multiple sources were used throughout the research to reaffirm and provide context to the problem statement, particularly the literature chapters (3,4,5). In addition, the experts mentioned in sections 5.7.3 of this chapter and 6.3.1 of chapter 6 took part in the study. The findings from iteration 1 of the expert review are presented in section 6.3 of chapter 6, and those from iteration 2 are presented in section 6.5. The information, sources and participants were treated with sensitivity to ensure reliability and credibility.

Table 5-1: Summary of principles for interpretive field research

Source: Adapted from (Klein & Myers, 1999)

Hermeneutics was also applied during the data analysis to support thematic analysis (Section 5.9). The section below outlines the approach to theory development.

## 5.5 APPROACH TO THEORY DEVELOPMENT

Two main choices can be used by the researcher, namely inductive or deductive approaches (Johnson-Laird, 2010; Matzel & Sauce, 2017).

- Inductive reasoning is a logical procedure whereby several premises, which are all generally accepted to be true or found to be true, are integrated to arrive at a conclusion or to provide evidence for the validity of a conclusion.
- Inductive thinking differs from deductive reasoning in that the degree of certainty is dependent on the strength (or consistency) of the evidence, whereas the conclusion of a deductive inference is certain, the truth of an inductive inference's conclusion is only probable.
- Deductive reasoning is the mental process of making logical inferences. It is just one type of reasoning.

The approach chosen by the researcher for this study is the inductive reasoning because this approach will allow essential themes and patterns to be selected from data during the qualitative data collection process. The information provided by people are examples or incidences from where conclusions can be drawn (Leedy & Ormrod, 2015). The following section presents the research approach chosen for the study.

### 5.6 THE RESEARCH APPROACH

Research approaches are strategies and tactics that cover everything from general hypotheses to specific techniques for gathering, analyzing, and interpreting data. This plan involves several decisions that a researcher should take to make sense of the results (Creswell & Creswell, 2018:p51).

There are mainly three different types of research approaches (Creswell & Creswell, 2018:p51):

• Qualitative research: Is a method for examining and determining the relevancy that different people or groups assign to social or human problems. The research process entails developing questions and collecting data that typically takes place in the participant's environment, inductive analysis that progresses from specifics to broad

themes, and the researcher's interpretation of the significance of the data. As a result, the final report's structure is adaptable.

- **Quantitative research:** Is a method for evaluating the relationships between variables to test objective theories. So that numbered data may be examined using statistical techniques, these variables can be measured, often using instruments. An introduction, literature and theory, methodology, findings, and discussions are all included in the final written version of the report.
- **Mixed methods research:** Is a procedure for investigating that includes acquiring both quantitative and qualitative data, fusing the two, and using a variety of designs that may contain theoretical underpinnings and philosophical presumptions. The underlying assumption of this kind of research is that using both qualitative and quantitative methods combined results in a deeper understanding of a studied subject than using simply one or the other.

The research approach applicable to this study is the qualitative approach, as it can help explain everyday life in new and intriguing ways (Mayan, 2016:p11). Qualitative inquiry is inductive, interpretive, and primarily collected in natural settings (Creswell, 2007:p53; Mayan, 2016:p12). Qualitative researchers try to understand or interpret the meaning people create from their experiences by studying naturally occurring phenomena (Mayan, 2016:p13). The Delphi technique is a qualitative approach that can be used in a qualitative exploratory study (Al-Busaidi, 2014; Habibi, Sarafrazi & Izadyar, 2014). The qualitative Delphi technique was adapted for the study as presented in Section 5.7 as a research strategy and used during data collection.

To help people, understand their worlds collectively, qualitative researchers love living and learning with them. Humanness is, therefore a crucial component of qualitative research (Mayan, 2016:p13). A diversity of approaches to improve the cogency and trustworthiness hereafter the credibility of the data they gather is used by qualitative researchers (Leedy & Ormrod, 2015), which is why this approach is best suited for this research study.

An interpretative paradigm, which stresses personal experiences and their significance to a particular person, is what distinguishes qualitative research. Therefore, the researcher's subjective perceptions of the circumstance is a key factor in the study's findings (Starman, 2013). Qualitative research also stresses an individual's viewpoint on the situation, process, and

relationships under investigation using an idiographic approach (Starman, 2013). The specific data collection instruments chosen to conceptualize the model further were coupled with this approach.

### 5.7 RESEARCH STRATEGIES

A research strategy outlines how the researcher intends to answer the research questions (Lewis & Saunders, 2018). The following is a brief description of the different types of strategies, as described by Lewis and Saunders (2018) and Creswell (2007:p53-73):

- In **case studies**, research questions are addressed within a real-life context, using various sources of evidence (data).
- In **ethnographic research**, first-hand experiences in the field are used to describe and interpret the social world.
- In **experiments**, a hypothesis is defined, samples are selected from known populations, they are assigned to different experimental groups, some variables are changed, and others are controlled.
- A grounded theory develops a theory based on observations and interviews using inductive inquiry.
- In **archival research**, administrative records and documents are analyzed as primary data sources.
- **Surveys** are structured research strategies that collect data from a large population through structured interviews or questionnaires; and
- An **action research** strategy aims to manage change and involves close collaboration between practitioners and researchers.

According to Habibi *et al.* (2014) a Delphi technique is a group method for acquiring knowledge and making qualitative decisions. Furthermore, Habibi *et al.* (2014) state that the Delphi approach is appropriate for exploratory qualitative research that seeks to understand the nature and essential components of occurrences. Therefore, the Delphi method will be applied for the purpose of this study. Delphi derives its name from the antiquity of the ancient oracle of Delphi, which provided guidance for Greeks and Romans seeking answers to crucial questions in their lives (Avella, 2016; Beiderbeck, Frevel, Heiko, Schmidt & Schweitzer, 2021).

A wide range of disciplines has utilized the Delphi method since it was developed and has undergone several expansions since its creation. Thus, this technique is considered an effective way to obtain expert consensus about a complex problem or forecast the future of a given field (Massaroli, Martini, Lino, Spenassato & Massaroli, 2018).

A Delphi method is a technique that, when applied, allows a team of experts (i.e., the Delphi Panel) to gather structured and systematic information about a particular subject through the use of questionnaires (Flostrand, 2017). When applied to communication processes, Delphi is a technique that helps group members effectively work together to provide a consensus on a problem (Turoff, 1970; Mukherjee, Sutherland, Khan, Berger, Schmitz, Dahdouh-Guebas & Koedam, 2014; Asghari, Nassiri, Monazzam, Golbabaei, Arabalibeik, Shamsipour & Allahverdy, 2017). Even when experts are not located in the same geographic area, the Delphi method enables them to provide insight and opinions (Dalkey & Helmer, 1963; Yeh, Van Hoof & Fischer, 2016).

There are many circumstances in which the Delphi method is necessary, including when group dynamics prevent effective communication (for example, time differences, distance, and personality conflicts) (Beiderbeck, Frevel, Von Der Gracht, Schmidt & Schweitzer, 2021). Experts who consented to participate are asked to complete the questionnaires in iterative rounds. Before each new round begins, the panel receives the responses from the previous round. After reviewing the answers provided by the other panellists, each participant can re-evaluate their original responses. Typically, 3 or 4 rounds are required to reach a consensus or stabilize the questionnaire answers (Linstone & Turoff, 1975; Yeh *et al.*, 2016). In this study, two iterations were applied to reach a consensus among experts when evaluating the model's components and validating the model.

Therefore, the Delphi method lends itself to the effective validation of subject matter integrity without physical access to domain experts, and it is suitable for use when one needs to do so. With the COVID-19 pandemic spreading, it became impossible to access schools or any research organization. Moreover, targeted participants and their staff tried to adjust to the virus' impact, making it even harder to gain access to participants for research (Daniel, 2020). Restoring normality is not a simple one-time process, schools, teachers, and students had to seek flexible methods to manage the difficulties caused by COVID-19's interruptions to learning (Daniel, 2020). As a result, the Delphi method was the best solution for the purpose

of this study, as it focused only on the experts to validate the purpose of this study. Skulmoski, Hartman and Krahn (2007) further allude that for graduate students completing master's and PhD research, the Delphi method is an attractive option to use when collecting data. Furthermore, the table below presents studies that applied the Delphi technique in the education field, this table reaffirms the choice of using the Delphi technique as the research strategy in this study.

Article	Purpose	Year	Author
Curriculum development through Delphi	The application of the Delphi method for curriculum creation in higher	1978	(Reeves &
	education is discussed in this article.		Jauch, 1978)
Validating Teaching Competencies for	The teaching abilities of faculty members in higher education were assessed in	1995	(Smith &
Faculty Members in Higher Education: A	this study using the Delphi Method.		Simpson,
National Study Using the Delphi Method			1995)
The Delphi Method for Graduate Research	Used to successfully explore novel ideas both inside and beyond the body of	2007	(Skulmoski et
	knowledge for information systems in the University of Calgary program.		al., 2007)
Using the Delphi Technique in Educational	The Delphi Technique is examined in this article along with its advantages and	2011	(Nworie, 2011)
Technology Research	potential uses for educational technology researchers.		
Development of ICT Competency Standard	In order to create a legitimate and trustworthy ICT (Information &	2013	(Fong, Ch'ng
Using the Delphi Technique	Communication Technology) competency standard for teachers, the goal of this		& Por, 2013)
	article is to explain the use of the Delphi technique in the research design.		
Using the Delphi technique to support	This paper's goal is to advance the body of knowledge regarding the use of the	2015	(Sitlington &
curriculum development	Delphi technique for updating business course curricula by providing an		Coetzer, 2015)
	analysis of its use to support curriculum development.		
The Delphi method: characterization and	The Delphi method is introduced in this article, along with its uses in education.	2018	(Marques &
potentialities for educational research			De Freitas,
			2018)
Technological aspect factors of E-learning	This paper's goal is to review the technological aspects of eLearning readiness	2019	(Al-Araibi,
readiness in higher education institutions:	using the Delphi technique and to create a preliminary model for these features		Mahrin &
Delphi technique	in higher education institutions.		Yusoff, 2019)
Adopting a modified Delphi technique for	In order to successfully transition academic endeavors during and beyond the	2020	(Rajhans,
revisiting the curriculum: a useful approach	COVID era, this study will describe a participatory qualitative research project		Rege, Memon
during the COVID-19 pandemic	employing the Delphi consensus technique and Internet technologies.		

# Table 5-2: Studies that applied the Delphi Technique

Article	Purpose	Year	Author
			& Shinde, 2020)
A Research Study by Delphi Technique in School Counselling	The Delphi technique, which a researcher can use to study school counseling, is examined in this conceptual article, along with the many processes that implement the technique to meet the study's goals.	2020	(Sivalingam, 2020)
Developing an empathy educational model (EEM) for undergraduate nursing students: A Delphi Technique	The Delphi technique was employed in this study to create an instructional model that teaches empathy to undergraduate nursing students.	2021	(Zhu, Yang, Zhang & Chen, 2021)
Application of the Delphi Technique to Determine the Technological Competencies of a Faculty Member	By using the technology of independent expert assessments—the Delphi method—the study aims to ascertain the current level of manifestation of technological competencies as well as generalized prospects for development and improvement of the identified level within the chosen group of freelance teachers.	2021	(Sosnytskyi, Sikorskyi, Bezborodykh, Morozova & Moroz, 2021)
The current situation of information literacy education and curriculum design in Pakistan: a discovery using the Delphi method	The current study used the Delphi procedure in developing nations as a unique research lens to examine information literacy education. The study's main goal is to provide a framework for information literacy in higher education.	2021	(Batool, Ur Rehman & Sulehri, 2021)
Exploring hospitality graduates' competencies in Malaysia for future employability using Delphi method: a study of Competency-Based Education	This study examines the hospitality competences required by higher education graduates for their employability in the sector in the future using competency-based education.	2022	(Shariff & Abd Razak, 2022)

It is evident from the table above that Delphi has been applied in the education sector, more specifically when conducting exploratory qualitative research. Section 5.7.1 discusses the advantages of the Delphi technique.

### 5.7.1 Advantages of the Delphi technique

The following presents advantages of the Delphi technique (Lilja, Laakso & Palomäki, 2011; Avella, 2016; Asghari *et al.*, 2017; Kallia & Sentance, 2017; Massaroli *et al.*, 2018):

- **Consensus:** In cases when there is uncertainty or where there is no clear cause, consensus can be reached. In this study the iterations continued until a consensus was reached among experts.
- Flexibility and simplicity: Designing a Delphi study can be quite simple and flexible, and evaluating the conceptualized components of the model is straightforward, and the design will be flexible enough to allow for valuable inputs to refine the final model.
- **Knowledge sharing:** One aspect of Delphi designs that isn't frequently discussed is on how diverse disciplines can exchange knowledge, generate innovative thinking, and advance the understanding of other panel members by applying to the panel's objective. In this study, the panel consisted of experts from various backgrounds (IT network specialists, administrators, principals, and teachers). This allows them to expand their knowledge base on SMS.
- **Cost effectiveness:** Following a Delphi design is quite cost-effective, unless the researcher wishes to incorporate an instrument that involves a "charge for usage." The main expense is time, both that of the researcher and that of the participants (who are all volunteers). Other than time, the study's specific costs are not significant. It utilizes currently existing resources and tools. This technique was valuable during data collection as access to schools was prevented due to CODIV-19 regulations and restrictions.
- **Freedom of expression:** The anonymity that is necessary for all Delphi panels directly results in freedom of expression. It provides panellists with a great deal of freedom in how they express their ideas and may allow them to share various viewpoints with others without worrying about being attacked. Therefore, the experts in this study had to express their views anonymously.
- Anonymity: This eliminates participants who are afraid to express their ideas and positions. Knowing who said what prevents other panellists from openly evaluating each other's comments. Experts could not judge each other as anonymity protected them in using this technique.

- Ease of communications: The Delphi Method's early iterations relied heavily on physical artifacts (papers that had to be typed, mailed, or faxed), which significantly reduced panel operations' efficiency and made it impossible for participants from other countries to participate. Although the accessibility and simplicity of electronic communications have greatly reduced these problems, they have also brought forth new ones, particularly for researchers. This is especially true when it comes to protecting participant anonymity. Consequently, all interactions between the researcher and each panellist should remain private. In this study, these communications did occur on a one-to-one basis.
- Membership variations: It is not necessary for the panel to consist of the same number of members for every round of the study. It can vary in several ways, including, but not limited to, members dropping out or skipping a round and coming back to it later or skipping the first round but joining the next. Maintaining the number of groups, ensuring that membership is balanced among the groups, and ensuring that the membership standards are upheld are all important considerations for researchers. In this study, the researcher did see that variations occurred as in the last iteration, the number declined, although valuable insights were collected.
- Lack of geographical limitations based on electronic communications: Geographical boundaries that affected the design of the physical artifact in Delphi investigations have almost entirely been eliminated by electronic communications. Therefore, researchers are no longer constrained by geography when seeking out participants or tackling pressing global concerns. In this study, the experts were not all from one province, and this allowed for a broader reach.

It is evident from the above advantages of the Delphi Technique that it is straightforward and effective at data collection. It also assures accuracy and does not need to physically meet participants to validate the outcome of this study. Similarly, the Delphi technique has disadvantages that need to be considered to have an effective model. The following section presents the disadvantages of the Delphi Technique.

### 5.7.2 Disadvantages of the Delphi method

Delphi has weaknesses like most research designs do, but the researcher causes those errors, not the design, as is the case with most research designs. Deficiencies in the researcher or

panellists can cause flaws to manifest. The following are some typical examples of the disadvantages when using the Delphi method (Shariff, 2015; Kallia & Sentance, 2017; Fink-Hafner, Dagen, Doušak, Novak & Hafner-Fink, 2019; Singh, Nicely, Cai & Day, 2021):

- **Researcher bias:** Because of the researcher's dominant strength and impact, bias may even enter the process unintentionally. The formulation of the question(s) and the selection of participants can be used to support the researcher's positions. Another manner in which a researcher can show bias is by choosing panel members who have predetermined opinions on the issue. Researcher bias was addressed in this study as the supervisors assisted in the selection of experts to ensure the trustworthiness of the data.
- **Researcher shortcomings:** A possible drawback of researchers forcing their views on respondents is evident, particularly in the case of a modified design where Round 1's literature review was created by the researcher. Another example would be a bad summary of the panellists' comments. The earlier discussion of modified Delphi designs stated that no matter how initiating material was generated, panel members should never be prevented from adding to the generated alternatives. Researchers need to recognize that their role is not one of the contributors but rather of facilitator. In this study, the researcher did facilitate the process and did not contribute to the study; and
- **Panel member anonymity and petulance:** A panel member could withdraw at any time and for any reason. In this study, all ten participants completed the two Delphi iterations.

These disadvantages presented what may go wrong when implementing the Delphi method. It is thus important to consider these disadvantages when selecting the experts and when verifying and analyzing the feedback from responses. Therefore, the next section discusses how to select experts.

### 5.7.3 Selecting experts

In qualitative research, sampling is when specific data sources are selected and gathered to meet the research objectives (Gentles, Charles, Ploeg & Mckibbon, 2015). Purposive sampling was used in this study as one selects purposive sampling for a specific purpose (Leedy & Ormrod, 2021). Expertise is the capacity for competence, or the exertion required to display extraordinarily high levels of performance on a specific task or within a given topic. An individual who reaches this level is referred to as an expert (Bourne Jr, Kole & Healy, 2014).

A list of the expert's knowledge, skills, desires, plans, and accomplishments is necessary for a description of competence (Bourne Jr *et al.*, 2014). Furthermore, expert opinions can assist in highlighting any vulnerabilities that might exist with respect to the topic being evaluated (Jansen & Hak, 2005; Jahoor, 2019).

There are several recommendations for choosing an expert group. Even though every circumstance for choosing from the pool of specialists is different (Manakandan, Rosnah, Mohd & Priya, 2017; Engelman, Fuller & Steer, 2018). Rowe and Wright (2001) summarize the following principles for choosing expert opinion in applications of Delphi (Jolson & Rossow, 1971; Rowe, Wright & Bolger, 1991; Rowe & Wright, 2001; Grime & Wright, 2016; Manakandan *et al.*, 2017; Hohmann, Cote & Brand, 2018; Ahmad & Wong, 2019):

- Use between 5 and 20 experts.
- Use experts with relevant fields of expertise.
- Make use of diverse experts.
- Provide the panel's mean or median estimate along with each panellist's justifications for their estimations for Delphi feedback.
- Delphi polling should continue until the results become stable. Usually, two formal rounds are sufficient.
- Calculate the final projection by adding up and evenly distributing the expert forecasts.

In this study, the researcher ensured that these principles guided the selection of experts as displayed in Table 5-3 below.

For a Delphi panel, a qualified expert should be (Lilja et al., 2011):

- The best in her/his field of expertise in a technical or scientific discipline.
- Has a broad interest in learning about everything, not only what is in their field but also what is in the world.
- Able to recognize linkages between many scientific domains and between existing and future development on a national and worldwide level.
- Able to look at problems from a variety of angles, even some that are unfamiliar and risky.
- A desire to produce new things.

Table 5-3 presents experts who were selected to participate in the study, the selection criteria were guided by Section 5.7.3 above.

Domain/Expert Reviewer	Years of experience	Role	Field of Expertise
Expert Reviewer 1	17	IT Manager and Support	ICT in Education
Expert Reviewer 2	30	IT Manager	ICT in Education
Expert Reviewer 3	12	Industry experts focusing on interoperability	Interoperability
Expert Reviewer 4	13	ICT manager working with various schools and their SMS	ICT4D, Interoperability, SMS
Expert Reviewer 5	30	Deputy principal	School Management Systems (SMS)
Expert Reviewer 6	11	IT manager of school system	ICT in Education
Expert Reviewer 7	16	ICT Coordinator	School Management Systems (SMS)
Expert Reviewer 8	12	Interoperability manager	Interoperability
Expert Reviewer 9	15	System Administrator	School Management Systems (SMS)
Expert Reviewr 10	20	ICT Coordinator	ICT4D, Interoperability

Table 5-3: Expert Reviewers

The study was originally supposed to include 21 participants, which included teachers, ICT coordinators, principals, system administrators, and experts, but only ten (10) experts participated in the study due to a lack of access to schools. The COVID-19 pandemic and the possibility of infection from face-to-face contact have contributed to the lack of participation. Despite lifting restrictions and removing masks, schools are still recovering from the effects of the COVID-19 pandemic, so the Delphi method was used to conduct the study. Ten (10) experts validated the model's quality, efficiency, and relevance. Following the literature review, the interim model was adapted (Figure 4-7) using empirical evidence to provide the final model.

In this study the researcher did address these suggestions during the selection of the experts. The above section is very important for the success of the study, only experts will participate in the study and thus, how they are selected is vital. The following section lists and summaries different Delphi Designs.

## 5.7.4 Different Delphi designs

The Delphi method is included in the broad category of "consensus development procedures," which is itself included in the broad category of action research methodologies (Vernon, 2009; Avella, 2016). There are two main Delphi designs, the conventional Delphi, and the modified Delphi (Kallia & Sentance, 2017). Table 5-4 below summarizes other Delphi designs as per the literature.

Form of Delphi	Description	References
Exploratory- /Conventional-/ Classical Delphi	To gather trustworthy data about potential trends relating to a certain problem or topic, a panel of experts is assembled.	(Ekström, 2020), (De Soria, Durán, Morrás, González & Varela, 2018).
Modified Delphi	The conventional Delphi method is altered, for instance, by enabling some experts to debate in- person even at the end of the process, recognizing the importance of in-person interactions for exchanging ideas, identifying the causes of differences, and resolving ambiguities.	(Griffey, Schneider, Adler, Capp, Carpenter, Farmer, Groner, Hodkins, Mccammon & Powell, 2020), (Bleijlevens, Wagner, Capezuti, Hamers & Workgroup, 2016).
Spatial Delphi	This is valid for decisions involving spatial location that are made after consultations. Expert contributions are mapped geographically, and the convergence of their views is denoted using basic geometric forms (circles or rectangles).	(Di Zio, 2018), (Di Zio & Pacinelli, 2011).
Policy-/ Focus-/ Decision Delphi	Used to examine several policy solutions with the most significant benefits and drawbacks for each policy resolution based on the assessments, viewpoints, and experiences of experts.	(Kattirtzi & Winskel, 2020), (Paraskevas & Saunders, 2012).
Real-time- /Consensus Conference-/ Normative Delphi	This is a system for asynchronous online conferences that is computer-mediated and ensures anonymity.	(Linstone & Turoff, 2011), (Yousuf, 2007).
E-Delphi (eDelphi), Technological-/ Online- / Argument Delphi	A modified e-Delphi survey is conducted online.	(Taylor, Feltbower, Aslam, Raine, Whelan & Gibson, 2016), (Hasson & Keeney, 2011).
Disaggregating Policy Delphi	This format assumes that expert communication cannot lead to consensus and will thus elicit several schools of thought because experts will	(Tapio, Paloniemi, Varho & Vinnari, 2011).

Table 5-4: Summary	of Delphi	Designs	(adapted	from Fisher.	Erasmus &	Vilioen.	2020)
1 abic 5 1. Summer y	of Deipin	Designs	(addpica	from i usitor,	Li asinino a	, 11/0011,	2020)

Form of Delphi	Description	References
	congregate around the competing claims that	
	receive the most support.	
Problem Solving	This kind is employed for group judgment by	(Linstone & Turoff,
Delphi	gathering participant rankings or paired	2011).
	comparisons.	
The Fuzzy Delphi	In contrast to traditional deterministic approaches,	(Manoliadis, 2018).
Method (FDM)	which express information as a single value, fuzzy	
	numbers are used instead.	
Ranking -type	The most applicable for IS research is probably	(Kobus & Westner,
Delphi	Ranking-type Delphi. It focuses on typical areas	2016).
	of interest in information systems research, such	
	as the identification and ranking of Critical	
	Success Factors (CSF), the breakdown of research	
	frameworks, and the prioritizing of selection	
	criteria.	

Fisher, Erasmus and Viljoen (2020) tabled different Delphi forms for this study. The exploratory Delphi design will be explained in detail in the section below.

## 5.7.5 Exploratory Delphi

An exploratory Delphi attempts to predict a future or series of events (Battersby, 1982). This method is a forum for establishing facts about a specific situation or topic (Franklin & Hart, 2007; Fisher et al., 2020). According to Dalkey and Helmer (1963), exploratory Delphi is a method of obtaining consensus from an expert panel by means of questionnaires and controlled feedback. A consensus is generally reached after three iterations in this design. It is possible, however, to reach consensus, theoretical saturation, or to uncover sufficient information with fewer rounds of analysis in a homogenous sample (Skulmoski et al., 2007; Fisher et al., 2020). Stopping the rounds at the right time is crucial. Results may not be meaningful if they are terminated too early, while too many rounds can lead to sample fatigue and resource waste (Hasson, Keeney & Mckenna, 2000; Fisher et al., 2020). Whenever a subsequent round is necessary, there should be a good reason for it (Landeta & Barrutia, 2011). Regardless of the nature of the response, the number of rounds depends largely on when stability in the responses is achieved (Linstone & Turoff, 2011). Consensus is more easily determined through this method than by calculating percentages (Hasson et al., 2000). In many cases, the third round is a repetition of the second round, except that the second-round results are provided to the panellists. In this study, a consensus was reached after two (2) iterations and the results are presented in Chapter 6.

The first step in exploratory Delphi is to choose a knowledge base from which to identify the various fields, relevant institutions, works of literature, and legal frameworks. After that, the first questionnaire (round one) serves as a plan for generating ideas to identify the problems in the study area (Ibiyemi, Adnan & Daud, 2016). In round one, the participants—referred to as panellists—should address as many relevant topics as they can. A second questionnaire that asks opinions on the topics raised complements the feedback provided from round one responses (Ibiyemi *et al.*, 2016). The subsequent rounds include each panellist's (expert's) response as well as those of the other panellists. In light of the comments from the other panellists, each expert is invited to re-evaluate their response (Ibiyemi *et al.*, 2016). The input process continues through several rounds until a consensus is reached. The process is best characterized as multi-staged, with each stage building on the results of the previous one (Nworie, 2011; Green, 2014). Somerville (2008) and Lang (1995) provide the list of the process as follows:

- The participation of experts.
- The panel's size.
- Two or more Delphi rounds
- Heterogeneity.
- Anonymity.
- Feedbacks from each run that are under control and include some textual and statistical data.
- Analysis of data and
- Consensus.

Exploratory Delphi meets the purpose of this study because it allows experts to reach a consensus from an expert panel by means of questionnaires and controlled feedback which is how data was collected in this study. This method establishes facts about a specific topic, and SMS as well IMM was the main topic for this study. The first iteration stated facts about SMS and validated the need for a SMSIM model that can be used to evaluate interoperability level in SMS, and a consensus at 70% was reached necessitating the second iteration that validates the SMSIM model. The following section tables different scaling techniques that can be used and adapted when using the Delphi method. The importance of this table is to provide a different insight into how the interactions with the experts can be structured and ranked.

#### 5.7.6 Scaling techniques

The scaling technique involves assigning participants to continue a progressive adjustment in the previously given values or numbers based on the characteristics of a certain object in accordance with the established regulations (Triantafyllidis & Samaras, 2015; Badr, Abdul Salam, Almotairi & Ahmed, 2021). All the scaling techniques are based on four pillars, i.e., order, description, distance, and origin. There are many scaling techniques used for Delphi ranking presented in Table 5-5 and Table 5-6 below (Calil, Reguero, Zamora, Losada & Méndez, 2017; El-Khoueiry, Sangro, Yau, Crocenzi, Kudo, Hsu, Kim, Choo, Trojan & Welling 3rd, 2017; Wu & Leung, 2017; Chyung, Swanson, Roberts & Hankinson, 2018; Hill, Lettington & Schmidt, 2018; Varatharajan, Manogaran, Priyan, Balaş & Barna, 2018; Nasrabadi, Dehnokhalaji, Korhonen & Wallenius, 2019; Taherdoost, 2019; Javaheri, Mirzaei & Lukowicz, 2020; Holmberg, Wolf, Eide, Großschädl, Schüttengruber, Patel & Heckemann, 2021; Suparji, Nugroho & Martiningsih, 2021; Manisera & Zuccolotto, 2022).

# Table 5-5: Primary Delphi Scaling Techniques

Scaling	Description	Ranking
Technique		
Nominal Scale	For non-quantitative variables that are distinct from one	
Types:	another and do not have a numerical consequence,	
	nominal scales are used.	
Dichotomous	Dichotomous means there are only two labels on a	Yes or No
	nominal scale.	
Nominal with	"Nominal with order" refers to labels on a nominal scale	Excellent, Good, Average, Poor, Worst
Order	that are placed in either ascending or descending order.	
Nominal	The term "nominal without order" refers to such a	Black, White
without Order	nominal scale that lacks a sequence.	
Ordinal scale	The ordinal scale is founded on the idea that things or	The ratings so provided are as follows:
	labels are positioned relative to one another according to	5 Star – Excellent
	a person's preferences or choices.	4 Star – Good
		3 Star – Average
		2 Star – Poor
		1 Star – Worst
Interval scale	The numerical labeling with the same difference between	1, 2, 3, 4, 5, 6
	successive measurement units is known as an interval	
	scale, sometimes known as a cardinal scale. Researchers	
	can compare the objects more effectively by using this	
	scaling technique.	
Ration scale	A ratio scale is an abstract number system, just like an	If a business wishes to determine the prevalence of obesity in a specific area,
	interval scale. It has the additional feature of starting	select a category to which your weight belongs to:
	from a set zero point and enables measurement at the	Less than 40 kilograms
	appropriate intervals, as well as order, classification, and	40-59 Kilograms

distance. In this case, the acquired ratio can be used to	60-79 Kilograms
make the comparison.	80-99 Kilograms
	100-119 Kilograms
	120 Kilograms and more
	-

Table 5-6: Other Comparative Scaling Techniques

Comparative scale types:	Description	Ranking
Paired comparison	The respondents utilize a comparative scale when comparing two or more factors. The respondent must choose one of two variables represented by a paired comparison. This method is mostly employed during product testing to enable customers to compare the two key products available on the market.	Ranking For instance, a market study was done to determine whether consumers preferred brands A and B of network service providers. The survey's findings are as follows: Brand 'A' = $57\%$ Brand 'B' = $43\%$
Rank Order	In rank order scaling, the respondent must rank or	For example, a soap production company used rank order scaling to
	arrange the supplied objects in the order of their choice.	determine the consumers' preferred order. It asked the respondents to rank         the following brands in the sequence of their choice:         Soap Brand       Rank         Brand V       4         Brand X       2         Brand Y       1         Brand Z       3         The above scaling shows that soap 'Y' is the most preferred brand, followed by soap 'X', then soap 'Z' and the least preferred one is the soap 'V'.

Comparative scale types:	Description	Ranking
Constant Sum	It is a scaling approach in which the respondents rate the merits, qualities, and significance of a certain good or service using a running total of units such as dollars, points, chits, chips, etc.	For example, the respondents from three distinct segments were asked to provide the following cosmetic product characteristic (P) 50 points:AttributesSegment1 Segment 2 Segment 3Finish118Skin Friendly1112Fragrance711Packaging98Price1211IT11From the above constant sum scaling analysis, we can see that:Segment 1 considers product 'P' due to its competitive price as a major factor.But segment 2 and segment 3, prefers the product because it is skin-friendly.
Q-Sort Scaling	Q-sort scaling is a method for selecting the best objects from a large set of provided variables. It places emphasis on sorting the provided objects into similar piles in decreasing order according to predetermined characteristics. It is appropriate when there are at least 60 and no more than 140 objects, with 60 to 90 being the most optimal number of objects.	O-Sort Scaling Example         High       A       D       T       L       R       D       M       U       G       F       Z       P       E       G       T       N       B       A       K       T       O       T       S         High       A       D       T       L       R       D       S       M       U       G       F       S       P       E       G       T       A       N       B       A       K       T       S       A       M       D       T       S       A       N       B       A       K       T       S       A       M       D       T       S       A       K       X       T       S       R       P       A       J       C       L       A       L       N       D       X       T       A       A       C       C       K       H       A       K       D       X       T       A       L       N       L       C       K       K       K       D       C       K       A       C       C       K       H       A       D       X       C
Non- comparative scale types: Continuous Rating Scales	To evaluate a product or object's performance across various parameters, a non-comparative scale is utilized. The respondents can position the object on the graphical rating scale however they like. A place that falls	Worst Best 0 10 20 30 40 50 60 70 80 90 100

Comparative scale types:	Description	Ranking
	between two extreme requirements, either vertically or horizontally, is chosen and marked.	
Itemized Rating Scale	Another crucial method for non-comparative scales is the itemized scale. It places emphasis on selecting a specific category from the range of ones provided by the respondents. The researchers briefly define each class to facilitate such selection.	
Likert Scale	The researcher presents various statements to the respondents and asks them to indicate how much they agree or disagree with each one by choosing one of the five options available on the Likert scale.	<ul> <li>1 - Strongly Disagree</li> <li>2 - Disagree</li> <li>3 - Neither Agree Nor Disagree</li> <li>4 - Agree</li> <li>5 - Strongly Agree</li> </ul>
Semantic Differential Scale	For each of the specified properties of the object, the respondent may select any one of the seven points on a bi-polar, seven-point non-comparative rating scale. demonstrating the respondent's attitude or view of the object.	Semantic Differential Scale         +3       +2       +1       0       -1       -2       -3         Stylish         X       Unfashionable         Simple        X        Complex         Affordable         Expensive         High Quality       X        Low Quality         Wide Variety        X          From the above diagram, we can analyze that the customer finds the product of superior quality; however, the brand needs to focus more on the styling of its watches.

Comparative	Description	Ranking		
scale types:		Trunning (		
Stapel Scale	A Stapel scale is an itemized rating scale that uses a unipolar rating to gauge respondents' reactions, perceptions, or attitudes toward a certain object. A Stapel scale can only go from -5 to +5, removing 0, and is therefore limited to 10 units.	For example, a tour operator requested respondents to score their vacar package according to its affordability and user-friendliness using the following criteria: <b>Stapel Scale</b> +5 +4 +3 +2 × +1 Value for Money User Friendly Interface -1 -2 -3 -4 -5 With the help of the above scale, we can say that the company needs to improve its package in terms of value for money. However, the decisive point is that the interface is user-friendly for the customers.	tion o /e	

The Likert Scale will be further discussed as it will be used for ranking during the evaluation of the components and for ranking these in this study during an iteration of questionnaires with the experts.

#### 5.7.6.1 Likert Scale

The Likert-scale is the most common instrument used to measure affective factors, including motivation and self-efficacy, as they make it very simple for researchers to collect significant amounts of data (Nemoto & Beglar, 2014; Chyung *et al.*, 2018; Matas, 2018). An initial 5-point psychometric scale for evaluating several propositions connected to attitudes was developed in the 1930s by American social psychologist Rensis Likert (Likert, 1932; Alan & Kabasakal, 2020). The wording used in the initial Likert scale was: Strongly Approve, Approve, Un-decided, Disapprove, and Strongly Disapprove. Over time, the wording changed from Approve to Agree, which resulted in the Likert scale we know today: Strongly Agree and Agree on one side, and Disagree and Strongly Disagree on the other side, with a midpoint in the middle.

The conventional Likert scale evolved into other Likert-type scale modifications because of its ease of use and popularity. These are commonly used in survey instruments designed for (Ryan, 1980; Garland, 1991; Raaijmakers, Van Hoof, T Hart, Verbogt & Vollebergh, 2000; Boone & Boone, 2012; Kulas & Stachowski, 2013; Purdey, 2013; Chyung *et al.*, 2018; Claveria, 2021):

- Evaluating how well employees do at work.
- Research on communications.
- Marketing research.
- Psychometric research
- Political opinion research.

Debates over validity arise from the variations in Likert-type scales in the context of these applications, such as the impacts of using the following formats (Chyung *et al.*, 2018):

- Including or not including a midpoint in the scale:
  - Strongly disagree, disagree, neutral, agree, strongly agree.
  - Strongly disagree, disagree, agree, strongly agree.
- Using descending order vs ascending order of the scale opinions:
  - Strongly agree, agree, neutral, disagree, strongly disagree.
  - Strongly disagree, disagree, neutral, agree, strongly agree.

Measuring positively and negatively stated survey items with the Likert scale

- The objectives were clear:
  - Strongly disagree, disagree, neutral, agree, strongly agree.
- The objectives were unclear:
  - Strongly disagree, disagree, neutral, agree, strongly agree.
- The objectives were not clear
  - Strongly disagree, disagree, neutral, agree, strongly agree.

### 5.7.6.2 The Likert scale as a measurement method

Another American psychologist, Stanley Smith Stevens, developed the nominal, ordinal, interval, and ratio scales as four different sorts (levels) of measurement in the 1940s (Stevens, 1946; Kuhlmann, Dantlgraber & Reips, 2017). The ordinal scale has levels that are ranked in sequence (for example, performance ratings such as Poor, Fair, and Good). The interval scale has levels with consistent spacing between them (for example, a scale consisting of -2, -1, 0, 1, and 2). Finally, the ratio scale has levels with proportionate ratios and a meaningful zero value (for example, test scores of zero to 100). The middle point, or "neutrality," on the Likert scale's five points (or "anchors"), is used to measure attitudes. As illustrated in Figure 5-4, the lengths between successive points on the scale must be equal for the Likert scale to qualify as an interval scale. To put it another way, the distance between Strongly Disagree and Neutral ("d"), which is the same as the distance between Neutral and Agree ("b"), as well as the distance between Agree and Strongly Agree ("c") (Sullivan & Artino, 2013; Chyung, Roberts, Swanson & Hankinson, 2017).

- Using Likert-type scales or slider scales
  - o Completely Dissatisfied 0 1 2 3 4 5 6 7 8 9 10
  - o Completely Satisfied A slider scale as shown in Figure 5-4



#### Figure 5-4: A Slider Scale

In this study, the Likert scale is used as a measurement method, the following section describes the Likert scale as a measurement method.



#### Figure 5-5: The Five-point Likert scale as an ordinal scale

There may also be a difference in the distances between the Likert scale's "a," "b," "c," and "d" points as seen in Figure 5-5. As a result, the Likert scale cannot be used as an interval scale and is instead classified as an ordinal scale. Furthermore, it's crucial to word a midpoint correctly. As a midpoint, Neutral (or Neither Agree Nor Disagree) denotes a level of opinion that is neutral. However, if Undecided is chosen as the midpoint, it is debatable whether it represents a point of view that lies halfway between disagreement and agreement or whether it should instead be viewed as an absence of opinion (Sullivan & Artino, 2013; Chyung *et al.*, 2017).

In the latter situation, as shown in Figure 5-6, the Likert scale resembles a 4-point ordinal scale with Undecided as a separate option off the scale. Other perspectives, such as I don't know or it depends on, should not be labelled as a midpoint but instead should be offered as response options off the scale to capture data that is more likely to be interval than ordinal in nature (Chyung *et al.*, 2017).



Figure 5-6: The Four-point ordinal Likert scale with undecided as a separate option

This study only used the five-point Likert type ranging from 1= strongly disagree to 5 = strongly agree (Dönmez & Toker, 2017). The experts were given questions through two (2) iterations, and they chose either of the options below:

- 1 Strongly Disagree
- 2 Disagree
- 3 Neither Agree nor Disagree
- 4 Agree
- 5 Strongly Agree

Table 5-5 and 5-6 above provides different scaling techniques that can be used for ranking in the Delphi method. For the purpose of this study, Likert scale was adopted as it allows facts from literature and the experts input. This scale complements the modified Delphi design and hence it is adapted for this study. The process of the Delphi technique that was followed in this study is outlined in the figure below:



Figure 5-7: Delphi Method 1st iteration (John-Matthews et al., 2017; Mukherjee et al., 2014)

In round one, the first four steps in the above process were followed. During this round, questionnaires were prepared to identify relevant experts. In section 5.7.3, Table 5-3, presents the selection criteria of experts and only experts that meet the criteria were selected to participate in the study. The first questionnaire was drafted from the literature findings. This is important to solidify the knowledge and experience in this field. The selected experts were invited to participate in the study.

Round 2 only began after the first round was complete. The questionnaire from round 2 were developed based on the feedback from round 1, the pre-conceptualized SMSIM model was provided to the experts in round 2 based on the feedback they provided in round 1, and this started the discussions of validating the model. The process repeated itself in two (2) iterations until a consensus was reached. Anonymity is key in this process, and all the experts anonymously received a copy of the results.



Figure 5-8: Delphi Method 2nd iteration (John-Matthews et al., 2017; Mukherjee et al., 2014)

Round 2 followed the above process and until the last step was reached (see Figure 5-8). Each round consisted of verifications and analysis of the feedback obtained. Section 5.4 of the hermeneutics circle determined the relevancy of the questions that were prepared for the next round. The following section discusses the data collection methods (Appendix C and D provide the detail of what was asked during the two Delphi rounds).

#### 5.8 THE DATA COLLECTION METHODS

In order to answer specific research questions, test hypotheses, and assess results, data collection is the act of acquiring and analyzing information on relevant variables in a planned, methodical manner (Kabir, 2016). All academic disciplines, including the humanities, social sciences, business, and natural and applied sciences, share the data collection component of research. Although techniques differ depending on the profession, the importance of ensuring accurate and truthful collection does not change. The objective of collecting data is to obtain high-quality data that can be used for rich data analysis and to create responses to posted questions that are convincing and believable (Kabir, 2016). This study applied questionnaires (online), and document analysis as its data collection methods.

#### 5.8.1 Questionnaires

Any text-based tool that provides users with a series of questions to answer or statements to respond to by marking a page, writing a number, or checking a box on paper or online is referred to as a questionnaire (Roopa & Rani, 2012; Young, 2015). For less experienced researchers, such as students working on dissertation projects, questionnaires look particularly alluring. This may be the case for the following reasons (Roopa & Rani, 2012; Young, 2015):

- They are simple to build, or at least it may seem that way.
- Many of the currently used questionnaires can be used or modified for usage, and they are frequently freely accessible to researchers.
- They are moveable or can be bought online.
- They can be utilized to swiftly compile massive datasets through direct contact, mail, the Internet, or email.
- In contrast to spoken data, which must be recorded and transcribed before analysis, the information they collect can be processed and analyzed rather fast.

Since interviews were not possible because of the unavailability of participants who were willing to go through the traditional interview process, a Delphi study was chosen and questionnaires were prepared for data collection. The following five important strategies were used during the qualitative data collection phase (Leedy & Ormrod, 2021):

• The researcher used self-reflection/reflexivity to identify individual, communal, and political prejudices that could negatively affect the capability to interpret data collected.

- Triangulation was used to find inconsistencies or consistencies among the data that was collected in multiple forms related to the same research question.
- Clearly distinguishing between data and memos helped the researcher's interpretations.
- Seeking exceptions and contradictory evidence.

Questionnaires were developed using the Likert scaling technique and provided online through Lime survey to experts to respond to in two iterations of the exploratory Delphi study until a consensus was reached. This provided a holistic view of how each person with a different role interacting with the system responds to the common question. A perception was then formed based on analysing the results from different trends of participants. Questionnaires were used throughout the process of verifying and validating the model to ensure that there is a difference, relevance, efficiency, and credibility to the newly conceptualized model.

## 5.8.2 Document Analysis

In document analysis, a type of qualitative research, the subject of the study is given voice and meaning by the researcher's interpretation of the documents (Bowen, 2009). One must follow a systematic process called document analysis to evaluate or assess printed and electronic (computer-based and Internet-transmitted) documents (Bowen, 2009; Wach & Ward, 2013). Like other analytical methods in qualitative research, document analysis requires that data be examined and interpreted to elicit meaning, gain understanding, and develop empirical knowledge (Bowen, 2009).

There are many different types of documents used for systematic examination as part of a study and these include manuals, diaries, letters and memorandums, advertisements, background papers, agendas, program proposals, attendance records, newspapers, and meeting minutes, scripts for radio and television programs, printed event programs, brochures, and journals, printed event programs, charts, press releases, application forms, institutional or organization reports, maps, survey information, books, and various public records were among them (Bowen, 2009).

## 5.8.2.1 Advantages and limitations of document analysis

The following are the advantages and limitations of document analysis, according to Bowen (2009):

- Efficient method: In comparison to other research techniques, document analysis takes less time and is therefore, more effective. Instead of data collection, it requires data selection.
- Availability: Since the invention of the Internet, many documents have been in the public domain and are accessible without the creators' consent.
- **Cost-effectiveness:** When new data gathering is not practical, document analysis is frequently the method of choice because it is less expensive than other research techniques. The information (included in the documents) has already been acquired; the documents' content and quality evaluation is now necessary.
- Lack of obtrusiveness and reactivity: Documents are "non-reactive" and "unobtrusive," meaning that the research process has no effect on them. (Previous research mentioned in documents is not considered here.
- **Stability:** Documents are stable because they are non-reactive. No changes are made to the study because of the researcher's participation (Merriam, 1988). In this way, documents are appropriate for further scrutiny.
- **Exactness:** Documents are useful in the research process because they contain precise names, references, and event information (Yin, 1994).
- **Coverage:** Documents offer comprehensive coverage; they span a long time, several events, and numerous settings (Yin, 1994). Analysis of documents is not always beneficial. Below are a few restrictions that documents naturally have.
- **Insufficient detail:** Documents are made outside of a research objective and are produced for a purpose other than research. As a result, they frequently don't offer enough information to address a research topic.
- Low retrievability: Sometimes retrieving documentation is impossible or very difficult. It's possible to purposefully restrict access to papers, as demonstrated by Yin (1994).
- **Biased selectivity:** A deficient collection of documents raises concerns about "biased selectivity" (Yin, 1994, p. 80). The available (chosen) documents in an organizational setting are probably going to be in line with corporate rules and procedures and the objectives of the organization's principals. However, they may also reflect the organizational unit's emphasis on record-keeping (for example, Human Resources).

Given its efficiency and cost-effectiveness, document analysis offers advantages that clearly outweigh the limitations.

For this study document analysis was not used, Chapters 2, 3 and 4 of this study focused on the literature findings. The next section describes the methodology used for literature findings.

## 5.8.3 Scoping Reviews

The literature was obtained through scoping reviews. The purpose was to better understand the available literature conducted by other authors prior to this study, and how this literature contributed to interoperability, interoperability maturity models (IMM) and SMS. According to Nakano and Muniz (2018, p.1), researchers contribute to meaningful research based on prior knowledge that they have developed through a "collective and cumulative endeavour".

A scoping review is a valuable tool in the arsenal of evidence synthesis techniques that continue to grow. Scoping reviews are now considered a valid alternative when systematic reviews are not adequate to meet the users' needs. A scoping review clearly indicates the volume and nature of the literature and studies on a given topic (Munn, Peters, Stern, Tufanaru, Mcarthur & Aromataris, 2018). It also describes the focus of the literature and studies. Generally, scoping reviews aim at identifying and mapping the evidence available (Munn *et al.*, 2018). Therefore, each literature chapter identifies interoperability, IMM and SMS themes.

### 5.8.3.1 The objective of scoping reviews

The following section outlines the rationale that enforces the scoping review approach. Among the reasons researchers may be more inclined to use scoping reviews, Arksey and O'Malley (2005) identify the following:

- Research findings are not presented in detail, but rather a range of information is consolidated.
- The mapping of literature may form the basis of a full systematic review.
- Summarizing research findings in a specific area, aiming to provide an overview of the information found.
- Finding gaps in literature where there has been minimal or no research.

In the context of this study, the scoping review was used to determine what has been covered about ICT, SMS and IMM, the main objective was to find literature that could support the problem statement of the study.

### 5.9 DATA ANALYSIS

Data analysis is a creative and dynamic process. Researchers attempt to increase a profound understanding of what they have studied and continually refine their interpretations throughout the analysis process (Taylor, Bogdan & Devault, 2015). Creswell and Creswell (2018:p376) define data analysis as when a researcher organizes and tries to make meaningful conclusions from collected data. Researchers draw first-hand experience with settings, informants, or documents on their data analysis to interpret their data (Taylor *et al.*, 2015). In qualitative research, data analysis is probably the most difficult aspect of communicating with others or teaching (Taylor *et al.*, 2015). The preliminary coding scheme will be adapted and refined during the study.

The qualitative data that was using the questionnaires was available as text data. Data analysis will also be done using Microsoft Excel and software analysis tools such as open coding to guide the analysis phase with Nvivo 12 (Nvivo, 2021). To improve data analysis, Creswell and Creswell (2018:p307) recommend developing code inside the primary documents. The Nvivo software lets you annotate, link, search, code, query, and visually represent qualitative data (Creswell & Creswell, 2018:p307). The analysis will be guided by hermeneutics, as was discussed in Section 5.4. The ability to explain written scripts is referred to as hermeneutics (Friston & Frith, 2015). The practical operation that provides knowledge in the sense of familiarity with something, the sort of understanding by which we integrate facts into a meaningful whole, is hermeneutics or interpretation (Zimmermann, 2015). Data analysis was conducted based on the feedback from the various experts in each round to ensure that the literature study trends match the empirical data.

#### 5.9.1 Thematic Analysis

A component of the qualitative research methodology is thematic analysis. It is a technique used for examining data collections to find and document any established patterns that could exist (Saunders *et al.*, 2016:p651). It is also a useful method for creating connected themes and codes in academic research. In contrast to grounded theory or hermeneutic phenomenology, which demand a higher level of interpretive complexity, the use of theme analysis is

appropriate for researchers who prefer to apply a relatively modest level of interpretation. Thematic analysis is primarily defined as a strategy for detecting, interpreting, and reporting patterns (themes) within data as an independent qualitative descriptive approach (Vaismoradi, Turunen & Bondas, 2013; Castleberry & Nolen, 2018; Neuendorf, 2018:13; Braun & Clarke, 2019). Finding and identifying recurring themes in data is done through thematic analysis (Desantis & Ugarriza, 2000; Neuendorf, 2018:13; Kiger & Varpio, 2020). Thematic analysis can be conducted within both realist/essentialist and constructionist paradigms, although the outcome and focus will be different for each (Braun & Clarke, 2006). The systematic aspect of content analysis that thematic analysis may provide allows the researcher to integrate meaning analysis with context analysis (Kiger & Varpio, 2020). Thematic analysis evaluates several facets of the study by applying minimum description to data sets (Braun & Clarke, 2006; Vaismoradi *et al.*, 2013; Kiger & Varpio, 2020). In this study, thematic analysis was used to identify possible themes presented in Table 5-7. The Table below outlines the processes of data analysis in thematic analysis that was followed.

*Table 5-7: Processes of data analysis in thematic analysis (Braun & Clarke, 2019; Kiger & Varpio, 2020).* 

Thematic	Descriptions				
Analysis					
Familiarising with	To obtain themes as presented in Table 5-8, a lot of reading occurred while				
data	noting down ideas. The Nvivo 12 was used to visually present qualitative data.				
Generating initial	Analyzed the data, systematically identified its most interesting features, and				
codes	collated relevant data.				
Searching for	Identified potential themes, and collected all relevant data related to each				
themes	potential theme				
Data Organization	This involved using different folders to separate themes and group related				
	themes based on the focus area. This ensured that the correct theme was				
	utilized when required to build the outcome of the study.				
Interpreting Data	To establish meaning in a broader sense, it was important to make sense of the				
	code and themes.				
Reviewing themes	The selected themes were checked if they work in relation to the data sets.				
Defining and	Clear definitions and names for each theme were generated, as in Table 5-8				
naming themes	through an ongoing analysis process.				
Producing the	The outcome of this process is documented throughout this study. Chapter 2, 3				
report	and 4 discuss these themes in more detail to form a base for the components of				
	the SMSIM model.				

The most important part of the study is to answer the main research question: How can a SMSIM model assist schools in improving their current SMS's interoperability and maturity levels? And to achieve this, the main research question was broken down into manageable sub questions to determine the bounds of what is already covered in literature and themes were extracted in this study across various literature chapters (Chapters 2, 3 and 4), as presented in the table below:

Table 5-8: T	hemes
--------------	-------

Chapter	Relevant SRQ	Extracted Themes	Definition
Chapter 2	What constitutes the	Theme 1: Information and	ICT Communication and
	components of a model	Communication	information technologies
	of interoperability	Technology (ICT)	can be thought of as a set
	maturity for Schools		of tools and resources that
	Management Systems?		facilitate communication
			and the creation,
			dissemination, storage, and
			management of
			information.
Chapter 3	What value will a	Theme 2: School	The Interoperability
	SMSIM model have for	Management Systems	Maturity Levels of SMS
	schools using Schools	(SMS)	will be assessed and
	Management Systems?		guidelines will be provided
			to improve the level of
			interoperability in SMS.
Chapter 4	What existing models	Theme 3: Interoperability	IMM is a model that
	of interoperability	Maturity Model (IMM)	measures interoperability
	maturity can be used to		in a specific domain.
	conceptualize a SMSIM		
	model?		
These themes ensured the researcher kept track of the purpose of the study and ensured there was no SRQ that was not covered through literature as presented in the table above. This section was extremely important as it streamlined the focus area of the study and excluded all irrelevant literature findings that did not focus on the themes to answer the SRQs. Overall, the literature considered for review was organized and classified into folders based on the research's themes, including ICT, SMS, and IMM literature. After reviewing the literature, concepts associated with these were analyzed. The codes were then used to determine which concepts were more prevalent throughout the literature reviewed based on their frequency of occurrence. The results are shown in Figure 5-9.



Figure 5-9: Most frequently used words across the selected literature

Several themes were generated using Nvivo 12, through the theoretical foundation in Chapter 2,3 and 4 of literature review. Word cloud shows the commonly used concepts found in literature chapters as displayed in Figure 5-9 above, which contributed to themes that form a foundation for conceptualizing the SMSIM model. This figure shows how the most prominent words are aligned with the theoretical basis of the research as well as the aspects that comprise the research themes.

#### 5.10 TRUSTWORTHINESS

A study is considered trustworthy if the research report's reader believes it to be so (Gunawan, 2015). Trustworthiness has been further divided into credibility, which corresponds roughly with the positivist concept of internal validity; dependability, which relates more to reliability and transferability, which is a form of external validity; and confirmability, which is largely an

issue of presentation (Gunawan, 2015). The aspect of trustworthiness will now be further discussed below.

# 5.10.1 Credibility

Credibility is a degree of honesty, character, also faith in an entity. Credibility may result from personal and independent mechanisms relating to the services and goods given by the entity. Credibility is constructed over time by customers' single experiences and others participating in viable transactions with the entity (Stibel, Stibel, Hackett, Kazerani & Loeb, 2015). To guarantee that the data is precise, to corroborate the findings and enrich their validity, data triangulation was used (Yin, 2009; Herselman, 2011):

 data triangulation contains the utilisation of a diversity of data sources in a study. This study involved the experts, documents relevant to the study, and SMS as explained in the literature reviews.

It is evident from the definition and the discussion provided above that credibility is an important aspect of trustworthiness. The two ensure that the research is not compromised in any way and that the outcome of the research is credible. Thus, this research will be trustworthy and credible. The section below discusses the ethical considerations of the study.

## 5.11 ETHICAL CONSIDERATIONS

David and Resnik (2011) define ethics as the difference between acceptable and unacceptable behaviour. Furthermore, David and Resnik (2011) define ethics as a point of view, a step-by-step process, or techniques for logically understanding difficult issues and problems. This research will follow ethical guidelines laid down by the University of South Africa (UNISA) and the DBE to protect the rights of all participants and ensure good research is conducted fairly.

The researcher will not expose research participants, and their rights and privacy will be highly respected. The findings will be reported wholly and truthfully, excluding twisting what was done by the participants (Leedy & Ormrod, 2021). A consent form detailing the purpose of the project and the rights and privacy of the participants during this research was provided to the participants to sign before applying the data collection methods (provided in Appendix C &D). Evidence of the ethical clearance from UNISA is available in Appendix A.

#### 5.12 SUMMARY

This chapter provided an overview of the applicable research methodology for this study. The research onion outlined the layers involving the research philosophy in Section 5.3 with the theoretical lens and the hermeneutic circle in Section 5.4. The research approach and using the Delphi technique in this exploratory qualitative study were discussed together with the advantages and disadvantages of applying the Delphi technique. The data collection methods (questionnaires and scoping literature reviews), the data analysis (hermeneutics and thematic analysis) and trustworthiness were outlined and finally the ethical considerations were provided.

From Chapter 5 it was evident that this is an exploratory qualitative study that applied the Delphi expert reviews as a strategy to focus on SMS at schools. The following chapter focuses on the results of the Delphi study.

# CHAPTER 6: ASSESSMENT AND VALIDATION OF THE SMSIM MODEL

## 6.1 INTRODUCTION

In Chapter 5, the research approach was covered. The findings of the assessment and evaluation of the Delphi expert reviews are presented in this chapter (6). This chapter covers phases two and three of the research procedure described below.



Figure 6-1: The research process and how Chapter 6 covers phases 2 and 3

The Delphi Method is important as it was used to collect data from experts, as explained in Chapter 5, Section 5.7 (Research strategies). With this approach, getting a group of carefully chosen experts to agree on a position is the primary goal (Deveci, Özcan, John, Covrig & Pamucar, 2020). Due to these reasons, Chapter 6 conceptualize the School Management Systems Interoperability Maturity (SMSIM) model and discusses expert assessments and evaluation of the Delphi study's findings.

A consensus was reached after two rounds of the Delphi Study, the first round focussed on basic knowledge of School Management Systems (SMS) to validate the problem statement in Chapter 1 of the study, ascertain the need for a SMSIM model, and assess the level of understanding from the experts in the subject matter. According to David and Roberta (2020),

the questions for each round are based in part of the findings of the previous one, allowing the study to evolve over time in response to earlier findings. In the first round a 70% consensus was reached that necessitated the second round that will verifying and validating the SMSIM model (Etikan, Musa & Alkassim, 2016). The section below presents the findings from round 1 of the Delphi Study.

First, the process of conducting the evaluation will be explained, followed by the analysis of the data and finally the considerations of the experts with the results in each round will finalise this chapter.

#### 6.2 THE PROCESS OF CONDUCTING THE EVALUATION

Determining merit, worth, value, or relevance through a methodical process is called evaluation (Picciotto, 2011; Wanzer, 2021). Furthermore, evaluation is defined by Preskill and Russ-Eft (2015:448) as a type of research that seeks to answer important questions about the effectiveness of a program, process, product, system, or organization. Most of the time, it is carried out for decision-making, and diverse stakeholders should use the results (Wager, Lee & Glaser, 2017).

An evaluation was conducted throughout the study to determine the merit, worth, value and significance of the SMSIM model. The focus was on ensuring the conceptualized SMSIM model addressed the study's problem statement in Chapter 1, Section 1.2. Throughout the process of conceptualizing the SMSIM model, questionnaires for the first round of the Delphi study were evaluated to ensure they appropriately aligned and addressed the problem statement. A fundamental question in the expert review process is answering the question, "who is an expert?" (Vesper, Reeves & Herrington, 2011). Thus, it was important to carefully select experts based on their years of experience, level of understanding, knowledge and expertise in the field, willingness to partake in the study, and availability. Purposive sampling was utilized to choose the experts who would participate in this research (Chapter 5, Section 5.7.3). Purposive sampling is a non-probability, judgmental sampling technique that carefully selects study participants (Etikan et al., 2016). Experts responded on the model's validity and efficiency, which validated the merit, worth, value and significance of the SMSIM model.

#### 6.3 EXPERT REVIEW RESULTS FROM ITERATION 1

The conclusions of the evaluation carried out through expert reviews are outlined in the following sections to determine the findings' significance, they are interpreted.

### 6.3.1 Expert Review Demographic Information

As stated in Chapter 5, Section 5.7, the Delphi method/technique was used, in which chosen experts (participants) were requested to respond anonymously and individually to questionnaires. The panellists are given the chance to alter their responses considering the opinions of other panel members after the moderator summarizes all the responses from the panel at the conclusion of each round. The procedure is repeated until a predetermined degree of stability or answer saturation is obtained. The Delphi questionnaires were structured using a Likert Scale. See Appendix C for the questionnaires. The researcher presented certain statements to the participants and asked them to indicate how much they agreed or disagreed with each one by choosing one of the five options from the Likert scale.

In addition, a questionnaire was composed of general knowledge statements to assess not only the level of knowledge and understanding of SMS from experts but also the need for a SMSIM model.

The target group was six (6) experts; however, a total of ten (10) heterogenous experts in various fields of expertise as was presented in Table 5-3, participated in the study. Figure 6-2 below presents experts who participated in the study and their fields of expertise are School Management Systems (SMS), ICT in Education, and Interoperability.



Figure 6-2: Experts who participated in the study.

# 6.3.2 Results of the Delphi Iteration 1

# 6.3.2.1 Round 1 Delphi Questionnaires

The section below presents the first Delphi questionnaires, and the result of the questionnaires are further articulated.

• School Management Systems (SMS) are large repositories that schools use to manage the day-to-day administrative activities of the school.



Figure 6-3: Q1 Delphi Round 1 Questionnaire

Figure 6-3 above shows that 50% of the experts strongly agreed, 30% agreed, and just 20% strongly disagreed with the posed statement. After considering the 50% of experts who strongly agreed, together with the 30% who agreed, 80% of experts agreed with the statement. Thus, the results reaffirm the literature findings where Sarker (2016) defined School Management Systems as large repositories that schools use to manage the school's daily administrative activities. The 20% of experts who disagreed with the statement could result from not fully understanding the system, not directly interacting with the system, or not fully understanding the statement. Despite the 20% of experts who disagreed with the statement, it is evident from

the high percentage of experts agreeing that both experts and literature support that SMS are big repositories used for daily management of the school's administrative activities.

• Multiple users use SMS, and these can be configured to meet the needs of each school independently



Figure 6-4: Q2 Delphi Round 1 Questionnaire

Based on Figure 6-4, 40% of experts strongly agreed, 40% of experts agreed, and 20% of experts disagreed. Thus, 80% of the experts agreed with this statement. The result of this statement solidifies what has been obtained from literature as Sarker (2016), and Akter (2018) indicated that they have also found that multiple users use SMS systems for different purposes and they configure these systems as they go along. Therefore, it is evident that both literature and experts' results support the fact that multiple users use SMS and configure these systems.

 SMS help to overcome monotonous paperwork in schools, are very easy to use, and can work accurately and smoothly in different scenarios. SMS reduce workload, increase efficiency in school management, and save time.



Figure 6-5: Q3 Delphi Round 1 Questionnaire

The statement above was derived from literature, Sarker (2016) argued that SMS helps to overcome monotonous paperwork, they are easy to use, reduce workloads, increase efficiency in management and save time, whereas 50% of experts strongly agreed, 40% of the experts agreed and only 10% of experts neither agreed nor disagreed to Sarker's statement. Other authors, also agreed with Sarker's statement (Ramazhamba *et al.*, 2018; Ncanywa & Sibiya, 2020; Ndaba, 2021). This resulted in 90% of the experts supporting the statement. Figure 6-5 has a higher percentage of experts supporting this statement, this advantage of SMS seems to be applicable for both paid, open, and free versions of SMS. It is thus evident from literature and these results that SMS overcome tedious paperwork, making these easy-to-use systems that can accurately and smoothly work in different scenarios, significantly reducing workloads, saving time, and increasing efficiency in school management.

 Many schools in South Africa use a specific SMS called SA-SAMS (South African School and Administration Management System) as mandated by the South African Department of Basic Education.



Figure 6-6: Q4 Delphi Round 1 Questionnaire

According to Figure 6-6 above, 30% of experts strongly agreed, 40% agreed, 20% neither agreed nor disagreed and only 10% disagreed; the result was 70% in favour of the statement. The 70% of the experts who responded in favour of the statement may use SA-SAMS for reporting purpose, for example, importing the school reports into SA-SAMS. Also, they could have knowledge that supports this statement. 20% of experts could not really support or go against the statement. Gxwati (2011) and Ramazhamba *et al.* (2018) alluded that many schools use SA-SAMS, and the DBE mandates this. As per literature findings, the DBE in seven provinces collects data from all schools quarterly from the SA-SAMS administrative data system (Van Der Berg, Van Wyk, Selkirk, Rich & Deghaye, 2019). Thus, literature and expert results support the use of SA-SAMS by many schools as mandated by the Department of Basic Education of South Africa.

• To make the administration of schools more effective, the DBE developed SA-SAMS to solve not only management and administration challenges but also to streamline data in schools.



Figure 6-7: Q5 Delphi Round 1 Questionnaire

The statement above was derived from literature, and a total of 80% of experts agreed with the statement, 80% including the 40% who strongly agreed and 40% who agreed. This question aimed to assess the knowledge the experts have on the background of SA-SAMS. However, not all the experts could confidently respond to this question as 10% chose neither to agree nor disagree, and 10% of the experts strongly disagreed with the statement. Muriithi and Masinde (2016) indicate that the DBE developed SA-SAMS to solve management and administration challenges and to streamline data in schools, thus making the administration of the schools more effective. The results of literature and expert reviews solidify that the DBE developed SA-SAMS to make the school's administration more effective in solving management and administration challenges and streamline data in schools.

• Not all schools in South Africa are using SA-SAMS, but their SMS should easily integrate with SA-SAMS to reduce duplicate work.



Figure 6-8: Q6 Delphi Round 1 Questionnaire

10% of the experts chose not to respond to the statement, 30% strongly agreed, 30% agreed, and 30% neither agreed nor disagreed with it, a total of 60% agreed to the statement. Compared to previous statements, this statement seems to have received a lower percentage of support, and this may be because the SA-SAMS system is used by schools in South Africa more for reporting purposes than as the main SMS system. Muriithi and Masinde (2016) indicate the disadvantage of SA-SAMS at that it cannot connect with other systems in the DBE, which creates manual means of report submissions. Literature findings has identified that manual means are only applicable to offline schools (no Internet), schools with internet access allows online report submission to the DBE using the Valistractor software (See section 3.5.6).

So far, there is no evidence of schools that have SMS that can integrate with SA-SAMS, therefore, reporting is still done manually by exporting the report from one system to another instead of clicking a button (Muriithi & Masinde, 2016). This validates the need to have a

SMSIM model that can assess the level of interoperability in SMS and then provide guidelines on how interoperability of the next level can be achieved.

In the comment section of this statement, experts expressed that "SA-SAMS is inherently flawed and unstable. Schools must spend a lot of money and time every time the DBE change the goalposts for data needed for SA-SAMS". Kuriakose (2014) also indicates that SA-SAMS provides patches or fixes for the bugs that crop up over time. Each school must download and install these patches onto the system, which may sometimes be a hassle to achieve. There is no stability or consistency in what the DBE is trying to achieve using SA-SAMS. Mbuqe (2020) indicates that system problems in SA-SAMS result in the unreliability of the information produced. There are three-term and four-quarter schools, which causes issues with deadlines and data availability. Others mentioned that "SA-SAMS is probably the worst SMS system in use in South Africa. It is not user-friendly, and the support is lacking. Every time data must be entered into SA-SAMS, the DBE changes the format of the database and what they need. This causes the schools to spend a lot of money to pay for development to be able to export data in the format the DBE requires".

It is evident from the results provided by experts and literature findings that SA-SAMS really needs improvement. Integrating with this system is not feasible right now as it is flawed. There is a lot of cost implication for the schools when importing data into SA-SAMS, which also causes many delays. The SMSIM model provides guidelines for achieving interoperability for schools in South Africa. This finding has validated the importance of interoperability in education systems and the need for integrating these systems. Experts also indicated that "Ed-Admin is a great School Management System, and experts who use this SMS feels it is the best system in the world as it is interoperable and can be accessed from anywhere in the world".

- Although SMS offer many advantages to schools, it also comes with many disadvantages, such as:
  - Increased workloads for teachers



Figure 6-9: Q7 Delphi Round 1 Questionnaire

Figure 6-9 resulted that 20% strongly agree with the statement, 50% agree, 10% neither agree nor disagree and 20% strongly disagree. This totalled 70% in support of the statement. It is clearly observed that despite paid and free versions of SMS in schools, the workload is still a big problem for teachers in the education sector. For SA-SAMS, this is a disadvantage because SA-SAMS is installed on a standalone computer which limits access to users (Muriithi & Masinde, 2016; Pillay, 2020). Only the school administrator has access to this system, and teachers are forced to manually record masks and submit them to the administrator to input into SA-SAMS (Muriithi & Masinde, 2016; Pillay, 2020). The administrator is often frustrated with the amount of work they must do and the limited time they must report the data to the provincial department (Muriithi & Masinde, 2016). A SMSIM model can help guide how this disadvantage can be reduced by assessing the level of interoperability the SMS of the school is currently on and providing guidelines to get to the next phase.

Experts who use Ed-Admin felt that all these disadvantages mentioned above only apply to SA-SAMS, as they do not have any issues using Ed-Admin. Ed-Admin is available worldwide, and staff and parents can access it 24 hours daily (Walsh, 2022). It works incredibly well and hardly duplicates work if the staff member uses the system as intended.

• Duplication of Work



Figure 6-10: Q8 Delphi Round 1 Questionnaire

As per the Figure 6-10 above, it is evident from the feedback that 60% of the experts agreed to this disadvantage of SMS. However, 40% of the experts disagreed with the statement, which could result from their SMS catering for this disadvantage. What is essential to observe from this finding is that paid SMS have fewer disadvantages than open and free SMS. This disadvantage results from SA-SAMS being installed and accessed on a standalone computer (Muriithi & Masinde, 2016). According to Maremi *et al.* (2020), all SMS used by schools should easily integrate with SA-SAMS to reduce duplicating work.

• Redundancy and errors



Figure 6-11: Q9 Delphi Round 1 Questionnaire

A total of 60% support the statement, with 10% neither agreeing nor disagreeing and 20% disagreeing and 10% strongly disagreeing. This can be influenced by the different SMS used in the schools, as different experts indicated who participated in the study. 10% of the experts strongly disagreed, which could be that the SMS used solved this disadvantage identified by literature for open and free versions of SMS. Unfortunately, there are schools without internet connectivity, and they cannot fully realise the benefits of SMS as submission to the DBE is still done manually using CDs and USBs.

• Delays for the submission of reports to the DBE by schools



Figure 6-12: Q10 Delphi Round 1 Questionnaire

90% of the experts supported this statement, with only 10% disagreeing. Although private schools use a paid version of SMS, which is more technologically advanced and can be accessed anywhere, these SMS are still not integrated with the DBE SA-SAMS SMS. Unfortunately, this means submitting reports is still made manually by either exporting data into SA-SAMS or mailing the reports to the DBE (Muriithi & Masinde, 2016). The SMSIM model aims to assess the maturity level of interoperability in SMS and provide guidelines on how to get to the next level (Maremi *et al.*, 2020). As evident from both literature findings and expert results, delays in submitting reports to the DBE are still a problem. Guidelines from the SMSIM model can benefit even advanced SMS used in South Africa by schools.

 Most schools use a free version of SMS which often lacks features such as discipline, custom reports, registration, and inventory. Some are narrowed by space, the number of learners, and storage.



Figure 6-13: Q11 Delphi Round 1 Questionnaire

60% of the participants agreed with this statement and 20% disagreed, whereas 20% neither agreed nor disagreed. This result could be a because of experts who are using a paid version of SMS or knowledge of some schools that use a free version of SMS. According to Saadi (2017), there are various features that the open-source version is missing as compared to the commercial version, including inventory, custom reports, registration, and discipline. It is evident from both literature and expert results that many schools in South Africa use an open and free version of SMS. This open-source version often lacks some features and is narrowed by space capacity, storage, and the number of learners.

 SMS are frequently installed on a single solitary computer, which restricts access, puts a lot of strain on the administrator working alone, and is the only resource accessible during periods of high demand, like the end of the term when schools must process learner reports. This causes inconvenient delays.



Figure 6-14: Q12 Delphi Round 1 Questionnaire

Unfortunately, SMSs are not completely interoperable and cannot be accessed anywhere hence 60% of the participants agreed with this statement. There is a lot of room for improvement in SMS hence the proposed SMSIM model can add a lot of value to schools (Maremi et al., 2020).

• Teachers spend most of the teaching time manually capturing students' marks on neverending sheets of paper, which increases the likelihood of mistakes, and have limited access to the system to accomplish simple activities like recording marks.



Figure 6-15: Q13 Delphi Round 1 Questionnaire

Not all the paid versions of SMS can be accessed from anywhere, which can be observed from the responded percentage of this statement. If you look at Ed-Admin, an SMS mentioned by some experts during the data collection process in round 1, Ed-Amin can be accessed by all users from anywhere through apps and the Internet. Ed-Admin is very interoperable (Walsh, 2022), which thoroughly delineates this disadvantage of SMS. The 10% of participants who strongly disagreed could have used Ed-admin as their SMS. Similarly, the 30% that disagreed could be that their SMS allows them to access the system from anywhere at any time. The SMSIM model can assess all SMS and evaluate what is needed to reach the next level.

 Non-availability of skilled ICT personnel who can utilise the system. On the government's end, there is a lack of maintenance culture, appropriate and adequate availability of relevant software, data security, and impurity.



Figure 6-16: Q14 Delphi Round 1 Questionnaire

Looking at the outcome of this statement, 20% strongly disagreed, and this is subject to the type of SMS currently installed in the school. However, despite 20% of participants who strongly disagreed and 10% who disagreed, 60% of the participants agreed with the statement. This is still a high number of schools needing an improvement of their SMS.



• SMS can only be accessed within the school's intranet.

Figure 6-17: Q15 Delphi Round 1 Questionnaire

The outcome of this statement is subject to how SMS is accessed in schools. As per the results, 30% of the participants disagreed, this can be because the school uses Ed-admin or other SMS, which does not connect only to the school's intranet. Thwala (2019) and Kuriakose (2014) indicate that most open and free SMS are accessed through the intranet in the school premises. A total of 50% of participants agreed with the statement of SMS being accessible in the school premises through the intranet. It can be observed from the results that both literature and experts confirm that SMS are mostly accessed through the intranet of the school.



o Limit learners and parents or guardians from accessing the learners' information

Figure 6-18: Q16 Delphi Round 1 Questionnaire

A total of 50% of experts or participants agreed with the statement, 20% neither agreed nor disagreed and a total of 20% disagreed. According to Muriithi and Masinde (2016), SMS that are installed on a standalone computer, limit access to information for learners, guardians and parents. It is, therefore, evident from the literature findings and the experts' results that SMS limits access to information for learners, guardians, and parents.

Experts who use Ed-Admin felt that it is probably the best thing to have happened to their school as it is accessible from anywhere in the world, and parents can access their child's information through a portal. In addition, termly reports can be obtained through the system, saving money (Walsh, 2022). SA-SAMS does not have any of these benefits, and literature finding supports this statement.



o SMS has room for improvement due to its disadvantages

Figure 6-19: Q17 Delphi Round 1 Questionnaire

90% of participants agreed that SMS has room for improvement. The reality is that SMSs different from SA-SAMS are not integrated with the DBE, and interoperability must be implemented to integrate these systems for a seamless information flow from one entity to another (Muriithi & Masinde, 2016).

• Interoperability has educational benefits that can help address the above-mentioned disadvantages of SMS. As these advantages include:



• Reducing the burden on staff members of the school to enter data

Figure 6-20: Q18 Delphi Round 1 Questionnaire

90% of participants agreed with the statement with interoperability, the burden on staff members of the school to enter data can be significantly reduced. Other experts in the comment section of this Likert statement, indicated that the burden on school staff members to enter data could only be significantly reduced if the SMS is a system that can be used at any time and is not limited by the disadvantages of the old version. Ed-Admin can be accessed from anywhere and anytime as per their website and experts' responses (Walsh, 2022). However, there is no evidence that this system is not limited by the disadvantages of the old version or legacy systems.

Cooper (2014) indicated that interoperability has educational benefits that can help address the disadvantages of SMS, thus reducing the burden on staff members to enter data. The literature and experts' results solidify the importance of interoperability in significantly reducing the burden on staff members to enter data.

## • Improving data quality



Figure 6-21: Q19 Delphi Round 1 Questionnaire

40% disagreed that interoperability can improve data quality. Despite that outcome, 60% of the participants support the statement. Cooper (2014) further identified another educational interoperability benefit: it improves data quality. Therefore, it is evident from the high percentage of experts who agreed with the statement and literature finding that interoperability can enhance data quality.



• Promoting adaptability and promote data sharing

Figure 6-22: Q20 Delphi Round 1 Questionnaire

70% of participants who understand how interoperability works and the benefits it provides support the statement. However, 30% of participants opted to neither agree nor disagree. With interoperability, systems can communicate and exchange data seamlessly. Rajkumar, Muzoora and Thun (2022) indicate that the benefits of interoperability are endless and further identify information sharing and adaptability (the capacity to be modified for a new use or purpose) as some of the benefits of interoperability. Both literature and expert results confirm that interoperability can benefit education by promoting adaptability and data sharing.



## • Supporting data-driven decision making

Figure 6-23: Q21 Delphi Round 1 Questionnaire

A total of 70% of participants support the benefit of interoperability. However, 30% disagreed or did not provide an answer to the statement. Rajkumar *et al.* (2022) further indicate that interoperability supports data-driven decision-making. The results showed that 20% of the participants neither agreed nor disagreed with the statement. This suggests that perhaps not everyone is aware of the benefits of interoperability in the education system and awareness must be made to enlighten people. Despite that, it is evident that both literature and expert results support the educational benefit of interoperability of data-driven decision-making.



• Enhancing efficiency and timeliness

Figure 6-24: Q22 Delphi Round 1 Questionnaire

A total of 90% of the participants supported this statement. However, 10% neither agreed nor disagreed. The comments section in the questionnaire, indicated that interoperability exists in SMSs different from SA-SAMS, especially in Ed-Admin. Among the many benefits of interoperability mentioned by Rajkumar et al. (2022), enhancing efficiency and timeliness is one of them. The literature and expert results confirm the usefulness of interoperability in the education sector, enhancing efficiency and timelessness.

• These advantages of interoperability can enhance SMS to ensure an improved overall information flow for effective decision-making and management of schools



Figure 6-25: Q23 Delphi Round 1 Questionnaire

This question was very tricky to answer especially when experts do not fully understand the benefits of interoperability. Nonetheless, 60% of participants supported the statement, 20% neither agreed nor disagreed, and 20% disagreed with the statement.

• Interoperability is the platform that provides continuous information exchange, for example, exchanging information that other systems can use to improve the performance, create services, control the processing of information systems and operations



Figure 6-26: Q24 Delphi Round 1 Questionnaire

A total of 70% of experts supported the statement, the benefits of interoperability are evident and are realised by many experts in the education department. The SMSIM model uses a combination of the Organizational Interoperability Maturity Model (OIM) and Information System Interoperability Maturity Model (ISIMM) to assess the level of interoperability in SMS and provide guidelines to get to the level (Maremi *et al.*, 2020). However, getting to the next level requires budget and commitment from the school to invest in what is needed to reach the suggested level of interoperability with their SMS.



• There is a need to conceptualize a model that will bring together interoperability in SMS

Figure 6-27: Q25 Delphi Round 1 Questionnaire

70% of participants agreed there is a need to conceptualize a model that will bring interoperability in SMS, and this is because interoperability is crucial for integrating all the systems. Unfortunately, interoperability also means the organization must allow access to their system from one department to the other and other departments prefers to work in silos to maintain privacy. This statement resulted in 20% disagreeing with it. This can be based on the disadvantages of interoperability, or the participants being delighted with the current SMS in use that they feel there is no need for improvements or SMS can be improved.

Experts who use Ed-Admin in the comment section of the questionnaire, indicated that interoperability educational benefit exists in Ed-Admin and not in SA-SAMS, as per their perception. This result motivated the need to test the validity, accuracy, and efficiency of the SMSIM model by assessing the level of interoperability in SMS.

Measures of central tendency (means, medians, and modes) as well as measures of dispersion, are the main statistics employed in Delphi studies (standard deviations and interquartile ranges)

(Hasson *et al.*, 2000; Hsu & Sandford, 2007). The use of Likert-type scores, based on median scores, is highly recommended in the literature (Hill & Fowles, 1975; Hsu & Sandford, 2007). According to estimates, the Delphi consensus varies between 55 and 100%, with 70% being the norm (Vernon, 2009; Avella, 2016). Means and standard deviations were calculated to quantify the level of agreement among experts. The mean of a data set represents its central tendency. In the analysis of a set of data, central tendency identifies the central position of the data within the set (Chakrabarty, 2021). The standard deviation measures how skewed the data are in relation to the mean. The values are near to the mean or expected value when the standard deviation is low. However, a high standard deviation suggests that the results are spread out over a wider range (Hargrave, 2022). Based on the mean and standard deviation, participants' perceptions were analyzed and grouped according to their Likert-scale scores.

The main purpose of the first round was to verify the problems of SMS and validate the need for a SMSIM model that can help improve SMS. The themes were converted into statements and questions and a Likert scale was used to rank these themes which developed a questionnaire, and the set agreement consensus was set at 55%. As observed from the figure below, questions about the disadvantages of SMS have a higher percentage than the set agreement consensus, and this validated the need for round 2 which will introduce a SMSIM model to experts for evaluation. As can be observed from the figure below, 70% of experts agreed that there is a need to conceptualize a model that will bring together interoperability in SMS, the overall feedback from the questionnaire also validated the need to for the SMSIM model and this made provision for the second iteration.

Furthermore, based on the std deviation of this figure, most experts favoured the "Strongly agree" and "Agree" ranks rather than the rest as it resonated more with their field of expertise and reality.

# Table 6-1: Iteration 1: Descriptive statistics

-

Questionnaires	Strongly	Agree	Neither Agree	Disagree	Strongly	No Answer	Mean	stdDeve	Percentage
	Agree		nor Disagree	_	Disagree				
School Management Systems (SMS) are large repositories that schools	5	3	0	0	2	0	1.666667	2.065591	80%
use to manage the day-to-day administrative activities of the school									
Multiple users use SMS; and these can be configured to meet the needs of	4	4	0	2	0	0	1.666667	1.966384	80%
each school independently			,	0		0	1 11117	2 250026	0.00/
SMS nelp to overcome monotonous paperwork in schools, are very easy	5	4	1	0		0	1.000007	2.250926	90%
to use, and can work accurately and smoothly in unterent scenarios. SMS									
feature workload, increases encicity in school management, and saves									
Many schools in South Africa use a SMS called SA-SAMS (South	3	4	2	1	0	0	1.666667	1.632993	70%
African School and Administration Management System) as mandated by	5	•		-		•	1.000007	1.052555	,,,,,
the South African Department of Education									
To make the administration of schools more effective, the DBE	4	4	1	0	1	0	1.666667	1.861899	80%
developed SA-SAMS to solve not only management and administration									
challenges but also to streamline data in schools.									
Not all schools in South Africa are using SA-SAMS, but their SMS	3	3	3	Û	0	i	1.666667	1.505545	60%
should easily integrate with SA-SAMS to reduce duplicate work									
Disadvantages of SMS includes: Increased workloads for teachers	2	5	1	0	2	0	1.666667	1.861899	70%
Disadvantages of SMS includes: Duplication of Work	3	3	0	3	1	0	1.666667	1.505545	60%
Disadvantages of SMS includes: Redundancy and errors	3	3	1	2	1	0	1.666667	1.21106	60%
Disadvantages of SMS includes: Delays for the submission of reports to	4	5	0	ì	0	0	1.666667	2.250926	90%
the DBE by schools									
Most schools use a free version of SMS which often lacks features such	3	3	2	1	1	0	1.666667	1.21106	60%
as discipline, custom reports, registration, and inventory. Some are									
harrowed by space capacity, the number of learners, and storage.	2	4	2	1	1	0	1 66667	1 26626	609/
Swiss are orden instance on one standarone computer, which minis access, which causes substantial tension and stress on the administrator working	2	4	2	1	1	0	1.000007	1.50020	00%
which causes substantial tension and succes on the administrator working									
as the end of the term when schools are required to process learner									
reports. This mostly results in annoving delays.									
Teachers have a small amount of access to the system for performing	1	4	0	3	1	1	1.666667	1.505545	50%
simple tasks like capturing marks and mostly consume teaching time by									
manually validating learners' marks on paper that are never-ending,									
increasing the chances of errors.									
Non-availability of skilled ICT personnel who can utilise the system.	2	4	1	1	2	0	1.666667	1.36626	60%
There is a lack of specific and inadequate availability of relevant									
software, lack of maintenance culture, and lack of data security and									
impurity on the government's side.		2		-		0			500/
SMS can only be accessed within the school's intranet.	2	3	1	3	1	0	1.666667	1.21106	50%
Limit tearners and parents or guardians from accessing the tearners	1	4	2	1	1	1	1.000007	1.21100	30%
Information SMS has mean for improvement due to its disadvantages	6	2	0	0	1	0	1 666667	2 42212	0.0%
Interpretability has educational benefits that can beln address the above-	4	5	0	0	1	0	1.666667	2.42212	90%
mentioned disadvantages of SMS_As these advantages include: Reducing	-	5	Ů	0	1	0	1.000007	2.230320	,0,0
the burden on staff members of the school to enter data									
Interoperability has educational benefits that can help address the above-	4	2	1	3	0	0	1.666667	1.632993	60%
mentioned disadvantages of SMS. As these advantages include:									
Improving data quality									
Interoperability has educational benefits that can help address the above-	6	1	3	0	0	0	1.666667	2.42212	70%
mentioned disadvantages of SMS. As these advantages include:									
Promoting adaptability and promote data sharing									
Interoperability has educational benefits that can help address the above-	4	3	2	1	0	0	1.666667	1.30384	70%
mentioned disadvantages of SMS. As these advantages include:									
Supporting data-driven decision making	-	2	, .	0	0	0	1 (((()))	2 72252	0.00/
Interoperationary has educational benefits that can help address the above-	/	2	1	0	0	0	1.000007	2.73232	90%
Inclution disadvantages of 5005. As these advantages include.									
These advantages of interoperability can enhance SMS to ensure an	4	2	2	1	1	0	1.666667	1.36626	60%
improved overall information flow for effective decision-making and	,	-	<u>_</u>	1	1	5		1.50020	0070
management of schools									
Interoperability is the platform that provides continuous information	5	2	1	2	0	0	1.666667	1.861899	70%
exchange, for example, exchanging information that other systems can use									
to improve the performance, create services, control the processing of									
information systems and operations									
There is a need to conceptualize a model that will bring together	4	3	1	2	Û	Û	1.666667	1.632993	70%
interoperability in SMS									

#### 6.3.2.2 Ed-Admin

During the first round of the Delphi questionnaire, various experts highly mentioned Ed-Admin as a School Management System (SMS) that is interoperable and 100% better than SA-SAMS. Based on this response, the researcher felt the need to investigate this SMS further. It was also important to use the SMSIM model to assess the level of interoperability in the Ed-Admin, which aimed to evaluate if the SMSIM model is accurate, is of quality and its ability to deliver what it intends to achieve. The following section discusses a summary of the Ed-Admin based on the information provided on the Ed-Admin website, a complete reflection of what the system can do can only be provided by the company to the client.

Ed-Admin is an all-in-one School Management Software. It is a powerful and intuitive learning management software, that offers a complete package for schools. Schools, Higher Education Institutions and Government, can use it as their core goal is to help institutions grow (Göpper, 2020; Walsh, 2022). This management software is designed to scale with the needs of universities and colleges. This one-platform design allows the school's information to always be at your fingertips. In addition, Ed-admin, as SMS, will enable institutions to coordinate and network effectively. Every element, from student performance to principal reports, is traceable and customizable to suit the school's needs.

With Ed-Admin, all the resources required to manage funds are kept in one location. At your fingertips are everything from customer records and employee payroll to tuition and application fees. Batch payments, automatic statements, and payment gateways are tooling the school can employ to save time and please parents, clients, and suppliers. By integrating the school's admission module with Ed-website, Admin's the admission process may be easily handled from inquiry to enrolment, and automated procedures from enrolment to re-enrolment to online fee collection can be enjoyed. Additionally, it gives the institution the chance to use analytics to expand its academy. With comprehensive analytics monitoring, schools can pinpoint their pupils' strengths and problems.

• More than 40 Modules: More than 40 modules make up Ed-Admin, which covers every area of managing education, from finances and academic resources to personnel administration and communication. Figure 6-8 shows the 40 modules of Ed-Admin.




Figure 6-28: Modules of Ed-Admin

• Why Choose Ed-Admin Learning

The figure below provides valuable reasons why schools should choose to buy Ed-Admin as a SMS:

Figure 6-29: Why choose Ed-Admin

From Figure 6-29, it is evident that this SMS presents the other systems and applications Ed-Admin is integrated with.



Figure 6-29: Ed-Admin represents other systems and applications – therefore it was selected



**Edana Parent Portal App** Through our Parent Portal App, you can access all the information relating to your child's school journey.

- Receive updates from your school
- Monitor your child's progress
- Follow your child's activities
- Communicate with teachers
- · View termly reports
- Access documents shared by the school

# **Integrated with**



Edana Staff Portal App The Staff Portal App allows teachers and administrators to view and edit information related to classes, student profiles and more.

- Communicate with students instantly.
- Easy-to-use-interface
- Important student into at a glance
- Simple grading function
- Manage assessment and attendance
- Health Screening functionality



Edana Student Portal App Give students everything they need to manage their education. Students can attend online classes, viewing class schedules and submit assignments, all from their phones.

- Access profile information •
- Check class schedule & timetables
- Receive newsletters
- Remove learning
   experience
- Virtual classes

reminders

Class & assignment



Edana Alumni Portal App Stay in touch with past students with our Alumni Portal app. Alumni can manage their own data and keep connected with their former institutions.

- Engage alumni in meaningful discissions
- Share news and promote ventures
- Create a stronger bond between alumni
- Announce events, reunions, and gatherings

# Figure 6-30: System Integrating with Ed-Admin

Ed-admin is integrating with other systems as seen in Figure 6-30, these systems allow Ed-Admin to be accessible using mobile apps from anywhere and at any time. Figure 6-31 below shows more systems integrating with Ed-Admin to enhance and better the School Management System (SMS).



#### Figure 6-31: Systems Integrating with Ed-Admin

Systems and applications in Figures 6-32 are included in Ed-Admin by request. This makes Ed-Admin customisable to schools as needed.



#### Figure 6-32: Integration

Integrations allow users to use all their favourite tools right from the Ed-admin interface. Streamline, sync, and even automate their tasks to make school management easier than ever.

More information about Ed-Admin can be obtained on their website (Ed-Admin, 2022).

Another SMS used by other schools is Saspac, another School Management System mentioned by some of the experts who participated in the study. The Saspac ERP for schools is created, maintained, and supported by Saspac, a company founded in 1979. Saspac ERP offers schools a comprehensive solution for both business and educational administration needs because it is both robust and simple to use (Göpper, 2020). This SMS is also not integrated with SA-SAMS. However, they export what is needed for the department into SA-SAMS. The section below presents the round 2 Delphi questionnaire.

# 6.4 USING THE SMSIM MODEL TO ASSESS ED-ADMIN

The results of the qualitative Delphi method in iteration one, identified Ed-Admin as a fully interoperable SMS that address disadvantages that other SMS have. This necessitated testing the SMSIM model for accuracy, validity, worth and efficiency. To do this, the SMSIM model was used to test Ed-Admin to identify the level of interoperability in the SMS and to provide guidelines on achieving the next level of interoperability. The section below, presents Ed-Admin and its level of interoperability. The accuracy of this section relies on the information provided on Ed-Admin website and can be further adjusted when implemented in schools based on the needs.

Chapter 4 detailed the SMSIM model and the steps that occur in this level. Section 6.5.1.2 of this chapter shows the final SMSIM model. According to the levels in the model, Ed Admin falls on the Intermediate level of interoperability. There is no evidence on the website that identifies and discusses the use of Ed-Admin to communicate with other schools. Ed-Admin is mostly used for school management rather than communication, thus this SMS is assessed and identified on the intermediate phase of the SMSIM model.

In the intermediate level, databases and applications are separate. Although there is basic data collaboration with other schools, no data is shared, and logical data models are shared. Communicating with other schools and sharing knowledge on specific topics is possible. The Ed-Admin website makes mention of how it can help institutions coordinate and network more efficiently. In contrast, it does not mention how institutions can coordinate and collaborate. Ed admin does not qualify in the secondary phase as this phase requires system integration between organizations. The tertiary phase is a long-term goal for many organizations worldwide, requiring full interoperability. Below is a picture view of where Ed-Admin lies in the SMSIM model.



Figure 6-33: Level of interoperability in Ed-Admin

After the first round of Exploratory Delphi questionnaires a report was prepared and anonymously provided to the experts to reflect and change or adjust their initial responses if required and from this report, all experts did not need to change their initial responses, instead all experts were happy with the report and their initial response and all reached a consensus that round 2 that evaluate the SMSIM model is required. The report is attached in appendix E.

# 6.5 EXPERTS' RESULTS IN ITERATION 2

The section below presents the Delphi round 2 questionnaire questions with the results:

#### 6.5.1 Round 2 of the Delphi Questionnaires

The first round of the Delphi study reached a 70% consensus that necessitated round 2 that eill focus on verifying and validating the SMSIM model (see Appendix E). The statements and questions in this section focus on the practical expertise of the subject matter experts, this section aims to unpack problems in SMS and how it can be improved using interoperability. When statements and questions were developed in this questionnaire, it was important to ensure that the round 1 questionnaires informed the round 2 questionnaire statements and an evaluation of the literature finding was conducted throughout the research to verify and validate the model based on determining the merit, worth, value, or significance. The section below presents the statements and questions of the Delphi round 2 questionnaires.



• No doctrines are put in place that enables SMS to cooperate with other SMS.

Figure 6-34: Q1 Delphi Round 2 Questionnaire

The main purpose of this statement was to confirm the lack of interoperability in SMS across schools. As per the results, 70% of the experts agreed with the statement, while the other 30% strongly agreed with the statement. The reality is that, although some schools in South Africa have a good SMS that is interoperable and can be accessed from anywhere in the world, and this was confirmed with round 1 questionnaires, access is still restricted to only the members of the school. Any other organization does not have a way rather than manual means to communicate with feeder schools or other schools. In this case, the SMSIM model places most schools in the foundation phase of interoperability, where systems are not connected, and communication is done partially manually, as in today's world, emails and telephones are used for communication.

 Schools cannot use SMS to share information and communicate, importing data into SA-SAMS is done manually.



Figure 6-35: Q2 Delphi Round 2 Questionnaire

The statement above focussed on unpacking the ability of SMS to interoperate with other SMS. This boiled down to communication and information sharing. 40% of the experts agreed with the statement, whereas 60% of the experts strongly agreed with the statement. Based on the results, it is evident that South Africa has a long way to go in terms of integrating its education systems. Interoperability is the ultimate integration solution (Ostadzadeh *et al.*, 2015). When organizations cannot share information and communicate, Margariti et al. (2020) place these organizations at an adhoc level, the first level of interoperability maturity. The SMSIM model indicates that finance and legacy systems are the main barriers to interoperability.



• Schools don't use SMS to share information and communicate with other schools.

Figure 6-36: Q3 Delphi Round 2 Questionnaire

The past two statements and this statement are the same in that they unpack the interoperability of SMS. While the passed statement focused on the inability of schools to share information and communicate, this statement focuses on utilizing SMS to share information and communicate with other schools. This resulted in 50% of experts agreeing to the statement and the other 50% strongly agreeing to the statement. As in round 1 of the questionnaires, some experts who use Ed-Admin confirmed that the SMS does not have many of the disadvantages SA-SAMS has. Their SMS is interoperable to some degree, but unfortunately, the information sharing issue is still open for improvements. SMSs in South Africa are not used as a means of communication and information sharing. This validates the need for a SMSIM model that can assess SMS and provide guidelines for integrating SMSs.



• In most schools, SMS is installed in a standalone computer with limited access.

Figure 6-37: Q4 Delphi Round 2 Questionnaire

10% of the experts disagreed with the statement, this was evident even in the round 1 questionnaire as other schools use Ed-Admin, which can be accessed from anywhere in the world. This completely falsifies this statement in such an incident. However, based on the 70% of experts who strongly agreed with the statement and 20% who agreed, SMS has a long way to go in implementing interoperability. Ed-Admin is a good paid SMS that other SMSs can use to learn from. When assessed using the SMSIM model, Ed-Admin laid on the intermediate phase, which is a good level of interoperability.

• All the other systems and applications that exist in schools are most probably not connected to SMS.



Figure 6-38: Q5 Delphi Round 2 Questionnaire

This statement resulted in 50% of experts agreeing and the other 50% strongly agreeing with the statement. As it can be observed, SMS can be interoperable on its own to a certain degree. However, it does not communicate with other systems and applications in the school. The SMSIM model provides guidelines that can help schools know what is required to get to the next level of interoperability. This is valuable because interoperability encourages system integrations; when systems are integrated, communication and information sharing become easy (Bone, Blackburn, Kruse, Dzielski, Hagedorn & Grosse, 2018).

• The systems in the schools are most probably unable to share data and communicate with each other.



Figure 6-39: Q6 Delphi Round 2 Questionnaire

This is another statement that has a higher percentage of experts strongly agreeing with the statement. The reality is that systems in the schools are still separate, and there is no communication among them. However, SMS such as Ed-Admin reduces workloads on staff members as it can be accessed from anywhere (Walsh, 2022). Despite this fact, systems are still unable to interoperate. The SMSIM model offers guidelines for achieving system integrations not only to SA-SAMS but also to other systems in the school.

• With the current state of SMS, the SMSIM model assesses the level of interoperability and shows guidelines for achieving the next level to address the disadvantages of SMS.



Figure 6-40: Q7 Delphi Round 2 Questionnaire

All experts who participated in the study, supported this statement. They agreed that the SMSIM model is able to assess the interoperability of SMS and provide guidelines as it intends to do, which is a good overall result that verifies and validates the accuracy, relevancy, merit and worth of the SMSIM model. Moreover, it shows that the model does what it intends to do.

 Is the SMSIM model effectively and accurately able to assess the level of interoperability in SMS and provide guidelines on how to get to the next level?



Figure 6-41: Q8 Delphi Round 2 Questionnaire

This question resulted in 90% of experts in support of the question. The 90% is formed by 60% of experts strongly agreeing with the posed question and 30% of experts who agreed with the statement. Without a reasonable doubt, experts in different fields of SMS and ICT in education, as well as interoperability as displayed in the demographic section of this chapter, agreed that this model is effectively and accurately able to do what it intends to do.

• Is the SMSIM model a valid model for providing guidelines to address the current problems of SMS?



Figure 6-42: Q9 Delphi Round 2 Questionnaire

The result of this question is also satisfactory, with an overall agreement of 90% of experts supporting the validity of the model. As observed from the figure above, 70% of experts strongly agreed with the statement. In the comment section, experts felt,

"It is essential to share this model with DoE - they urgently need to fix SA-SAMS. This model is a good solution".

"This model of yours can be regarded as a possible solution- I think it is innovative and well-researched".

"Great model and worth pursuing to overcome current interoperability problems"

Thus it can be worthwhile to utilise this model to potentially solve currently experienced SMS problems.



#### • SMS will be improved when utilizing this model

Figure 6-43: Q10 Delphi Round 2 Questionnaire

As with other questions about the model, this question also resulted in 90% of experts agreeing to the posed question. And the comment above bears' reference. This model is seen as a good model to overcome the current SMS problem. Moreover, it is said to be most valuable for improving SA-SAMS.

As observed from Table 6-2 below of iteration 2, the overall percentage is higher than the set agreement of consensus. Most experts agree that the SMSIM model is relevant to solve current SMS problems, specifically, problems in SA-SAMS. The overall purpose of this iteration was

to validate and verify the SMSIM model based on relevancy, accuracy, worth, validity, and efficiency to solve current problems experienced with SMS of schools in South Africa. Like iteration 1, themes were converted into statements and questions to develop the round 2 questionnaire, and the consensus of the agreement was set at 55% and a consensus for round 2 was reached at 100%.

The last question on how the model can be improved was answered by only one expert, indicating that the model can be simplified. The researcher felt that if the detailed information within the model is deleted, it may not be understood how the detail of the middle column is supported by the depth of each of the selected models on the side.

# Table 6-2: Iteration 2: Descriptive statistics

	Strongly Agree	Agree	Neither Agree	Disagree	<b>Strongly Disagree</b>	No Answer	Mean	stdDeve	Percentage
Questionnaires			nor Disagree						
Schools cannot use SMS to share information and communicate,									100%
importing data into SA-SAMS is done manually.	6	4	0	0	0	0	1.666667	2.65832	
Schools don't use SMS to share information and communicate with									100%
other schools.	5	5	0	0	0	0	1.666667	2.581989	
In most schools, SMS is installed in a standalone computer and									90%
there is limited access.	7	2	0	1	0	0	1.666667	2.73252	
All the other systems and applications that exists in schools are									100%
most probably not connected to SMS.	5	5	0	0	0	0	1.666667	2.581989	
The systems in the schools are most probably unable to share data									100%
& communicate with each other.	7	3	0	0	0	0	1.666667	2.875181	
With the current state of SMS, the SMSIM Model assesses the									100%
level of interoperability and shows guidelines for achieving the next									
level to address the disadvantages of SMS.	7	3	0	0	0	0	1.666667	2.875181	
Is the SMSIM Model effectively and accurately able to assess the									100%
level of interoperability in SMS and provide guidelines on how to									
get to the next level?	6	3	1	0	0	0	1.666667	2.42212	
Is the SMSIM Model a valid model for providing guidelines to									90%
address the current problems of SMS?	7	2	1	0	0	0	1.666667	2.73252	
SMS will be improved when utilizing this model	7	2	1	0	0	0	1.666667	2.73252	90%

# 6.6 THE FINAL SCHOOL MANAGEMENT SYSTEMS INTEROPERABILITY MATURITY (SMSIM) MODEL



Figure 6-44: School Management Systems Interoperability Maturity (SMSIM) model

At the end of Round 1 of the Delphi expert reviews, the SMSIM model was tested by assessing Ed-Admin's current level of interoperability and then providing guidelines for advancing to the next level. The result of the assessment is based on the information gathered from the Ed-admin website, experts who directly interact with Ed-Admin may use the SMSIM model more efficiently to assess the level of interoperability in Ed-Admin and obtain guidelines to get to the next level of interoperability, and this can be seen in the final version of the SMSIM model as presented in the figure 6-44 above.

Based on the results from both Delphi expert review results, the final SMSIM model did not change or was not adapted. This model has been validated by experts for integrity, validity, accuracy, and efficiency. Based on the results, the model addresses the current SMS problems for South African schools, mainly SA-SAMS. Guidelines for moving to the next level of interoperability were provided in Chapter 4 Section 4.7. Furthermore, a summary of these guidelines is provided in section 6.7 below.

Experts were provided with the report for the round 1 questionnaire, and this report included the use of the interim SMSIM model to assess the level of interoperability in Ed-Admin and provided guidelines on what is required to move to the next level of interoperability. As a result, the model met its intended purpose and was found to be valid, accurate, and efficient. The description of the model can be found in Chapter 4, Section 4.7. This will not be repeated here as it was already provided, and the Delphi results did not add any changes to the interim model. The next section provides a refined summary of the guidelines for implementing the SMSIM model.

# 6.7 SUMMARY OF THE SMSIM MODEL GUIDELINES

The section below presents a summary of guidelines for SMSIM model, which can be used by schools in South Africa to improve SMS. The guideline in this section and the ones in Chapter 4, Section 4.7 are the same. Nothing was changed as experts found that the initial guidelines met the requirements of improving SMS. These guidelines were further used to evaluate the level of interoperability in Ed-Admin, see Section 6.4:

• Foundation phase requires the guidelines below to achieve the primary phase of interoperability maturity level:

- Communication Interoperability such as:
  - Basic data sharing.
  - Separate application & databases.
  - Point to point data exchange.
- General Guidelines of achieving interoperability.
- Electronic communication such as:
  - Telephones.
  - Emails.
- A shared purpose with other schools.
- The primary phase requires guidelines below to achieve the intermediate phase of interoperability maturity level:
  - Software Interoperability that enables:
    - Broader connections to legacy systems.
    - Basic collaborations.
    - Logical data models that are shared.
  - General doctrine put in place and some experience.
  - Shared communication and knowledge about specific topics.
  - Shared purpose and a shared goal with other schools and SMS.
- The intermediate phase requires the guidelines below to achieve the secondary phase of interoperability maturity level:
  - Data Interoperability that enables:
    - Data to be shared to some extent.
    - Collaboration at an advanced domain level.
    - Data is exchanged between independent applications.
    - System integrations is implemented between organizations.
  - Detailed doctrines & experience in using it.
  - Shared communications and shared knowledge.
  - One chain of command and integration with home organizations.

- Secondary phases require the guidelines below to achieve the **tertiary phase** of interoperability maturity level:
  - Data that is fully distributed between organizations.
  - Collaboration that is on an advanced enterprise level.
  - The organization is at an advanced enterprise level.
  - Fully interoperable front and back-office systems.
  - Data that has a common interpretation.
  - Fully shared data and applications.
  - Automated processes.
  - Complete and normal day-to-day working.
  - A shared understanding with other organizations.
  - Homogeneous systems.
  - o Uniform ethos.

The tertiary phase is the last level of interoperability maturity and at this level, there is no further improvements required. See Section 6.6 for more details. After using the exploratory qualitative design of Delphi, it became evident that this is a good technique to collect data without face-to-face contact and without jeopardizing the integrity, quality, and accuracy of the study. This technique assisted in conceptualizing the SMSIM model by reaffirming literature findings and experts' expertise. Thus, the SMSIM model was successfully assessed, verified, and validated to achieve what it intends to achieve. More information about the design can be found in Chapter 5, Section 5.7.

# 6.8 POSSIBILITIES OF ADDRESSING SMS PROBLEMS USING SMSIM MODEL

Each level of the SMSIM model has guidelines that shows what is required, and to advance to the next level, the SMS must meet the requirements of the next level as per the specified guidelines. In guideline one for example which lies in level 1 of foundation phase in the SMSIM model, schools share data with the DBE manually through USB, CD etc, this will apply for schools without internet access as depicted in a diagram (Figure 3-2) in chapter
 Also, there is increased workloads for teachers because the SMS can only be accessed on one computer as a results teachers are manually capturing marks and then providing the

to the system administrator who has to later on capture into the SMS and this creates a lot of erros and redundancy due to human error. In this level, a lot of problems arise that are caused by the actual SMS and others by the under resourced school as an organisation itself. The systems in the school are not connected nor with the DBE and the staff members are not prepared for the changes, not in skills development nor in understanding of what the technology advancements will do. The SMSIM model intends to guide the school on seeing and knowing the level of interoperability the school is on. This provides a perspective to the school on what they need to do to advance to the next level of interoperability. In guideline two for example, which lies on the primary phase in the SMSIM model, schools can share basic data, and this can be seen through many SMS that has tools and applications to conduct a lot of administrative tasks in the system, although data is not shared with other schools. Other SMSs will meet the requirements of level threeintermediate phase which at a click of a button, schools are able to submit data to the DBE. With every advancement to the next level of interoperability, problems of SMS listed in the chapter 1, section 1.2 are resolved. School can thus use this model to determine the level of interoperability they are on and see what the next level requires, then schools can formulate a plan on how to financially budget to meet the requirements of the next level. When there is a sufficient budget set aside for improvements, then updates can be made on SMS and the school itself-this include skills development of the staff members especially in the rural development. Training can be conducted so changed is easily embraced and handled. There are few certainties in life, but one of them is that change is inevitable (Sampathkumar, 2020), progress is impossible without change and those who cannot change their minds cannot change anything. However, change disrupts this comfort creating a situation filled with anxiety and uncertainty among change recipients (Issah, 2018). The guidelines provided in the SMSIM model can ease uncertainty that comes with technology advancements.

#### 6.9 SUMMARY

In conclusion, the outcome of the Delphi study Round 1 questionnaire suggested that SMS requires improvements, especially SA-SAMS. All the experts who participated mentioned that SA-SAMS is problematic and requires a lot of improvements. It is also highly costly for private schools that

already use their paid version of SMS, and this is a result of features and changes made by the Department of Basic Education to the goalposts for data needed for SA-SAMS. A SMSIM model can thus be a valuable tool for improving SMS in schools in South Africa. Round 2 of the Delphi Study questionnaires focused on the SMSIM model that can be used as a guideline for solving SMS problems. A consensus was reached after two rounds of the Delphi questionnaire, and thus, a SMSIM model was successfully conceptualized.

# CHAPTER 7: CONCLUSION, REFLECTION AND RECOMMENDATIONS.

#### 7.1 INTRODUCTION

Chapter 6 discussed and analyzed the results of the expert review questionnaires for the round 1 and round 2 Delphi study. The School Management Systems Interoperability Maturity (SMSIM) model was further conceptualized, verified, and validated based on the expert review results. To reflect on the process undertaken to conceptualize this SMSIM model, this chapter provides a reflection on how the research questions and objectives were addressed and recommendations that can be made for implementing this model. This is accomplished by first outlining the goals and research questions that guided this study. This chapter next considers the methodological contributions. The research's drawbacks and suggestions for further investigation are then discussed. The third phase of the research process is covered in this chapter:



Figure 7-1: The research process and how this chapter supports Phase 3 of the study

#### 7.2 RESEARCH OVERVIEW

During the COVID-19 Pandemic, schools had no choice but to close all the gates and start finding ways to ensure learning continuity by utilizing digital platforms (Mhlanga & Moloi, 2020). This showed the power of technology when used properly. With the help of the SMSIM model, the level of interoperability in SMS can be improved and with every interoperability maturity level, problems of SMS may be addressed. Change is not easy and does not come cheap, but it's worth it (Ayala, 2022). Therefore, the experts who participated in the study are from different fields of expertise as per section 5.7.3 to provide feedback through the Exploratory Delphi questionnaire for a SMSIM model that will not only assess the level of interoperability in SMS but to also solve SMS problems potentially. The next section provides a perspective of how the research questions of this study were addressed.

# 7.2.1 Addressing the Research Questions

**Main Research Question:** How can a SMSIM model assist schools in improving the interoperability and maturity levels of their current SMS?

The main research question is addressed by the fact that the SMSIM model can be used by any school in South Africa to improve SMS. This model assesses the level of interoperability in SMS and provide guidelines on achieving the next level of maturity. Developers can use this model to build an evaluation feature that sets SMS at different level of maturity. Schools can use this model to check which level of interoperability their SMS is currently at and refer to the guidelines to determine what is needed to improve their SMS, In section 6.8 more information on how schools can use the SMSIM model to improve SMS in provided. This MRQ is also addressed through the answers to the three SRQs and will be addressed after the findings of the three SRQs.

**SRQ 1:** What constitutes the components of a model of interoperability maturity for Schools Management Systems?

This sub question 1 was addressed by the different phases in the SMSIM model, such as the foundation phase, primary phase, intermediate phase, secondary phase, and tertiary phase. These phases at each level feed into Organizational Interoperability Maturity Model (OIM) and

Information System Interoperability Maturity Model (ISIMM) to assess the level of interoperability at each level and provide guidelines for reaching the next level, which were presented in Chapter 6.

**SRQ 2**: What existing models of interoperability maturity can be used to conceptualize a SMSIM model?

This sub question 2 was addressed by the different Interoperability Maturity Model (IMM) listed and discussed in Chapter 4, but mostly the selected IMM (OIM and ISIMM) was used to conceptualize the SMSIM model found in Section 4.6.

# SQR 3: What value will a SMSIM model have for schools using Schools Management Systems?

This sub question 3 is addressed by the expert review findings in Chapter 6 and comments provided by the experts. The conceptualized SMSIM model (Chapter 6) assessed interoperability and provided guidelines to help improve SMS. The SMSIM model also solve the problem statements reflected in Chapter 1. When schools use this model to improve SMS, they can realise the benefits of interoperability as documented in section 1.2. The SMSIM model may potentially address the problem in SMS as schools achieve higher interoperability maturity level. This may result in all SMS that could become interoperable and could be accessed from anywhere at any place instead of being accessed only in the school premises using the school's intranet. Finally, all users using SMS could be trained to assess the interoperability maturity levels of their SMS by using the SMSIM model.

In summary, the main research question is addressed by the sub questions and the outcome of the expert reviews, which conceptualized the verified, and validated SMSIM model. The SMSIM model can assist schools in improving the interoperability and maturity levels of their current SMS by a feeder that goes into each level to assess the level of interoperability and provide guidelines for advancing the SMS to move to the next level.

#### 7.3 THE RESEARCH CONTRIBUTIONS

This research aimed to conceptualize a SMSIM model that will assess the level of interoperability in SMS and provide guidelines on achieving the next level of interoperability to improve SMS. This model will guide not just SMS in private schools but SMS worldwide to improve SMS.

The following sections describe the contributions of this study both theoretically and practically.

# 7.3.1 Theoretical Contribution

The Institutional theory was applied as the theoretical lens in this research to define the setting and contextual influences present when schools use SMS. Through this theory, the research was able to understand the complexities associated with integrating SMS as an IS/IT-based solution (through the conceptual model) in the context of the educational environment.

#### 7.3.2 Practical Contribution

The SMSIM model has been conceptualized to improve SMS and has well-defined, verified, and validated guidelines that can be used to improve SMS practically. This model is designed to contribute to theory in the form of literature and solve real-time problems, as highlighted in Chapter 1 of the study. The SMSIM model is suitable for all schools worldwide that would like to assess the interoperability level of SMS and obtain guidelines on what is required to get to the next level of interoperability to improve SMS. The expert reviews also alluded to this. In addition to this, the software developers of the SMS can use this model to build in a component on evaluating the interoperability maturity of SMS.

# 7.4 LIMITATIONS OF THE RESEARCH

There are several limitations that occurred during the research as listed below:

• Availability to participants: Originally, this research targeted the Principals, ICT Coordinators, Administrators of the school and Experts, however, due to the negative effect of the COVID-19 Pandemic that restricted participants' availability and access to schools, this research only focused on experts using the Delphi Questionnaire. Numerous attempts were made to get access to the school Principals to conduct interviews however only a few

responded with a decline, others opted to remain silent and only when I sent a follow up email of thanking them for accepting to participate in the study did, they respond with a decline. A few of the school Principals indicated that due to the negative impact of the COVID-19 pandemic, the targeted participants have a lot to do, and it will be inconsiderate add to their workload by asking them to participate in the study. Interview through remote access was not even anything to consider as a "no" was provided regardless of how the interview will be conducted. There was a lot of begging as well to get the experts to also participate in the study, I had to promise them that there won't be any interviews, they will participate in a Delphi study which will allow them to complete the questionnaire at the comfort of their own homes.

- **Quantity:** The initial plan was to interview six (6) principals, six (6) ICT Coordinators, six (6) administrators, and six (6) experts, however due to the inability to access schools, this research only used 10 experts' reviews by applying a Delphi technique.
- **Data collection:** This research used the Delphi method to elicit the reviews of experts, allowing the researcher to reach participants from different locations without needing face-to-face contact.
- **Time:** Initially, the researcher aimed to conduct the research in a two (2) year's period, however, due to the COVID-19 Pandemic, this period was extended to four (4) years.

The following section elaborates on the lesson learned and the personal reflections during the research. This section aims to reflect on the experience of the researcher during the research and what could be done differently should an opportunity arise to conduct the research again in future.

# 7.5 LESSON LEARNED AND PERSONAL REFLECTIONS

This research was motivated by the need to address the problems schools are encountering when using SMS, and the need to positively contribute to SMS by conceptualizing a SMSIM model that can assess the level of interoperability and provide guidelines for achieving the next level of interoperability to improve it. The impact of this research can be used to potentially improve SMS of schools in South Africa by maturing the interoperability of SMS and by hopefully addressing the problems in SMS as the maturity level advances. Without a clear procedure shown by the research methodology used across all chapters of this research, the outcome of this study would

not have been accomplished. The research methodology was critically important to guide the research, collect the data using the Delphi method, engage with participants and interpret the literature findings and experts' reviews.

There were great difficulties encountered during this research and conceptualizing the SMSIM model was certainly not easy. The COVID-19 Pandemic restricted access to schools and thus, interacting directly with the anticipated target group of participants (Principals, ICT Co-ordinators, and Administrators) was impossible. With time, the COVID-19 pandemic restrictions were lifted and there was hope to gain access to schools, but the negative effect of the pandemic remained. All the schools which were asked to participate in the research, either did not respond to the request or completely refused to partake in the study as they had indicated that "there is a lot of work to do and catch up on, and so little time", "Staff members are burdened with a lot of work, it is impossible to expect them to additionally add more work to a very hectic workload". Unfortunately, my progress was derailed, not only by the negative impact of the COVID-19 pandemic on restrictions but also emotionally by the people around me who lost their lives due to this, the worry of the unknown, who will lose their lives next, who will be infected, should I contract the virus, will I be able to survive it? In future, I will ensure I find a balance to deal with not only the hardships that life may bring but any natural disaster that may come my way.

Through this research, I have learned the importance of investing myself in literature to understand the context of my research and ensure the sources are recent. This process has really improved my thinking and my views on research in general. I feel that my mind has been broadly opened and I value other authors who also went through this journey to write and publish papers, books, and articles that really helped me support the problem statement and build a base for this research. I have also learned the importance of the methodology in the research which provided guidance, structure, and direction to the research.

Most importantly, I have learned the importance of having support from family members at home who understand the need to work after hours and the support from supervisors. Although they were exposed to many students, and despite also being negatively impacted by the COVID-19 Pandemic, they took the time to go through my work and provide fruitful feedback that helped

build and shape the outcome of this research. Working with my two (2) supervisors has been an honour, and I learned a lot from them.

#### 7.6 RECOMMENDATIONS FOR FUTURE STUDIES

Literature findings proved that there are schools, especially in the private sectors, that are using different SMSs from SA-SAMS, but the biggest problem is system integration as data must be manually imported into SA-SAMS, it is recommended that a new study on SA-SAMS can be conducted based on the findings in this research and to add the model from this research to the SA\_SAMS for improvement. Future research can then focus on how the DBE can use interoperability for integrating systems in the Department and different SMS used by all schools in South Africa to ensure reporting is conducted with a click of a button. Another recommendation is that a study can be done to adjust SA-SAMS to allow for the best features found in the best SMS discussed in this study. After adding the best features, SA-SAMS can then be piloted to determine its maturity to address interoperability.

# 7.7 CONCLUSION

• This exploratory qualitative study identified two Interoperability Maturity models (IMM) from literature findings that could be used to conceptualize a SMSIM model. The conceptualized SMSIM model was then assessed to ensure it could solve currently identified SMS problems and to be certain that the model does what it is intended to do. For this purpose, expert reviews were conducted following the exploratory Delphi design. The feedback and insights obtained from these reviews allowed for the refinement of the conceptualized SMSIM model for SMS and guidelines were provided to use this model. Through this chapter, the researcher considered the value and significance of the research, personal learning objectives, the limits discussed, and suggestions for further research. Lastly, the results of the study regarding SA-SAMS identified this system as a flawed system that would drastically benefit from the SMSIM model, this outcome could be a result of not having the targeted participants (Principals, ICT Coordinators, and Administrators) in the study. Only one deputy principal and administrator participated in the study, the rest of the participants were IT managers etc. as listed in section 5.7.5. The aim of this research is not to downgrade any SMS but to help improve it by assessing the level of interoperability in SMS and provide

guidelines to advance to the next level and the conceptualised SMSIM model is for all SMSs.

# REFERENCES

- Adam, L., Butcher, N., Tusubira, F., Sibthorpe, C. & Souter, D. 2011. Transformation-Ready: The strategic application of information and communication technologies in Africa. In: (ed.). African Development Bank, the World Bank and the African Union Africa. Available from: https://milunesco.unaoc.org/wp-content/uploads/2012/12/Education Fullreport.pdf.
- Adarkwah, M.A. 2021. "I'm not against online teaching, but what about us?": ICT in Ghana post Covid-19. In: (ed.). Available from: <u>https://link.springer.com/article/10.1007/s10639-020-10331-z</u>.
- Adebayo, A.O., Kanyinsola, O. & David, T. 2014. Patient record management information system. *Researchjournali's Journal of Information Technology*, 1(1):1-23. Available from:<u>https://researchjournali.com/pdf/669.pdf</u>.
- Adnan, M. & Anwar, K. 2020. Online Learning amid the COVID-19 Pandemic: Students' Perspectives. *Online Submission*, 2(1):45-51. Available from:<u>https://doi.org/10.33902/JPSP.2020261309</u>.
- Agrawal, A. & Alharbe, N.R. 2019. Need and Importance of Healthcare Data Integrity. *International Journal* of Engineering and Technology (IJET), 11(4):854-859. Available from:<u>https://www.enggjournals.com/ijet/docs/IJET19-11-04-033.pdf</u>.
- Ahmad, S. & Wong, K.Y. 2019. Development of weighted triple-bottom line sustainability indicators for the Malaysian food manufacturing industry using the Delphi method. *Journal of Cleaner Production*, 229:1167-1182. Available from:<u>https://doi.org/10.1016/j.jclepro.2019.04.399</u>.
- Akbaba Altun, S. & Bulut, M. 2021. The Roles and Responsibilities of School Administrators during the Emergency Remote Teaching Process in COVID-19 Pandemic. *Research in Educational Administration* & *Leadership*, 6(4):870-901. Available from:https://doi.org/DOI:10.30828/real/2021.4.4.
- Akbaba Altun, S. & Bulut, M. 2023. How did school administrators manage the crises during the COVID-19 outbreak? *Journal of Global Education and Research*, 7(3):234-248. Available from:<u>https://digitalcommons.usf.edu/jger/vol7/iss3/4/</u>.
- Akter, N. 2018. A web-based system for school management. Project Report. Dhaka, Bangladesh: DaffodilInternationalUniversity.from:<a href="http://dspace.daffodilvarsity.edu.bd:8080/handle/20.500.11948/2555">http://dspace.daffodilvarsity.edu.bd:8080/handle/20.500.11948/2555</a>.
- Al-Araibi, A.a.M., Mahrin, M.N.R.B. & Yusoff, R.C.M. 2019. Technological aspect factors of E-learning readiness in higher education institutions: Delphi technique. *Education and Information Technologies*, 24(1):567-590. Available from:<u>https://doi.org/DOI:10.1007/s10639-018-9780-9</u>.
- Al-Busaidi, K.A. 2014. SWOT of social networking sites for group work in government organizations: An exploratory Delphi study from IT managers' perspective. *VINE: The journal of information and knowledge management systems,* 44(1):121-139. Available from:<u>https://doi.org/DOI:10.1108/VINE-06-2012-0019</u>.
- Al-Hayek, A. 2016. *E-School–School Management System*. Bachelor Degree of Information Technology. Palestine: University of Palestine. Available from: <u>https://core.ac.uk/reader/84656470</u>.

- Alabbadi, H. & Al-Masaeed, S. 2020. The impact of ICT on coronavirus crisis management. *Journal of Theoretical and Applied Information Technology*, 98(18):3128-3129. Available from:http://www.jatit.org/volumes/Vol98No18/18Vol98No18.pdf.
- Alan, Ü. & Kabasakal, K.A. 2020. Effect of number of response options on the psychometric properties of Likert-type scales used with children. *Studies in Educational Evaluation*, 66:100895. Available from:<u>https://doi.org/DOI:10.1016/j.stueduc.2020.100895</u>.
- Ali, W. 2020. Online and remote learning in higher education institutes: A necessity in light of COVID-19 pandemic. *Higher education studies,* 10(3):16-25. Available from:<u>https://doi.org/DOI:10.5539/hes.v10n3p16</u>.
- Alkalbani, A., Deng, H., Kam, B. & Zhang, X. 2016. Investigating the impact of institutional pressures on information security compliance in organizations. Australasian Conference on Information Systems 2016 Proceedings. Wollongong, Australia. Available from:<u>https://aisel.aisnet.org/cgi/viewcontent.cgi?article=1025&context=acis2016</u>.
- Almeida Prado Cestari, J.M., Loures, E.D.F.R., Santos, E.a.P. & Panetto, H. 2020. A capability model for public administration interoperability. *Enterprise Information Systems*, 14(8):1071-1101. Available from:<u>https://doi.org/DOI:10.1080/17517575.2018.1564154</u>.
- Almgren, K. 2014. Exploring Enterprise Resource Planning Systems: A Comprehensive Analysis of ERP Systems. International Journal of Business and Social Science, 5(7). Available from:<u>https://ijbssnet.com/journals/Vol 5 No 7 June 2014/24.pdf</u>.
- Amua-Sekyi, E.T. & Asare, P.Y. 2016. A survey of Information Communication Technology literacy among lecturers. Information and Knowledge Management. Available from:<u>https://www.academia.edu/61727085/A\_Survey\_of\_Information\_Communication\_Techno</u> logy\_Literacy\_among\_Lecturers.
- Anggoro, B.K., Hubeis, M. & Sailah, I. 2018. Information system interoperability maturity model. *Bulletin* of Social Informatics Theory and Application, 2(1):22-33. Available from:<u>https://doi.org/DOI:10.31763/businta.v2i1.103</u>.
- Arshad, M. 2020. COVID-19: It's time to be thankful to our ICT professionals. *Information Technology & Electrical Engineering,* 9(2):23-31. Available from:<u>https://doi.org/DOI:10.22937/IJCSNS.2021.21.4.14</u>.
- Arslan, M.O. 2013. Free and open source software as a public good: Implications for education. *European Journal of Research on Education and Information Technologies*, (2):158-165. Available from:<a href="https://www.researchgate.net/profile/M-Oguz-Arslan/publication/264693270">https://www.researchgate.net/profile/M-Oguz-Arslan/publication/264693270</a> Free and open source software as a public good Implication and Information Technologies, (2):158-165. Available from:</a>
   <u>https://www.researchgate.net/profile/M-Oguz-Arslan/publication/264693270
   Free and open source\_software as a public good\_Implication and Information for\_education/links/53eb70f80cf23b8116a9c026/Free-and-open-source-software-as-apublic-good-Implication.pdf.
  </u>
- Asghari, M., Nassiri, P., Monazzam, M.R., Golbabaei, F., Arabalibeik, H., Shamsipour, A. & Allahverdy, A. 2017. Weighting Criteria and Prioritizing of Heat stress indices in surface mining using a Delphi Technique and Fuzzy AHP-TOPSIS Method. *Journal of Environmental Health Science and Engineering*, 15(1):1-11. Available from: https://doi.org/DOI:10.1186/s40201-016-0264-9.
- Avella, J.R. 2016. Delphi panels: Research design, procedures, advantages, and challenges. *International Journal of Doctoral Studies*, 11:305. Available from: <u>https://doi.org/DOI:10.28945/3561</u>.
- Awadallah, K. 2016. *School Management System Based on Web.* University of Palestine, College of Applied Engineering and Urban Planning .... Available from:<u>https://core.ac.uk/download/pdf/53035113.pdf</u>.
- Awasthy, R., Flint, S., Jones, R.L. & Sankaranarayana, R. 2018. UICMM: A Maturity Model for University-Industry Collaboration. 2018 IEEE International Conference on Engineering, Technology and Innovation (ICE/ITMC). Stuttgart, Germany. Available from:https://doi.org/DOI:10.1109/ICE.2018.8436266.
- Ayala, G. 2022. Change Is Incremental but Worth Fighting for: Breaking the Silence About HIV Intersectional Stigma. *American Journal of Public Health*, 112(S4):S333-S333. Available from: <a href="https://doi.org/DOI:10.2105/AJPH.2022.306746">https://doi.org/DOI:10.2105/AJPH.2022.306746</a>.
- Aytaç, T. 2020. The problems and opinions of school administrators during COVID-19 pandemic: A qualitative study from Turkey. *European Journal of Educational Sciences*, 7(4):57-79. Available from:<u>http://dx.doi.org/DOI:10.19044/ejes.v7no4a5</u>.
- Badr, E., Abdul Salam, M., Almotairi, S. & Ahmed, H. 2021. From linear programming approach to metaheuristic approach: scaling techniques. *Hindawi Complexity*, 2021:1-10. Available from:https://doi.org/DOI:10.1155/2021/9384318.
- Batool, S.H., Ur Rehman, A. & Sulehri, I. 2021. The current situation of information literacy education and curriculum design in Pakistan: a discovery using Delphi method. *Library Hi Tech*, 40(6):1705-1720. Available from:https://doi.org/DOI:10.1108/LHT-02-2021-0056.
- Battersby, D. 1982. The Delphi Technique: Its Methodology and Application in Higher Education. *Practical Assessment, Research, and Evaluation (PARE),* 12:1-285. Available from: <a href="https://doi.org/DOI:10.7275/pdz9-th90">https://doi.org/DOI:10.7275/pdz9-th90</a>.
- Beiderbeck, D., Frevel, N., Heiko, A., Schmidt, S.L. & Schweitzer, V.M. 2021. Preparing, conducting, and analyzing Delphi surveys: Cross-disciplinary practices, new directions, and advancements. *MethodsX*, 8:101401. Available from:<u>https://doi.org/DOI:10.1016/j.mex.2021.101401</u>.
- Beiderbeck, D., Frevel, N., Von Der Gracht, H.A., Schmidt, S.L. & Schweitzer, V.M. 2021. Preparing, conducting, and analyzing Delphi surveys: Cross-disciplinary practices, new directions, and advancements. *MethodsX*, 8:1-20. Available from:https://doi.org/DOI:10.1016/j.mex.2021.101401.
- Berryman, R., Yost, N., Dunn, N. & Edwards, C. 2013. Data interoperability and information security in healthcare. Transactions of the International Conference on Health Information Technology Advancement 2013 Michigan, United State. Available from:<u>https://scholarworks.wmich.edu/cgi/viewcontent.cgi?article=1020&context=ichita\_transa\_ ctions</u>.
- Bhattacharjee, B. & Deb, K. 2016. Role of ICT in 21st century's teacher education. International Journal ofEducationandInformationStudies,6(1):1-6.Availablefrom:https://www.ripublication.com/ijeis16/ijeisv6n101.pdf.

- Bhatti, M.S., Iqbal, A., Noreen, S., Mukhtar, R. & Javed, Z. 2021. Investigating the Role of ICT for teaching English at Elementary Level. *AsiaCALL Online Journal*, 12(2):87-93. Available from:https://asiacall.info/acoj/index.php/journal/article/view/35.
- Bindu, C. 2016. Impact of ICT on teaching and learning: A literature review. *International Journal of Management* and *Commerce Innovations*, 4(1):24-31. Available from:<u>https://www.academia.edu/37800061/Impact\_of\_ICT\_on\_Teaching\_and\_Learning\_A\_Lite</u> <u>rature\_Review</u>.
- Blau, I. & Presser, O. 2013. e-L eadership of school principals: Increasing school effectiveness by a school data management system. *British Journal of Educational Technology*, 44(6):1000-1011. Available from:https://doi.org/DOI:10.1111/bjet.12088.
- Bleijlevens, M.H., Wagner, L.M., Capezuti, E., Hamers, J.P. & Workgroup, I.P.R. 2016. Physical restraints: consensus of a research definition using a modified Delphi technique. *Journal of the American Geriatrics Society*, 64(11):2307-2310. Available from:<u>https://doi.org/DOI:10.1111/jgs.14435</u>.
- Boell, S.K. & Cecez-Kecmanovic, D. 2010. Literature reviews and the hermeneutic circle. *Australian Academic* & *Research Libraries*, 41(2):129-144. Available from:https://doi.org/DOI:10.1080/00048623.2010.10721450.
- Bone, M., Blackburn, M., Kruse, B., Dzielski, J., Hagedorn, T. & Grosse, I. 2018. Toward an interoperability and integration framework to enable digital thread. *Systems*, 6(4):46. Available from:<u>https://doi.org/DOI:10.3390/systems6040046</u>.
- Boone, H.N. & Boone, D.A. 2012. Analyzing likert data. *Journal of extension*, 50(2):1-5. Available from:<u>https://archives.joe.org/joe/2012april/pdf/JOE v50 2tt2.pdf</u>.
- Bourne Jr, L.E., Kole, J.A. & Healy, A.F. 2014. Expertise: defined, described, explained. 5:186. Available from: <a href="https://doi.org/DOI:10.3389/fpsyg.2014.00186">https://doi.org/DOI:10.3389/fpsyg.2014.00186</a>.
- Bowen, G.A. 2009. Document analysis as a qualitative research method. *Qualitative Research Journal*, 9(2):27-40. Available from: <a href="https://doi.org/DOI:10.3316/QRJ0902027">https://doi.org/DOI:10.3316/QRJ0902027</a>.
- Brailer, D.J. 2005. Interoperability: The Key To The Future Health Care System: Interoperability will bind together a wide network of real-time, life-critical data that not only transform but become health care. *Health Affairs*, 24(Suppl1):W5-19-W15-21. Available from:<u>https://doi.org/DOI:10.1377/HLTHAFF.W5.19</u>.
- Braun, V. & Clarke, V. 2006. Using thematic analysis in psychology. *Qualitative research in psychology*, 3(2):77-101. Available from: <u>http://dx.doi.org/DOI:10.1191/1478088706qp063oa</u>.
- Braun, V. & Clarke, V. 2019. Reflecting on reflexive thematic analysis. *Qualitative research in sport, exercise* and *health,* 11(4):589-597. Available from:https://doi.org/DOI:10.1080/2159676X.2019.1628806.
- Bucherer, E. & Uckelmann, D. 2011. Business models for the internet of things. Springer. [Online]. Available from: <u>https://link.springer.com/chapter/10.1007/978-3-642-19157-2\_10</u> [Accessed 16 October 2022].

- Calil, J., Reguero, B.G., Zamora, A.R., Losada, I.J. & Méndez, F.J. 2017. Comparative Coastal Risk Index (CCRI): a multidisciplinary risk index for Latin America and the Caribbean. *PLOS ONE*, 12(11):e0187011. Available from:https://doi.org/DOI:10.1371/journal.pone.0187011.
- Campos, C., Chalmeta, R., Grangel, R. & Poler, R. 2013. Maturity model for interoperability potential measurement. *Information Systems Management*, 30(3):218-234. Available from:<u>https://doi.org/DOI:10.1080/10580530.2013.794630</u>.
- Carvalho, S., Rossiter, J., Angrist, N., Hares, S. & Silverman, R. 2020. *Planning for school reopening and recovery after COVID-19.* [Online]. Available from: <u>https://www.cgdev.org/sites/default/files/planning-school-reopening-and-recovery-after-covid-19.pdf</u> [Accessed 20 May 2022].
- Castleberry, A. & Nolen, A. 2018. Thematic analysis of qualitative research data: Is it as easy as it sounds? *Currents in pharmacy teaching and learning,* 10(6):807-815. Available from:<u>https://doi.org/DOI:10.1016/j.cptl.2018.03.019</u>.
- Cestari, J.M.a.P., Loures, E.D.F.R. & Santos, E.a.P. 2018. A method to diagnose public administration interoperability capability levels based on multi-criteria decision-making. *International Journal of Information Technology & Decision Making*, 17(01):209-245. Available from:<u>https://doi.org/DOI:10.1142/S0219622017500365</u>.
- Chakrabarty, D. 2021. Arithmetic-harmonic mean: a measure of central tendency of ratio-type data. *International Journal of Advanced Research in Science, Engineering and Technology*, 8(5):17324-17333. Available from:<u>https://www.ijarset.com/volume-8-issue-5.html</u>.
- Chyung, S.Y., Roberts, K., Swanson, I. & Hankinson, A. 2017. Evidence-based survey design: The use of a midpoint on the Likert scale. *Performance Improvement*, 56(10):15-23. Available from:<u>https://doi.org/DOI:10.1002/pfi.21727</u>.
- Chyung, S.Y., Swanson, I., Roberts, K. & Hankinson, A. 2018. Evidence-based survey design: The use of continuous rating scales in surveys. *Performance Improvement*, 57(5):38-48. Available from: <a href="https://doi.org/DOI:10.1002/pfi.21763">https://doi.org/DOI:10.1002/pfi.21763</a>.
- Claveria, O. 2021. A new metric of consensus for Likert-type scale questionnaires: an application to consumer expectations. *Journal of Banking and Financial Technology*, 5(1):35-43. Available from:<u>https://doi.org/DOI:10.1007/s42786-021-00026-5</u>.
- Cooper, A. 2014. Learning analytics interoperability-the big picture in brief. *Learning Analytics Community Exchange*:1-11. Available from:<u>https://docplayer.net/7848232-Learning-analytics-interoperability-the-big-picture-in-brief.html</u>.
- Craig, L. 2017. A hermeneutic study of service improvement experiences in nursing: from student to newly registered nurse. Doctor of Philosophy degree. Northumbria United Kinddom: University of Northumbria at Newcastle (United Kingdom). Available from:https://nrl.northumbria.ac.uk/id/eprint/32554/1/craig.lynn\_phd.pdf.
- Creswell, J.W. 2007. *Qualitative Inquiry & Research Design: Choosing Among Five Approaches, 2nd edition.* SAGE. Available from:<u>https://revistapsicologia.org/public/formato/cuali2.pdf</u>.

- Creswell, J.W. & Creswell, J.D. 2018. *Research Design: Qualitative, Quantitative & Mixed Methods Approaches, 5ed.* California: SAGE Publications, Inc. Available from:https://spada.uns.ac.id/pluginfile.php/510378/mod\_resource/content/1/creswell.pdf.
- Dalkey, N. & Helmer, O. 1963. An experimental application of the Delphi method to the use of experts. *Management Science*, 9(3):458-467. Available from:<u>https://doi.org/DOI:10.1287/mnsc.9.3.458</u>.
- Daniel, H. 2017. Why the use of SA-SAMS as an administration system is not compulsory for schools. *SA-SAMS is not compulsory for schools*. Available from: <u>https://blog.schooladvisor.co.za/sa-sams-is-not-compulsory-for-schools/</u> 19 May 2023].
- Daniel, S.J. 2020. Education and the COVID-19 pandemic. *Prospects*, 49(1):91-96. Available from: <a href="https://doi.org/DOI:10.1007/s11125-020-09464-3">https://doi.org/DOI:10.1007/s11125-020-09464-3</a>.
- Das, K. 2019. The role and impact of ICT in improving the quality of education: An overview. *International Journal of Innovative Studies in Sociology and Humanities*, 4(6):97-103. Available from:<u>https://ijissh.org/storage/Volume4/Issue6/IJISSH-040611.pdf</u>.
- David, B. & Resnik, J. 2011. *What is ethics in research & why is it important.* [Online]. Available from: <u>https://www.niehs.nih.gov/research/resources/bioethics/whatis/</u> [Accessed 15 November 2021].
- David, B. & Roberta, H. 2020. What are Delphi studies? *Evidence Based Nursing*, 23(3):68. Available from:<u>http://dx.doi.org/DOI:10.1136/ebnurs-2020-103303</u>.
- Davis, N. 2016. *What Is the Fourth Industrial Revolution?* [Online]. Available from: <u>https://www.weforum.org/agenda/2016/01/what-is-the-fourth-industrial-revolution/</u> [Accessed 22 September 2021].
- DBE. 2021. The Data Driven Dashboard providing instant online access to education-related data. [Online]. Available from: <u>https://www.education.gov.za/ArchivedDocuments/ArchivedArticles/TheDataDrivenDashboard.</u> <u>aspx</u> [Accessed 19 May 2023].
- DCDT. 2020. Report of the Presidential Commission on the 4th Industrial Revolution. In: (ed.). Department Of Communications And Digital Technologies South Africa. Available from:<u>https://www.gov.za/documents/report-presidential-commission-4th-industrial-revolution-23-oct-2020-0000</u>.
- De Bruin, T., Rosemann, M., Freeze, R. & Kaulkarni, U. 2005. Understanding the main phases of developing a maturity assessment model. 16th Australasian Conference on Information Systems (ACIS) 2005 Proceeding. Sydney, Australia. Available from:<u>https://aisel.aisnet.org/acis2005/109/</u>.
- De Soria, A.B.M., Durán, C.N., Morrás, Á.S., González, J.P.D. & Varela, A. 2018. Questions and answers regarding character education in latin-american countries (Mexico, Colombia and Argentina): an exploratory delphi study. *Edetania: estudios y propuestas socio-educativas,* (53):23-44. Available from:<u>https://www.academia.edu/37053246</u>.
- Del Carmen Trillo-Luque, M., Vilches-Vilela, M.J., Quintero-Ordoñez, B., Fuentes-Gómez, F. & Dauder, M.S. 2020. Learning of English as a foreign language and gifted and talented Students: The role of ICT

in educational innovation. In: Gómez-Parra, M. & Huertas Abril, C. (eds.). *Handbook of Research on Bilingual and Intercultural Education*. Spain: IGI Global. Available from: https://doi.org/DOI:10.4018/978-1-7998-2588-3.ch011.

- Desantis, L. & Ugarriza, D.N. 2000. The concept of theme as used in qualitative nursing research. *Western journal* of *nursing research*, 22(3):351-372. Available from:https://doi.org/DOI:10.1177/019394590002200308.
- Deveci, M., Özcan, E., John, R., Covrig, C.-F. & Pamucar, D. 2020. A study on offshore wind farm siting criteria using a novel interval-valued fuzzy-rough based Delphi method. *Journal of Environmental Management*, 270:1-16. Available from:<u>https://doi.org/DOI:10.1016/j.jenvman.2020.110916</u>.
- Di Zio, S. 2018. Convergence of experts' opinions on the territory: The Spatial Delphi and the Spatial Shang. *Innovative research methodologies in management.* Springer. Available from:https://link.springer.com/chapter/10.1007/978-3-319-64400-4 1.
- Di Zio, S. & Pacinelli, A. 2011. Opinion convergence in location: A spatial version of the Delphi method. *Technological Forecasting and Social Change*, 78(9):1565-1578. Available from:https://doi.org/DOI:10.1016/j.techfore.2010.09.010.
- Dönmez, S. & Toker, Y. 2017. Construction of a Likert-type transformational leadership scale. *Ankara Üniversitesi Dil ve Tarih-Coğrafya Fakültesi Dergisi,* 57(2):753-775. Available from:https://dergipark.org.tr/en/pub/dtcfdergisi/issue/66796/1044503.
- Dube, B. 2020. Rural online learning in the context of COVID 19 in South Africa: Evoking an inclusive education approach. *REMIE: Multidisciplinary Journal of Educational Research,* 10(2):135-157. Available from:<u>https://doi.org/DOI:10.17583/remie.2020.5607</u>.
- Durizzo, K., Asiedu, E., Van Der Merwe, A., Van Niekerk, A. & Günther, I. 2021. Managing the COVID-19 pandemic in poor urban neighborhoods: The case of Accra and Johannesburg. *World Development*, 137:105175. Available from:<u>https://doi.org/DOI:10.1016/j.worlddev.2020.105175</u>.
- Ed-Admin. 2022. *Ed-admin Learning*. [Online]. Available from: <u>https://www.ed-admin.com/products/ed-admin-learning</u> [Accessed 07 November 2022].
- Education, D.O.B. 2021. Admission of Learners to Public Schools. [Online]. Available from: <u>https://www.education.gov.za/Informationfor/ParentsandGuardians/SchoolAdmissions.aspx</u> [Accessed 19 May 2023].
- Ekström, T. 2020. Segmentation and differentiation in defence supply chain design: A dynamic purchasing portfolio model for defence procurement. Jönköping University, School of Engineering. Available from:<u>https://www.diva-portal.org/smash/get/diva2:1501195/FULLTEXT01.pdf</u>.
- El-Khoueiry, A.B., Sangro, B., Yau, T., Crocenzi, T.S., Kudo, M., Hsu, C., Kim, T.-Y., Choo, S.-P., Trojan, J. & Welling 3rd, T.H. 2017. Nivolumab in patients with advanced hepatocellular carcinoma (CheckMate 040): an open-label, non-comparative, phase 1/2 dose escalation and expansion trial. *The Lancet*, 389(10088):2492-2502. Available from:https://www.sciencedirect.com/science/article/abs/pii/S0140673617310462.

#### KENEILWE MAREMI: 44977883

- Elizabeth, M. 2021. *Privacy is a human right': POPIA and your school's obligations to protect your information.* [Online]. Available from: <u>https://www.news24.com/life/archive/privacy-is-a-human-right-popia-and-your-schools-obligations-to-protect-your-information-20210701</u> [Accessed 19 May 2023].
- Engelman, D., Fuller, L. & Steer, A. 2018. for the International Alliance for the Control of Scabies Delphi panel. *Consensus criteria for the diagnosis of scabies: A Delphi study of international experts. PLoS Negl Trop Dis,* 12(5):e0006549. Available from:https://doi.org/DOI:10.1371/journal.pntd.0006549.
- Esmailzadeh, M., Noori, S., Aliahmadi, A., Nouralizadeh, H. & Bogers, M. 2020. A functional analysis of technological innovation systems in developing countries: An evaluation of Iran's photovoltaic innovation system. *Sustainability*, 12(5):2049. Available from:https://doi.org/DOI:10.3390/su12052049.
- Espadinha-Cruz, P., Grilo, A., Gonçalves-Coelho, A. & Mourão, A. 2019. An axiomatic design framework to design interoperable buyer–supplier dyads. *Enterprise Information Systems*, 13(10):1392-1426. Available from:<u>https://doi.org/DOI:10.1080/17517575.2019.1650959</u>.
- Etikan, I., Musa, S.A. & Alkassim, R.S. 2016. Comparison of convenience sampling and purposive sampling. *American Journal of Theoretical and Applied Statistics*, 5(1):1-4. Available from:https://doi.org/DOI:10.11648/j.ajtas.20160501.11.
- Fernandes, J.C., Neto, V.V.G. & Santos, R.P.D. 2018. Interoperability in systems-of-information systems: A systematic mapping study. Proceedings of the 17th Brazilian Symposium on Software Quality. Available from: <a href="https://doi.org/DOI:10.1145/3275245.3275259">https://doi.org/DOI:10.1145/3275245.3275259</a>.
- Fink-Hafner, D., Dagen, T., Doušak, M., Novak, M. & Hafner-Fink, M. 2019. Delphi method: strengths and weaknesses. Advances in Methodology and Statistics, 16(2):1–19. Available from:https://doi.org/DOI:10.51936/fcfm6982.
- Fisher, H., Erasmus, A.C. & Viljoen, A.T. 2020. Adaptation of the Delphi technique for electronic application in the food industry. Available from: <u>https://repository.up.ac.za/handle/2263/77192</u>.
- Flostrand, A. 2017. Finding the future: Crowdsourcing versus the Delphi technique. *Business Horizons,* 60(2):229-236. Available from: <a href="https://doi.org/DOI:10.1016/j.bushor.2016.11.007">https://doi.org/DOI:10.1016/j.bushor.2016.11.007</a>.
- Fong, S.F., Ch'ng, P.E. & Por, F.P. 2013. Development of ICT competency standard using the Delphi technique. *Procedia-Social and Behavioral Sciences*, 103:299-314. Available from:https://doi.org/DOI:10.1016/j.sbspro.2013.10.338.
- Fraillon, J., Ainley, J., Schulz, W., Friedman, T. & Duckworth, D. 2020. *Preparing for life in a digital world: IEA international computer and information literacy study 2018 international report.* Springer Nature. Available from:<u>https://doi.org/DOI:10.1007/978-3-030-38781-5</u>.
- Franklin, K.K. & Hart, J.K. 2007. Idea generation and exploration: Benefits and limitations of the policy Delphi research method. *Innovative higher education*, 31(4):237-246. Available from:<u>https://doi.org/DOI:10.1007/s10755-006-9022-8</u>.

- Friston, K.J. & Frith, C.D. 2015. Active inference, communication and hermeneutics. *cortex*, 68:129-143. Available from:<u>https://doi.org/DOI:10.1016/j.cortex.2015.03.025</u>.
- Garland, R. 1991. The mid-point on a rating scale: Is it desirable. *Marketing bulletin*, 2(1):66-70. Available from:<u>https://www.rangevoting.org/MB\_V2\_N3\_Garland.pdf</u>.
- Gasser, U. & Palfrey, J.G. 2007. Breaking down digital barriers: When and how ICT interoperability drives innovation. *Berkman Center Research Publication,* (2007-8). Available from:<u>http://dx.doi.org/DOI:10.2139/ssrn.1033226</u>.
- Gentles, S.J., Charles, C., Ploeg, J. & Mckibbon, K.A. 2015. Sampling in qualitative research: Insights from an overview of the methods literature. *The qualitative report*, 20(11):1772-1789. Available from:<u>https://pdfs.semanticscholar.org/3bd0/66b5d7ea4e3a933699576689a855d09f08b6.pdf</u>.
- Ghavifekr, S., Afshari, M., Siraj, S. & Seger, K. 2013. ICT application for administration and management: A conceptual review. *Procedia-Social and Behavioral Sciences*, 103:1344-1351. Available from:<u>https://doi.org/DOI:10.1016/j.sbspro.2013.10.705</u>.
- Global, D. 2018. *Global Business Coalition for Education*. [Online]. Available from: <u>https://www.deloitte.com/global/en/issues/work/gx-preparing-tomorrow-workforce-for-the-fourth-industrial-revolution.html</u> [Accessed 12 June 2021].
- Göpper, J. 2020. Shattering the glass ceiling: women progressing into leadership positions at secondary schools in South Africa. Masters. University of Pretoria. Available from: <a href="https://repository.up.ac.za/handle/2263/78455">https://repository.up.ac.za/handle/2263/78455</a>.
- Green, R.A. 2014. The Delphi technique in educational research. *Sage Open,* 4(2):2158244014529773. Available from: <a href="https://doi.org/DOI:10.1177/2158244014529773">https://doi.org/DOI:10.1177/2158244014529773</a>.
- Griffey, R.T., Schneider, R.M., Adler, L.M., Capp, R., Carpenter, C.R., Farmer, B.M., Groner, K.Y., Hodkins, S., Mccammon, C.A. & Powell, J.T. 2020. Development of an emergency department trigger tool using a systematic search and modified delphi process. *Journal of patient safety*, 16(1):e11-e17. Available from:<u>http://doi.org/DOI:10.1097/PTS.00000000000243</u>.
- Grime, M.M. & Wright, G. 2016. Delphi method. *Wiley StatsRef: Statistics Reference Online*:1-6. Available from: <a href="http://doi.org/DOI:10.1002/9781118445112.stat07879">http://doi.org/DOI:10.1002/9781118445112.stat07879</a>.
- Guba, E.G. & Lincoln, Y.S. 1994. Competing paradigms in qualitative research. *Handbook of qualitative research,* 2(163-194):105. Available from:<u>https://miguelangelmartinez.net/IMG/pdf/1994\_Guba\_Lincoln\_Paradigms\_Quali\_Research\_h\_chapter.pdf</u>.
- Guédria, W., Naudet, Y. & Chen, D. 2008. Interoperability maturity models–survey and comparison–. OTM Confederated International Conferences" On the Move to Meaningful Internet Systems". Available from:http://doi.org/DOI:10.1007/978-3-540-88875-8\_48.
- Gunawan, J. 2015. Ensuring trustworthiness in qualitative research. *Belitung Nursing Journal*, 1(1):10-11. Available from: <u>https://doi.org/DOI:10.33546/bnj.4</u>.

- Gürdür, D. & Asplund, F. 2018. A systematic review to merge discourses: Interoperability, integration and cyber-physical systems. *Journal of Industrial Information Integration*, 9:14-23. Available from:https://doi.org/DOI:10.1016/j.jii.2017.12.001.
- Gwenzi, W. & Rzymski, P. 2021. When silence goes viral, Africa sneezes! A perspective on Africa's subdued research response to COVID-19 and a call for local scientific evidence. *Environmental research*, 194:110637. Available from:<u>https://doi.org/DOI:10.1016/j.envres.2020.110637</u>.
- Gxwati, N.I. 2011. The education management information system of the Free State Department of Education: a systems analysis. Masters. Stellenbosch: University of Stellenbosch. Available from:https://scholar.sun.ac.za/handle/10019.1/6487.
- Habibi, A., Sarafrazi, A. & Izadyar, S. 2014. Delphi technique theoretical framework in qualitative research. *The International Journal of Engineering and Science*, 3(4):8-13. Available from:<u>https://parsmodir.com/wp-content/uploads/2018/11/Delphi2014-En.pdf</u>.
- Haghighinasab, M. & Khosravi, S. 2011. Evaluating the Organizational Interoperability Maturity Level in ICT Research Center. *Journal of Information Technology Management*, 3(6). Available from.
- Harper, S. 2018. Public Health Providers' Perceptions of Electronic Health Records in a Disaster. PhD. Walden University. Available from:<u>https://www.proquest.com/openview/e04ca58442fa1725b9bcf33f98b45119/1?cbl=18750</u> <u>&pq-origsite=gscholar</u>.
- Hasson, F. & Keeney, S. 2011. Enhancing rigour in the Delphi technique research. *Technological Forecasting* and *Social Change*, 78(9):1695-1704. Available from:https://doi.org/DOI:10.1016/j.techfore.2011.04.005.
- Hasson, F., Keeney, S. & Mckenna, H. 2000. Research guidelines for the Delphi survey technique. *Journal* of advanced nursing, 32(4):1008-1015. Available from: <u>https://doi.org/DOI:10.1046/j.1365-2648.2000.t01-1-01567.x</u>.
- Hatakka, M., Thapa, D. & Sæbø, Ø. 2020. Understanding the role of ICT and study circles in enabling economic opportunities: Lessons learned from an educational project in Kenya. *Information Systems Journal*, 30(4):664-698. Available from: <u>https://doi.org/DOI:10.1111/isj.12277</u>.
- Hendrawan, M.R. 2018. THE IMPLEMENTATION OF TECHNICAL INTEROPERABILITY MODELS IN LARAS VERSION 1.0 DIGITAL LIBRARY APPLICATIONS: A CASE STUDY IN THE CENTER FOR SCIENTIFIC DOCUMENTATION AND INFORMATION–INDONESIAN. *Journal of Documentation and Information Science*, 2(2):53-63. Available from: http://dx.doi.org/DOI:10.33505/jodis.v2i2.133.
- Hernandez, R.M. 2017. Impact of ICT on Education: Challenges and Perspectives. Journal of EducationalPsychology-PropositosyRepresentaciones,5(1):337-347.from:http://dx.doi.org/DOI:10.20511/pyr2017.v5n1.149.

Herselman, M. 2011. Research methodology. Available from.

Hill, K.Q. & Fowles, J. 1975. The methodological worth of the Delphi forecasting technique. *Technological Forecasting and Social Change*, 7(2):179-192. Available from: <u>https://doi.org/DOI:10.1016/0040-1625(75)90057-8</u>.

- Hill, S., Lettington, M.C. & Schmidt, K.M. 2018. Block representations and spectral properties of constant sum matrices. *Electronic Journal of Linear Algebra*, 34:170-190. Available from:<u>https://doi.org/DOI:10.13001/1081-3810.3530</u>.
- Hiller, J. 2016. Epistemological foundations of objectivist and interpretivist research. Available from: <a href="https://ecommons.udayton.edu/books/53/">https://ecommons.udayton.edu/books/53/</a>.
- Hinds, M. 2016. *The US, the UK and Saudi Arabia in World War II: the Middle East and the origins of a special relationship.* Bloomsbury Publishing. Available from:<u>https://books.google.co.za/books?hl=en&lr=&id=\_-</u> <u>eKDwAAQBAJ&oi=fnd&pg=PP1&dq=Hinds,+M.+2016.+The+US,+the+UK+and+Saudi+Arabia+in+</u> <u>World+War+II:+the+Middle+East+and+the+origins+of+a+special+relationship.+Bloomsbury+Pub</u> <u>lishing.+Available+from.&ots=1e32rCy\_nR&sig=Aej8p5kERAuPsaMp5C2GKzMLSYY&redir\_esc=y</u> <u>#v=onepage&q=Hinds%2C%20M.%202016.%20The%20US%2C%20the%20UK%20and%20Saudi</u> <u>%20Arabia%20in%20World%20War%20II%3A%20the%20Middle%20East%20and%20the%20Ori</u> <u>gins%20of%20a%20special%20relationship.%20Bloomsbury%20Publishing.%20Available%20fro</u> <u>m.&f=false</u>.
- Hirschhorn, F. 2019. Reflections on the application of the Delphi method: lessons from a case in public transport research. *International Journal of Social Research Methodology*, 22(3):309-322. Available from:<u>https://doi.org/DOI:10.1080/13645579.2018.1543841</u>.
- Hlongoane, M.T. 2019. *Financial accountability and reporting by governing bodies of public schools in South Africa.* PhD. Available from: <u>https://core.ac.uk/download/pdf/237210358.pdf</u>.
- Hohmann, E., Cote, M.P. & Brand, J.C. 2018. Research pearls: expert consensus based evidence using the Delphi method. *Arthroscopy: The Journal of Arthroscopic & Related Surgery*, 34(12):3278-3282. Available from:https://doi.org/DOI:10.1016/j.arthro.2018.10.004.
- Holmberg, C., Wolf, A., Eide, C., Großschädl, F., Schüttengruber, G., Patel, H. & Heckemann, B. 2021.
   Classic tool, new opportunities: A psychometric analysis of a Swedish online version of the Aging Semantic Differential Scale. *Research on Aging*, 43(9-10):378-387. Available from:https://doi.org/DOI:10.1177/0164027520963618.
- Hsu, C.-C. & Sandford, B.A. 2007. The Delphi technique: making sense of consensus. *Practical assessment, research, and evaluation,* 12(1):10. Available from: <a href="https://doi.org/DOI:10.7275/pdz9-th90">https://doi.org/DOI:10.7275/pdz9-th90</a>.
- Hussain, M.A., Elyas, T. & Nasseef, O.A. 2013. Research paradigms: a slippery slop for fresh researchers. Journal of Chemical Information and Modeling, 10(4):2374-2381. Available from:http://doi.org/10.1017/CB097811074153.004.
- Ibiyemi, A.O., Adnan, Y.M. & Daud, M.N. 2016. The validity of the classical Delphi applications for assessing the industrial sustainability-correction factor: an example study. *foresight*. Available from:<u>https://doi.org/DOI:10.1108/FS-04-2016-0016</u>.
- Ibrahim, M. 2020. The Fourth Industrial Revolution Combatting COVID-19: The Role of Smart and Sustainable Cities. United Nations Department of Economic and Social Affairs: New York, NY, USA:6. Available from:<u>https://www.un.org/development/desa/dspd/wpcontent/uploads/sites/22/2020/09/Maysoun-Ibrahim\_4IR-and-SCs-in-the-Time-of-COVID19.pdf</u>.

- Imam, N.A. & Itodo, A. 2020. ROLE OF ICT IN EDUCATION AND ECONOMY IN NIGERIA. Nation. Available from:<u>https://www.cambridgenigeriapub.com/wp-</u> content/uploads/2021/01/CJCER Vol18 No8 Sept 2020-20.pdf.
- Irene, M.K. 2013. MOBILE TELEPHONY AND THE MANAGEMENT OF EDUCATION INSTITUTIONS IN KENYA: A CASE STUDY OF BRIDGE INTERNATIONAL ACADEMIES. UNIVERSITY OF NAIROBI. Available from:<u>http://erepository.uonbi.ac.ke/bitstream/handle/11295/60097/Mukhwana\_Mobile%20tel</u> <u>ephony%20and%20the%20management%20of%20education%20institutions%20in%20Kenya.pd</u> <u>f?sequence=3</u>.
- Issah, M. 2018. Change leadership: The role of emotional intelligence. *Sage Open*, 8(3):2158244018800910. Available from.
- Jabin, J., Dimyadi, J. & Amor, R. 2019a. From a meagre 'testing for interoperability' to the 'measurement of interoperability' in BIM. Proceedings of the 36th CIB W78 Conference "ICT in Design, Construction and Management in Architecture, Engineering, Construction and Operations (AECO)", September. Available from:<u>https://itc.scix.net/pdfs/w78-2019-paper-030.pdf</u>.
- Jabin, J., Dimyadi, J. & Amor, R. 2019b. Systematic literature review on interoperability measurement models. Available from: <u>https://doi.org/DOI:10.13140/RG.2.2.33957.35047</u>.
- Jahoor, F. 2019. A framework for mobile digital literacy skills of educators using mobile technology in rural formal education. Masters. Available from:<u>https://core.ac.uk/download/pdf/231923025.pdf</u>.
- Jakimoski, K. 2016. Challenges of interoperability and integration in education information systems. *International Journal of Database and Theory and Application*, 9(2):33-46. Available from:<u>http://dx.doi.org/DOI:10.14257/ijdta.2016.9.2.05</u>.
- Jaleel, A., Mahmood, T., Hassan, M.A., Bano, G. & Khurshid, S.K. 2020. Towards medical data interoperability through collaboration of healthcare devices. *IEEE Access*, 8:132302-132319. Available from:<u>https://ieeexplore.ieee.org/stamp/stamp.jsp?arnumber=9142173</u>.
- Jansen, H. & Hak, T. 2005. The productivity of the three-step test-interview (TSTI) compared to an expert review of a self-administered questionnaire on alcohol consumption. *Journal of Official Statistics: an international quarterly*, 21(1):103-120. Available from:<u>http://hdl.handle.net/1765/20960</u>.
- Jardine, D.W. 2015. On hermeneutics: "Over and above our wanting and doing". *Doing educational research.* Brill. Available from: <u>https://doi.org/DOI:10.1163/9789463000765\_012</u>.
- Javaheri, H., Mirzaei, M. & Lukowicz, P. 2020. How far can Wearable Augmented Reality Influence Customer Shopping Behavior. MobiQuitous 2020-17th EAI International Conference on Mobile and Ubiquitous Systems: Computing, Networking and Services. Available from:https://doi.org/DOI:10.1145/3448891.3448930.
- Johnson-Laird, P. 2010. Deductive reasoning. *Wiley Interdisciplinary Reviews: Cognitive Science*, 1(1):8-17. Available from:<u>https://doi.org/DOI:10.1002/wcs.20</u>.
- Jolson, M.A. & Rossow, G.L. 1971. The Delphi process in marketing decision making. *Journal of Marketing Research*, 8(4):443-448. Available from: <u>https://doi.org/DOI:10.1177/00222437710080040</u>.

- Kabir, S.M.S. 2016. Basic guidelines for research. *An introductory approach for all disciplines*, 4(2):168-180. Available from:<u>https://www.scirp.org/(S(czeh2tfqw2orz553k1w0r45))/reference/referencespapers.aspx?r</u> <u>eferenceid=2705906</u>.
- Kallia, M. & Sentance, S. 2017. Computing teachers' perspectives on threshold concepts: Functions and procedural abstraction. Proceedings of the 12th workshop on primary and secondary computing education. Available from: <a href="https://doi.org/DOI:10.1145/3137065.3137085">https://doi.org/DOI:10.1145/3137065.3137085</a>.
- Kattirtzi, M. & Winskel, M. 2020. When experts disagree: Using the Policy Delphi method to analyse divergent expectations and preferences on UK energy futures. *Technological Forecasting and Social Change*, 153:1-10. Available from:https://doi.org/DOI:10.1016/j.techfore.2020.119924.
- Kayembe, C. & Nel, D. 2019. Challenges and opportunities for education in the Fourth Industrial Revolution. *African Journal of Public Affairs*, 11(3):79-94. Available from:<u>https://journals.co.za/doi/pdf/10.10520/EJC-19605d342e</u>.
- Kearney, A.T. 2019. Beyond The Numbers. In: (ed.). National Education Collaboration Trust South Africa. Available from:<u>https://nect.org.za/publications/technical-reports/msdf-beyond-the-numbers\_a4\_20aug19\_final.pdf</u>.
- Kerber, W. & Schweitzer, H. 2017. Interoperability in the digital economy. J. Intell. Prop. Info. Tech. & Elec.Com.L.,8:39.Availablefrom:<a href="https://www.econstor.eu/bitstream/10419/162658/1/880542403.pdf">https://www.econstor.eu/bitstream/10419/162658/1/880542403.pdf</a>.
- Khan, S. 2020. Social psychology and pandemics: Exploring consenus about research priorities and strategies using the Delphi method. *Asian Journal of Social Psychology*, 23:363-371. Available from:<u>https://doi.org/DOI:10.1111/ajsp.12442</u>.
- Kiger, M.E. & Varpio, L. 2020. Thematic analysis of qualitative data: AMEE Guide No. 131. *Medical teacher,* 42(8):846-854. Available from: <a href="https://doi.org/DOI:10.1080/0142159X.2020.1755030">https://doi.org/DOI:10.1080/0142159X.2020.1755030</a>.
- Kirpes, B., Danner, P., Basmadjian, R., Meer, H.D. & Becker, C. 2019. E-mobility systems architecture: a model-based framework for managing complexity and interoperability. *Energy Informatics*, 2(1):1-31. Available from:<u>https://doi.org/DOI:10.1186/s42162-019-0072-4</u>.
- Klein, H.K. & Myers, M.D. 1999. A set of principles for conducting and evaluating interpretive field studies in information systems. *Mis Quarterly*:67-93. Available from:<u>https://doi.org/DOI:10.2307/249410</u>.
- Kobus, J. & Westner, M. 2016. Ranking-type delphi studies in IS research: step-by-step guide and analytical extension. IADIS Int. Conf. Inf. Syst. Available from:<u>https://www.academia.edu/24031436/RANKING\_TYPE\_DELPHI\_STUDIES\_IN\_IS\_RESEARC\_H\_STEP\_BY\_STEP\_GUIDE\_AND\_ANALYTICAL\_EXTENSION</u>.
- König, J., Jäger-Biela, D.J. & Glutsch, N. 2020. Adapting to online teaching during COVID-19 school closure: teacher education and teacher competence effects among early career teachers in Germany. *European Journal of Teacher Education*, 43(4):608-622. Available from:<u>https://doi.org/DOI:10.1080/02619768.2020.1809650</u>.

- Kuhlmann, T., Dantlgraber, M. & Reips, U.-D. 2017. Investigating measurement equivalence of visual analogue scales and Likert-type scales in Internet-based personality questionnaires. *Behavior research methods*, 49:2173-2181. Available from:<u>https://link.springer.com/article/10.3758/s13428-016-0850-x</u>.
- Kulas, J.T. & Stachowski, A.A. 2013. Respondent rationale for neither agreeing nor disagreeing: Person and item contributors to middle category endorsement intent on Likert personality indicators. *Journal of Research in Personality*, 47(4):254-262. Available from:<u>https://doi.org/DOI:10.1016/j.jrp.2013.01.014</u>.
- Kunda, D. & Chama, V. 2016. Design and Implementation of Students' Information System (SIS) for Mulungushi University Based on Spring Framework. *development*, 5(6). Available from:<u>https://docplayer.net/44877602-Design-and-implementation-of-students-informationsystem-sis-for-mulungushi-university-based-on-spring-framework.html</u>.
- Kuriakose, E.B. 2014. A cost effective school management system for disadvantaged schools in the Free State province using the software as a service (SaaS) delivery model. [Bloemfontein?]: Central University of Technology, Free State. Available from.
- Landeta, J. & Barrutia, J. 2011. People consultation to construct the future: a Delphi application. *International Journal of Forecasting*, 27(1):134-151. Available from:<u>https://doi.org/DOI:10.1016/j.ijforecast.2010.04.001</u>.
- Lane, J.A. & Valerdi, R. 2011. System interoperability influence on system of systems engineering effort. Proceedings of the Conference on Systems Engineering Research. Available from:<u>https://www.academia.edu/2851366/System\_Interoperability\_Influence\_on\_System\_of\_S</u> ystems\_Engineering\_Effort.
- Lang, T. 1995. An overview of four futures methodologies. *Manoa Journal of Fried and Half-Fried Ideas*, 7:1-43. Available from:<u>https://citeseerx.ist.psu.edu/document?repid=rep1&type=pdf&doi=870cbdc2ef88883d2c bb55821ededb4c40527101</u>.
- Leal, G.D.S.S., Guédria, W. & Panetto, H. 2019. Interoperability assessment: A systematic literature review. *Computers in Industry*:106, 111-132. Available from:<u>https://hal.archives-ouvertes.fr/hal-01982217/file/Leal%20et%20al.pdf</u>.
- Leedy, P.D. & Ormrod, J.E. 2015. Practical research: Planning and design, global edition. *England: Pearson Education Limited*. Available from.
- Leedy, P.D. & Ormrod, J.E. 2021. Practical research: Planning and design, global edition. *England: Pearson Education* from:<u>http://students.aiu.edu/submissions/profiles/resources/onlineBook/a2g3B6\_Practical%20</u> <u>Research\_%20Planning%20And%20Design-2020.pdf</u>.
- Lewis, P. & Saunders, M. 2018. Doing research in business and management. *Essex, England: Pearson Education Limited*. Available from:<u>https://www.worldcat.org/title/doing-research-in-business-and-management-an-essential-guide-to-planning-your-project/oclc/1004569117</u>.

- Li, G., Hou, Y. & Wu, A. 2017. Fourth Industrial Revolution: technological drivers, impacts and coping methods. *Chinese Geographical Science*, 27:626-637. Available from:https://doi.org/DOI:10.1007/s11769-017-0890-x
- Likert, R. 1932. A technique for the measurement of attitudes. *Archives of psychology*. Available from: https://doi.org/DOI:10.1037/h0057138.
- Lilja, K.K., Laakso, K. & Palomäki, J. 2011. Using the Delphi method. 2011 Proceedings of PICMET'11: Technology Management in the Energy Smart World (PICMET). Available from:<u>https://ieeexplore.ieee.org/stamp/stamp.jsp?arnumber=6017716&casa\_token=PS2u4O2R</u> <u>xHkAAAAA:0846dA2u0ec1u52fHk1Lb6CBCbzgD6LDr6O8\_wosSJGclCQnsvzZDqLWz\_4TFKALf0KsY</u> <u>xlsltj38Q</u>.
- Lin, R., Xie, J., Jeng, Y.-C. & Wang, Z.-H. 2011. The relationship between job involvement and school administrative effectiveness as perceived by administration teachers. *Management in Education*, 25(3):112-118. Available from:<u>https://doi.org/DOI:10.1177/0892020610392434</u>.
- Linstone, H. & Turoff, M. 1975. The delphi method: Addison-Wesley Reading, MA. Available from: <a href="https://doi.org/DOI:10.2307/3150755">https://doi.org/DOI:10.2307/3150755</a>.
- Linstone, H.A. & Turoff, M. 2011. Delphi: A brief look backward and forward. *Technological Forecasting and Social Change*, 78(9):1712-1719. Available from:https://doi.org/10.1016/j.techfore.2010.09.011.
- Luo, A., Zhang, M., Mao, Y., Kou, Y. & Zhang, X. 2019. A Structural Complexity Metric Method for Complex Information Systems. J. Softw., 14(7):332-339. Available from:https://doi.org/DOI:10.17706/jsw.14.7.332-339.
- Malan, A.C. 2014. *Training needs of primary school principals in the use of ICTs in school management of township* and rural schools. Masters. Available from:https://repository.nwu.ac.za/handle/10394/15386.
- Manakandan, S.K., Rosnah, I., Mohd, R. & Priya, R. 2017. Pesticide applicators questionnaire content validation: A fuzzy delphi method. *Med J Malaysia*, 72(4):228-235. Available from:https://pubmed.ncbi.nlm.nih.gov/28889134/.
- Manisera, M. & Zuccolotto, P. 2022. A mixture model for ordinal variables measured on semantic differential scales. *Econometrics and Statistics*, 22:98-123. Available from:https://doi.org/DOI:10.1016/j.ecosta.2021.07.002.
- Manoliadis, O. 2018. The assessment of construction project risks with the use of fuzzy Delphi methodology—case study photovoltaic project in Greece. *International Journal of Management and Fuzzy Systems,* 4(1):7-14. Available from:<u>https://www.sciencepublishinggroup.com/journal/paperinfo?journalid=353&doi=10.1164 8/j.ijmfs.20180401.12</u>.
- Maponya, S.H. 2015. The role of the principal as instructional leader in improving learner achievement in South African primary schools. PhD. University of South Africa. Available from:<u>https://core.ac.uk/download/pdf/43177707.pdf</u>.

- Maremi, K.J., Herselman, M.E. & Botha, A. 2020. Interoperability maturity models for consideration when using School Management Systems (SMS) in South Africa: A scoping review. Available from:<u>https://publications.waset.org/10011790/interoperability-maturity-models-forconsideration-when-using-school-management-systems-in-south-africa-a-scoping-review</u>.
- Marinov, O. & Tsankova, R. 2015. Impact of ICT on administrative management processes. VII МЕЖДУНАРОДНА НАУЧНА КОНФЕРЕНЦИЯ "Е-УПРАВЛЕНИЕ" в рамките на "Дни на науката— 2015 "на ТУ-София VII-th INTERNATIONAL SCIENTIFIC CONFERENCE. Available from.
- Marques, J.B.V. & De Freitas, D. 2018. The delphi method: characterization and potentialities for educational research. *Pro-Posições,* 29(2):389. Available from:http://dx.doi.org/DOI:10.1590/1980-6248-2015-0140.
- Massaroli, A., Martini, J.G., Lino, M.M., Spenassato, D. & Massaroli, R. 2018. THE DELPHI METHOD AS A METHODOLOGICAL FRAMEWORK FOR RESEARCH IN NURSING1. *Texto & Contexto-Enfermagem*, 26. from:https://www.scielo.br/j/tce/a/7Q7Xg6BBXBtXgmvxyYtjNTG/?format=pdf&lang=en.
- Matas, A. 2018. Likert-type scale format design: state of art. *Revista electrónica de investigación educativa*, 20(1):38-47. Available from:<u>https://doi.org/DOI:10.54517/met.v1i2.1777</u>.
- Mathevula, M.D. & Uwizeyimana, D.E. 2014. The challenges facing the integration of ICT in teaching and learning activities in South African rural secondary schools. *Mediterranean Journal of Social Sciences*, 5(20):1087. Available from:<u>https://doi.org/DOI:10.5901/mjss.2014.v5n20p1087</u>.
- Matzel, L.D. & Sauce, B. 2017. Department of Psychology, Rutgers University, Piscataway, NJ, USA. Available from:<u>https://brunosauce.net/wp-content/uploads/2018/08/Matzel-Sauce-2017-</u> <u>lq.pdf</u>.
- Mayan, M.J. 2016. *Essentials of qualitative inquiry.* Routledge. Available from: https://doi.org/DOI:10.4324/9781315429250.
- Mbuqe, B. 2020. *Procurement of learner teacher support materials in East London District public secondary schools*. Masters. Available from:<u>https://core.ac.uk/download/pdf/334609817.pdf</u>.
- Mcdaniel, N. 2020. Military Interoperability: Definitions, Models, Actors, and Guidelines. Available from.
- Mcleod, S., Babb, C. & Barlow, S. 2020. How to 'do'a bike plan: Collating best practices to synthesise a Maturity Model of planning for cycling. *Transportation research interdisciplinary perspectives*, 5:100130. Available from: https://doi.org/DOI:10.1016/j.trip.2020.100130.
- Melnikovas, A. 2018. Towards an explicit research methodology: Adapting research onion model for futures studies. *Journal of Futures Studies*, 23(2):29-44. Available from:<u>https://doi.org/DOI:10.6531/JFS.201812\_23(2).0003</u>.
- Mhlanga, D. & Moloi, T. 2020. COVID-19 and the digital transformation of education: What are we learning on 4IR in South Africa? *Education sciences*, 10(7):180. Available from:<u>https://doi.org/DOI:10.3390/educsci10070180</u>.

- Mingqian, W. & Shuai, C. 2020. A survey on C4ISR system architecture technique. *Global Journal of Engineering* and *Technology Advances*, 2(3):054-066. Available from:https://doi.org/DOI:10.30574/gjeta.2020.2.3.0019.
- Molepo, G. 2015. The use of technology to enhance administration of schools in Kgetleng river area office: is it really beneficial to school administration? SOUTH AFRICA INTERNATIONAL CONFERENCE ON EDUCATIONAL TECHNOLOGIES. Available from.
- Moyo, H. 2018. Effectiveness of the information for planning purposes with particular reference to the South African-school administration management systems in John Taolo Gaetsewe District in the Northern Cape. Available from: <u>http://hdl.handle.net/11602/1211</u>.
- Mukherjee, N., Sutherland, W.J., Khan, M.N.I., Berger, U., Schmitz, N., Dahdouh-Guebas, F. & Koedam, N. 2014. Using expert knowledge and modeling to define mangrove composition, functioning, and threats and estimate time frame for recovery. *Ecology and Evolution*, 4(11):2247-2262. Available from: <a href="https://doi.org/DOI:10.1002/ece3.1085">https://doi.org/DOI:10.1002/ece3.1085</a>.
- Munn, Z., Peters, M.D., Stern, C., Tufanaru, C., Mcarthur, A. & Aromataris, E. 2018. Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach. *BMC Medical Research Methodology*, 18(1):1-7. Available from:<u>https://doi.org/DOI:10.1186/s12874-018-0611-x</u>.
- Muriithi, G.M. & Masinde, M. 2016. Cloud SAMS: Cloud computing solution for public schools within South Africa's 'second economy'. 2016 IST-Africa Week Conference. Available from:<u>https://doi.org/DOI:10.1109/ISTAFRICA.2016.7530689</u>.
- Mwalongo, A. 2011. Teachers' perceptions about ICTs for teaching, professional development, administration and personal use. *International journal of Education and Development using ICT*, 7(3):36-49. Available from: https://www.learntechlib.org/p/42357/.
- Mwangi, T.M. 2014. Sustainable School management systems in Kenyan secondary schools: A comparison between cloud based systems versus on premise based systems. Masters. University of Nairobi. Available from: <a href="http://41.204.161.209/handle/11295/74011">http://41.204.161.209/handle/11295/74011</a>.
- Mythili, G., Mishtra, A. & Singh, N. 2018. Block-4 Support Systems, Legal and Ethical Issues. Available from: <a href="http://egyankosh.ac.in//handle/123456789/46326">http://egyankosh.ac.in//handle/123456789/46326</a>.
- Nasrabadi, N., Dehnokhalaji, A., Korhonen, P. & Wallenius, J. 2019. A stepwise benchmarking approach to DEA with interval scale data. *Journal of the Operational Research Society*, 70(6):954-961. Available from:https://doi.org/DOI:10.1080/01605682.2018.1471375.
- Ncanywa, T. & Sibiya, Z. 2020. A case study of what determines learner performance in a combined school in Mpumalanga Province. Available from:<u>http://ulspace.ul.ac.za/handle/10386/3228?show=full</u>.
- Ndaba, E.M. 2021. Generation, management and use of school data by the school management team in one primary school. Masters. Available from:<u>https://ukzn-dspace.ukzn.ac.za/handle/10413/20480</u>.
- Nemoto, T. & Beglar, D. 2014. Likert-scale questionnaires. JALT 2013 conference proceedings. Available from: <a href="https://jalt-publications.org/sites/default/files/pdf-article/jalt2013">https://jalt-publications.org/sites/default/files/pdf-article/jalt2013</a> 001.pdf.

- Neuendorf, K.A. 2018. Content analysis and thematic analysis. Advanced research methods for applied psychology. Routledge. Available from:<u>https://www.taylorfrancis.com/chapters/edit/10.4324/9781315517971-21/content-analysis-thematic-analysis-kimberly-neuendorf</u>.
- Novakouski, M. & Lewis, G.A. 2012. Interoperability in the e-Government Context. In: (ed.). Carnegie-Mellon Univ Pittsburgh PA Software Engineering Inst Available from:<u>https://resources.sei.cmu.edu/asset\_files/TechnicalNote/2012\_004\_001\_15365.pdf</u>.
- Nsaghurwe, A., Dwivedi, V., Ndesanjo, W., Bamsi, H., Busiga, M., Nyella, E., Massawe, J.V., Smith, D., Onyejekwe, K., Metzger, J. & Taylor, P. 2021. One country's journey to interoperability: Tanzania's experience developing and implementing a national health information exchange. *BMC Medical Informatics and Decision Making*, 21(1):139. Available from:<u>https://doi.org/DOI:10.1186/s12911-021-01499-6</u>.
- Nureni, Y. 2014. Information communication technology (ICT)[Concepts and application]. *Nigeria: Hasfem Publication Center*:13-15. Available from:<u>http://doi.org/DOI:10.13140/RG.2.1.1802.7289</u>.
- Nvivo. 2021. *Homepage*. [Online]. Available from: <u>https://www.qsrinternational.com/</u> [Accessed 31 May 2021].
- Nworie, J. 2011. Using the Delphi technique in educational technology research. *TechTrends*, 55(5):24-30. Available from: <u>https://link.springer.com/article/10.1007/s11528-011-0524-6</u>.
- Ohwo, K.O. 2020. INTEROPERABILITY AND EFFECTIVENESS OF INDUSTRIES IN THE SUB-SAHARAN AFRICA (SSA) EVIDENCE FROM NESTLE NIGERIA PLC. *Management & Marketing Journal*, 18(2). Available from:<u>http://mnmk.ro/en/documents/2020\_2/4-2-2-20.pdf</u>.
- Olaore, I.B. 2014. *The impacts (positive and negative) of ICT on education in Nigeria*. PhD. Available from: <u>https://core.ac.uk/download/pdf/234682114.pdf</u>.
- Ostadzadeh, S.S., Shams, F. & Badie, K. 2015. An architectural model framework to improve digital ecosystems interoperability. *New trends in networking, computing, e-learning, systems sciences, and engineering.* Springer. Available from:<u>http://doi.org/DOI:10.1007/978-3-319-06764-3\_65</u>.
- Othman, M.F.I. 2016. *Barriers to the adoption of formal IT governance practice: A Malaysian case.* PhD. Queensland University of Technology. Available from: <u>https://eprints.qut.edu.au/95791/</u>.
- Ożadowicz, A. 2020. Modified blended learning in engineering higher education during the COVID-19 lockdown—Building automation courses case study. *Education sciences*, 10(10):292. Available from:https://doi.org/DOI:10.3390/educsci10100292.
- Paraskevas, A. & Saunders, M.N. 2012. Beyond consensus: an alternative use of Delphi enquiry in hospitality research. *International Journal of Contemporary Hospitality Management*. Available from:<u>https://doi.org/DOI:10.1108/09596111211247236</u>.
- Patel, K.K., Patel, S.M. & Scholar, P. 2016. Internet of things-IOT: definition, characteristics, architecture, enabling technologies, application & future challenges. *International journal of engineering science* and *computing*, 6(5). Available from:<u>http://www.opjstamnar.com/download/Worksheet/Day-110/IP-XI.pdf</u>.

- Pholotho, T.J. 2017. Toward a Broadband Services Delivery Model over Wireless Technologies to Resource-Constrained Public High Schools in South Africa. Masters. Available from:https://core.ac.uk/download/pdf/95521706.pdf.
- Picciotto, R. 2011. The logic of evaluation professionalism. *Evaluation*, 17(2):165-180. Available from: <a href="https://doi.org/DOI:10.1177/1356389011403362">https://doi.org/DOI:10.1177/1356389011403362</a>.
- Pillay, D. 2020. Exploring South African primary school teachers' experiences on the implementation of the school administration and management system. Masters. Available from:<u>https://ukzn-dspace.ukzn.ac.za/handle/10413/19229</u>.
- Pöppelbuß, J. & Röglinger, M. 2011. What makes a useful maturity model? A framework of general design principles for maturity models and its demonstration in business process management. Available from:<u>https://fim-rc.de/Paperbibliothek/Veroeffentlicht/327/wi-327.pdf</u>.
- Preskill, H. & Russ-Eft, D.F. 2015. *Building evaluation capacity: Activities for teaching and training.* Sage Publications. Available from: <u>https://doi.org/DOI:10.1177/1098214017746499</u>.
- Proença, D. & Borbinha, J. 2016. Maturity models for information systems-a state of the art. *Procedia Computer Science*, 100:1042-1049. Available from.
- Purdey, B. 2013. Occupant stimulus response workplace productivity and the vexed question of measurement. *Facilities*. Available from: <u>https://doi.org/DOI:10.1108/F-03-2012-0021</u>.
- Qureshi, Z.H. & Qazi Abro, M. 2016. Efficient use of ict in administration. *International Journal of Economics, Commerce and Management, IV (10)*:540-550. Available from:https://ijecm.co.uk/wp-content/uploads/2016/10/41030.pdf.
- Raaijmakers, Q.A., Van Hoof, J., T Hart, H., Verbogt, T. & Vollebergh, W.A. 2000. Adolescents' midpoint responses on Likert-type scale items: neutral or missing values? *International Journal of Public Opinion Research*, 12:208-216. Available from: <u>https://doi.org/DOI:10.1093/ijpor/12.2.209</u>.
- Rajhans, V., Rege, S., Memon, U. & Shinde, A. 2020. Adopting a modified Delphi technique for revisiting the curriculum: A useful approach during the COVID-19 pandemic. *Qualitative Research Journal*. Available from:<u>https://doi.org/DOI:10.1108/QRJ-05-2020-0043</u>.
- Rajkumar, N., Muzoora, M. & Thun, S. 2022. Dentistry and interoperability. *Journal of Dental Research*:00220345221100175. Available from:https://doi.org/DOI:10.1177/00220345221100175.
- Rakhee, D. & Era, G. 2021. *POPIA compliance: what schools, colleges and universities need to consider.* [Online]. Available from: <u>https://www.lexology.com/library/detail.aspx?g=7bad4c1a-94b2-4be7-86aa-033a2fd2d532</u> [Accessed 19 May 2023].
- Ramazhamba, P.T., Mashiane, T. & Dlamini, Z. 2018. Grade monitoring system: A prototype for Thulamela secondary schools. 2018 IST-Africa Week Conference (IST-Africa). Available from:<u>https://ieeexplore.ieee.org/document/8417327</u>.

- Rana, K. & Rana, K. 2020. ICT Integration in Teaching and Learning Activities in Higher Education: A Case Study of Nepal's Teacher Education. *Malaysian Online Journal of Educational Technology*, 8(1):36-47. Available from:<u>https://files.eric.ed.gov/fulltext/EJ1239982.pdf</u>.
- Ratheeswari, K. 2018. Information communication technology in education. *Journal of Applied and Advanced research*, 3(1):45-47. Available from: <u>https://doi.org/DOI:10.21839/jaar.2018.v3iS1.169</u>.
- Ratheeswari, K. 2020. ICT: A BOON TO TEACHER EDUCATION. *Journal of Natural Remedies*, 21(3):157-159. Available from: <u>https://doi.org/DOI:10.21839/jaar.2018.v3iS1.169</u>.
- Raynard, M., Johnson, G. & Greenwood, R. 2015. Institutional theory and strategic management. *Advanced strategic management: A multi-perspective approach*:9-34. Available from:<u>https://www.wu.ac.at/fileadmin/wu/d/i/cmmd/Mia\_Paper-</u> <u>Links/Raynard Johnson Greenwood Strategy Chapter 2014 .pdf</u>.
- Reddy, V., Prinsloo, C.H., Netshitangani, T., Moletsane, R., Juan, A. & Janse Van Rensburg, D. 2010. An investigation into educator leave in the South African ordinary public schooling system. Available from:<u>https://hsrc.ac.za/uploads/pageContent/593/AnInvestigationintoEducatorLeavedec2010.p</u> df.
- Reeves, G. & Jauch, L.R. 1978. Curriculum development through Delphi. *Research in Higher Education,* 8(2):157-168. Available from:<u>https://link.springer.com/article/10.1007/BF00992116</u>.
- Rehman, A.A. & Alharthi, K. 2016. An introduction to research paradigms. International Journal of<br/>Educational Investigations, 3(8):51-59. Available<br/>from:<a href="https://www.semanticscholar.org/paper/An-Introduction-to-Research-Paradigms-</a><br/>Rehman-Alharthi/84ff3551748fcecb99d15bdf90b0c428d7dd14a3.
- Rezaei, R., Chiew, T.K., Lee, S.P. & Aliee, Z.S. 2014. Interoperability evaluation models: A systematic review. *Computers in Industry*, 65(1):1-23. Available from:https://doi.org/DOI:10.1016/j.compind.2013.09.001.
- Roopa, S. & Rani, M. 2012. Questionnaire designing for a survey. *Journal of Indian Orthodontic Society,* 46(4\_suppl1):273-277. Available from.
- Rowe, G. & Wright, G. 2001. Expert opinions in forecasting: the role of the Delphi technique. *Principles of forecasting*. Springer. Available from: <u>https://doi.org/DOI:10.1007/978-0-306-47630-3\_7</u>.
- Rowe, G., Wright, G. & Bolger, F. 1991. Delphi: a reevaluation of research and theory. *Technological Forecasting and Social Change*, 39(3):235-251. Available from: <u>https://doi.org/DOI:10.1016/0040-1625(91)90039-I</u>.
- Ryan, M. 1980. The Likert scale's midpoint in communications research. *Journalism Quarterly*, 57(2):305-313. Available from: <a href="https://doi.org/DOI:10.1177/10776990800570021">https://doi.org/DOI:10.1177/10776990800570021</a>.
- Saadi, K. 2017. Web application for student follow-up at educational institutions. Masters. Faculté des Mathématiques et de l'Informatique-Université Mohamed BOUDIAF-M'sila. Available from: http://dspace.univ-msila.dz:8080//xmlui/handle/123456789/2198.

#### KENEILWE MAREMI: 44977883

- Sampathkumar, A. 2020. Change is inevitable. 28(1):5-6. Available from: <a href="https://doi.org/DOI:10.1177/0218492319897805">https://doi.org/DOI:10.1177/0218492319897805</a>.
- Sansa, N.A. 2020. Analysis for the Impact of the COVID-19 to the Petrol Price in China. Available at SSRN 3547413. Available from: http://dx.doi.org/DOI:10.2139/ssrn.3547413.
- Saquin, M.D.C. & Marcial, D.E. 2016. Developing an Automated Student Academic Record Management with Business Intelligence Approach. *Jurnal Informatika*, 12(2). Available from:<u>https://ti.ukdw.ac.id/ojs/index.php/informatika/article/view/453/236</u>.
- Sarantis, D., Charalabidis, Y. & Psarras, J. 2008. Towards standardising interoperability levels for information systems of public administrations. *The Electronic Journal for E-commerce Tools & Applications (eJETA) Special Issue on Interoperability for Enterprises and Administrations Worldwide,* 2:1-15. Available from:<u>https://www.academia.edu/2852708/Towards standardising interoperability\_levels\_for information systems of public administrations</u>.
- Sarkar, S. 2012. The role of information and communication technology (ICT) in higher education for the 21st century. *Science*, 1(1):30-41. Available from:<u>https://silo.tips/download/the-role-of-information-and-communication-technology-ict-in-higher-education-for</u>.
- Sarker, A. 2016. Online school management system. Available from: <u>http://dspace.bracu.ac.bd/xmlui/handle/10361/9154</u>.
- Saunders, M. & Lewis, P. 2017. Doing Research in Business and Management. An essential guide to planning your project, 2nd ed. United Kingdom: Pearson. Available from.
- Saunders, M., Lewis, P. & Thornhill, A. 2016. Research Methods for Business Students. Prentice Hall. [Online]. Available from: <u>https://pubhtml5.com/enuk/rykz/CHAPTER\_1-7\_research-methods-for-business-students-eighth-edition-v3f-2/</u> [Accessed 24 January 2022].
- Schumacher, A., Erol, S. & Sihn, W. 2016. A maturity model for assessing Industry 4.0 readiness and maturity of manufacturing enterprises. *Procedia CIRP*, 52:161-166. Available from:<u>https://doi.org/DOI:10.1016/j.procir.2016.07.040</u>.
- Schwab, K. 2016a. *The Fourth Industrial Revolution: what it means, how to respond.* [Online]. Available from: <u>https://www.weforum.org/agenda/2016/01/the-fourth-industrial-revolution-what-it-means-and-how-to-respond/</u> [Accessed 1].
- Schwab, K. 2016b. *The Fourth Industrial Revolution: what it means, how to respond.* [Online]. Available from: <u>https://www.weforum.org/agenda/2016/01/the-fourth-industrial-revolution-what-it-means-and-how-to-respond/</u> [Accessed 17 May 2022].
- Schwab, K. 2019. Why we need the "Davos Manifesto" for a better kind of capitalism. [Online]. Available from: <u>https://www.weforum.org/agenda/2019/12/why-we-need-the-davos-manifesto-forbetter-kind-of-capitalism/</u> [Accessed 20 January 2022].
- Seyal, A. 2012. A preliminary study of school administrators' use of information & communication technologies: Bruneian perspective. International journal of Education and Development using ICT, 8(1):29-45. Available from:<u>https://eric.ed.gov/?id=EJ1084147</u>.

- Shaikjee, F. 2020. Exploring the use of ICTs in education: evidences from South African secondary school managers. Available from.
- Shariff, N. 2015. Utilizing the Delphi survey approach: A review. *J Nurs Care*, 4(3):246. Available from:<u>https://ecommons.aku.edu/cgi/viewcontent.cgi?article=1037&context=eastafrica\_fhs\_son</u> <u>am</u>.
- Shariff, N.M. & Abd Razak, R. 2022. Exploring hospitality graduates' competencies in Malaysia for future employability using Delphi method: a study of Competency-Based Education. *Journal of Teaching in Travel & Tourism,* 22(2):144-162. Available from:https://doi.org/DOI:10.1080/15313220.2021.1950103.
- Shava, E. & Hofisi, C. 2017. Challenges and opportunities for public administration in the fourth industrial revolution. *African Journal of Public Affairs*, 9(9):203-215. Available from:<u>https://journals.co.za/doi/abs/10.10520/EJC-c13dbfea9</u>.
- Sibiya, Z.C. 2019. Determinants of learner perfomance in a combined school in Mpumalanga Province: education production function approach. Masters. Available from:http://ulspace.ul.ac.za/bitstream/handle/10386/2979/sibiya\_zc\_2019.pdf?sequence=1.
- Silva, C., Ribeiro, P., Pinto, E.B. & Monteiro, P. 2021. Maturity Model for Collaborative R&D University-Industry Sustainable Partnerships. *Procedia Computer Science*, 181:811-817. Available from:<u>https://doi.org/DOI:10.1016/j.procs.2021.01.234</u>.
- Singh, S., Nicely, A., Cai, L. & Day, J. 2021. Visitor (tourist) harassment research: Delphi panels. Available from:<u>https://docs.lib.purdue.edu/cgi/viewcontent.cgi?article=1002&context=htmvh</u>.
- Sitlington, H. & Coetzer, A. 2015. Using the Delphi technique to support curriculum development. *Education+ Training*. Available from:<u>https://doi.org/DOI:10.1108/ET-02-2014-0010</u>.
- Sivalingam, A.D. 2020. A Research Study by Delphi Technique in School Counseling. Available from: <a href="https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=3574362">https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=3574362</a>.
- Sjarov, M., Kißkalt, D., Lechler, T., Selmaier, A. & Franke, J. 2021. Towards "Design for Interoperability" in the context of Systems Engineering. *Procedia CIRP*, 96:145-150. Available from:<u>https://doi.org/DOI:10.1016/j.procir.2021.01.067</u>.
- Skulmoski, G.J., Hartman, F.T. & Krahn, J. 2007. The Delphi method for graduate research. *Journal of Information Technology Education: Research,* 6(1):1-21. Available from:<u>https://doi.org/DOI:10.28945/199</u>.
- Smith, K.S. & Simpson, R.D. 1995. Validating teaching competencies for faculty members in higher education: A national study using the Delphi method. *Innovative higher education*, 19(3):223-234. Available from:<u>https://link.springer.com/content/pdf/10.1007/BF01191221.pdf</u>.
- Somerville, J. 2008. Critical factors affecting the assessment of student learning outcomes: A Delphi study of the opinions of community college personnel. Dissertation. Available from:<u>https://www.proquest.com/openview/0bf2c8d11fbbb7409a87c5ae3bf1d9d3/1?pq-</u> origsite=gscholar&cbl=18750.

- Sosnytskyi, Y.O., Sikorskyi, P.I., Bezborodykh, S.M., Morozova, M.M. & Moroz, V.P. 2021. Application of the Delphi Technique to Determine the Technological Competencies of a Faculty Member. *European Journal of Educational Research*, 10(4):2089-2103. Available from:https://doi.org/DOI:10.12973/eu-jer.10.4.2089.
- Ssempala, M., Kabuunga, H., Taremwa, J. & Magero, T. 2018. Academic school management mobile application. In: (ed.). Available from: <u>http://hdl.handle.net/20.500.12281/4356</u>.
- Starman, A.B. 2013. The case study as a type of qualitative research. Journal of Contemporary EducationalStudies/SodobnaPedagogika,64(1).Availablefrom:<a href="https://www.academia.edu/40744063/The\_case\_study\_as\_a\_type\_of\_qualitative">https://www.academia.edu/40744063/The\_case\_study\_as\_a\_type\_of\_qualitative</a> research JOURNAL OF CONTEMPORARY EDUCATIONAL STUDIES 1 2013 28 43.
- Stevens, S.S. 1946. On the theory of scales of measurement. *Science*, 103(2684):677-680. Available from: <a href="https://doi.org/DOI:10.1126/science.103.2684.677">https://doi.org/DOI:10.1126/science.103.2684.677</a>.
- Stibel, J.M., Stibel, A.B., Hackett, J.G., Kazerani, M. & Loeb, J. 2015. *Indices for credibility trending, monitoring, and lead generation*. [Online]. Available from: <a href="https://patents.google.com/patent/US9202200">https://patents.google.com/patent/US9202200</a> [Accessed 2 April 2020].
- Strydom, H.M.J. 2020. Teachers' Use of a Web-Based Application of the Special Needs Assessment Forms. University of Johannesburg (South Africa). Available from:<u>https://www.proquest.com/openview/79371e40dc596b6641c8b37cb474f661/1?pq-</u> origsite=gscholar&cbl=2026366&diss=y.
- Suleiman, M.M.M., Zakari, I., Sani, S. & Ukashatu, A. 2020. Role of ICT for Authentic Assessment in Higher Education. *Tathapi Journal UGC Care*, 19(38):128-136. Available from:<u>https://www.academia.edu/43507290/Role\_of\_ICT\_for\_Authentic\_Assessment\_in\_Higher\_Education</u>.
- Sullivan, G.M. & Artino, A.R. 2013. Analysing and Interpreting Data From Likert-Type Scales. *Journal of Graduate Medical Education*, 5(4):541-542. Available from:<u>https://doi.org/DOI:10.4300/JGME-5-4-18</u>.
- Suparji, S., Nugroho, H.S.W. & Martiningsih, W. 2021. Tips for Distinguishing Nominal and Ordinal Scale Data. Aloha International Journal of Multidisciplinary Advancement (AIJMU), 1(6):133-135. Available from: <u>https://doi.org/DOI:10.33846/aijmu10602</u>.
- Taherdoost, H. 2019. What is the best response scale for survey and questionnaire design; review of different lengths of rating scale/attitude scale/Likert scale. *Hamed Taherdoost*:1-10. Available from:https://ssrn.com/abstract=3588604.
- Tapio, P., Paloniemi, R., Varho, V. & Vinnari, M. 2011. The unholy marriage? Integrating qualitative and quantitative information in Delphi processes. *Technological Forecasting and Social Change*, 78(9):1616-1628. Available from:<u>https://doi.org/DOI:10.1016/j.techfore.2011.03.016</u>.
- Taylor, R.M., Feltbower, R.G., Aslam, N., Raine, R., Whelan, J.S. & Gibson, F. 2016. Modified international e-Delphi survey to define healthcare professional competencies for working with teenagers and young adults with cancer. *BMJ Open*, 6(5):e011361. Available from:<u>https://doi.org/DOI:10.1136/bmjopen-2016-011361</u>.

- Thwala, N.P.N. 2019. Exploring how a district teacher development centre supports teachers' professional learning: a case study in Phumelela district. Available from:https://doi.org/DOI:10.1080/02331934.2014.926356.
- Triantafyllidis, C. & Samaras, N. 2015. Three nearly scaling-invariant versions of an exterior point algorithm for linear programming. *Optimization*, 64(10):2163-2181. Available from:<u>https://doi.org/DOI:10.1080/02331934.2014.926356</u>.
- Turoff, M. 1970. The design of a policy Delphi. *Technological Forecasting and Social Change*, 2(2):149-171. Available from: <u>https://doi.org/DOI:10.1016/0040-1625(70)90161-7</u>.
- Vaismoradi, M., Turunen, H. & Bondas, T. 2013. Content analysis and thematic analysis: Implications for conducting a qualitative descriptive study. *Nursing & health sciences*, 15(3):398-405. Available from:<u>https://doi.org/DOI:10.1111/nhs.12048</u>.
- Van Alsenoy, B., Coudert, F., Jasmontaite, L. & Verdoodt, V. 2015. Cultures of Accountability: A crosscultural perspective on current and future accountability mechanisms. Available from:<u>https://doi.org/DOI:10.1016/j.obhdp.2011.11.001</u>.
- Van Der Berg, S., Van Wyk, C., Selkirk, R., Rich, K. & Deghaye, N. 2019. The promise of SA-SAMS & DDD data for tracking progression, repetition and drop-out. In: (ed.). Available from: <a href="https://resep.sun.ac.za/the-promise-of-sa-sams-ddd-data-for-tracking-progression-repetition-and-drop-out/sample-post/">https://resep.sun.ac.za/the-promise-of-sa-sams-ddd-data-for-tracking-progression-repetition-and-drop-out/sample-post/</a>.
- Van Der Berg, S., Wills, G., Selkirk, R., Adams, C. & Van Wyk, C. 2019. The cost of repetition in South Africa. Available from:<u>https://doi.org/DOI:10.2139/ssrn.3505854</u>.
- Van Staden, S. & Mbale, J. 2012. The Information Systems Interoperability Maturity Model (ISIMM): towards standardizing technical interoperability and assessment within government. International Journal of Information Engineering and Electronic Business, 4(5):36. Available from:http://dx.doi.org/DOI:10.5815/ijieeb.2012.05.05.
- Van Wyk, C. 2021. Learner flow through patterns in the Western Cape using CEMIS datasets from 2007 to 2019: A longitudinal cohort analysis. Available from:<u>https://resep.sun.ac.za/wp-content/uploads/2021/06/wp012021-2.pdf</u>.
- Van Wyk, C., Gondwe, A. & De Villiers, P. 2017. Learner flow through patterns in the Western Cape using CEMIS datasets from 2007 to 2014: A longitudinal cohort analysis. In: (ed.). Available from:<u>https://ideas.repec.org/p/sza/wpaper/wpapers361.html</u>.
- Varatharajan, R., Manogaran, G., Priyan, M., Balaş, V.E. & Barna, C. 2018. Visual analysis of geospatial habitat suitability model based on inverse distance weighting with paired comparison analysis. *Multimedia tools and applications*, 77(14):17573-17593. Available from:<u>https://doi.org/DOI:10.1007/s11042-017-4768-9</u>.
- Veerasamy, A.K. & De Souza-Daw, T. 2012. Impact of ICT on society—Higher education students in South-East Asia. 2012 IEEE Symposium on Business, Engineering and Industrial Applications. Available from:<u>https://cmaaustralia.edu.au/wp-content/uploads/2021/10/JAMARv10.1-Research\_Noteon-Research-Methods.pdf</u>.

- Vernon, W. 2009. The Delphi technique: a review. *International Journal of Therapy and rehabilitation*, 16(2):69-76. Available from:<u>https://doi.org/DOI:10.12968/IJTR.2009.16.2.38892</u>.
- Vesper, J., Reeves, T. & Herrington, J. 2011. The application of expert review as a formative evaluation strategy within an educational design research study. E-Learn: World Conference on E-Learning in Corporate, Government, Healthcare, and Higher Education. Available from:https://www.learntechlib.org/primary/p/38835/.
- Wach, E. & Ward, R. 2013. Learning about qualitative document analysis. Available from: <a href="https://www.ids.ac.uk/publications/learning-about-qualitative-document-analysis/">https://www.ids.ac.uk/publications/learning-about-qualitative-document-analysis/</a>.
- Wager, K.A., Lee, F.W. & Glaser, J.P. 2017. Laws, Regulations, and Standards That Affect Health Care Information Systems. *Health Care Information Systems: A Practical Approach for Health Care Management*
- 4ed. San Francisco: John Wiley & Sons. Available from:<u>https://www.amazon.com/Health-Care-Information-Systems-Management/dp/1119337186</u>.
- Wahyuni, D. 2012. The research design maze: Understanding paradigms, cases, methods and methodologies. *Journal of Applied Management Accounting Research*, 10(1):69-80. Available from:<u>https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=2103082</u>.
- Walsh, K. 2022. Future of online learning Ed-Admin. Available from:<u>https://www.ed-admin.com/ed-admin-resources/blog/future-of-online-learning</u>.
- Wanzer, D.L. 2021. What Is Evaluation?: Perspectives of How Evaluation Differs (or Not) From Research.AmericanJournalofEvaluation,42(1):28-46.Availablefrom:<a href="https://doi.org/DOI:10.31234/osf.io/v9g8f">https://doi.org/DOI:10.31234/osf.io/v9g8f</a>.
- Warnke, G. 2011. The hermeneutic circle versus dialogue. *The Review of Metaphysics*, 65(1):91-112. Available from: <a href="https://www.jstor.org/stable/23055684#metadata">https://www.jstor.org/stable/23055684#metadata</a> info tab contents.
- Weber-Jahnke, J., Peyton, L. & Topaloglou, T. 2012. eHealth system interoperability. *Information Systems Frontiers*, 14(1):1-3. Available from: <u>https://doi.org/DOI:10.1007/s10796-011-9319-8</u>.
- Wei, L.M., Piaw, C.Y., Kannan, S. & Moulod, S.A. 2016. Relationship between Teacher ICT Competency and Teacher Acceptance and Use of School Management System (SMS). *Malaysian Online Journal of Educational Technology*, 4(4):36-52. Available from:<u>https://eric.ed.gov/?id=EJ1116214</u>.
- Worldbank. 2020. Guidance Note: Remote Learning and COVID-19. In: (ed.). World Bank Available from:<u>https://documents1.worldbank.org/curated/en/531681585957264427/pdf/Guidance-Note-on-Remote-Learning-and-COVID-19.pdf</u>.
- Wozak, F., Ammenwerth, E., Hörbst, A., Sögner, P., Mair, R. & Schabetsberger, T. 2008. IHE based Interoperability-Benefits and Challenges. MIE. Available from: <u>https://doi.org/DOI:10.3233/978-1-58603-864-9-771</u>.
- Wu, H. & Leung, S.-O. 2017. Can Likert scales be treated as interval scales?—A Simulation study. Journal of Social Service Research, 43(4):527-532. Available from:<u>https://doi.org/DOI:10.1080/01488376.2017.1329775</u>.

- Xavier, A., Reyes, T., Aoussat, A., Luiz, L. & Souza, L. 2020. Eco-innovation maturity model: A framework to support the evolution of eco-innovation integration in companies. *Sustainability*, 12(9):3773. Available from:<u>https://doi.org/DOI:10.3390/su12093773</u>.
- Yeh, J.S., Van Hoof, T.J. & Fischer, M.A. 2016. Key features of academic detailing: development of an expert consensus using the Delphi method. *American health & drug benefits*, 9(1):42. Available from:<u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4822978/</u>.
- Yin, R.K. 2009. *Case study research: Design and methods.* sage. Available from:<u>https://doi.org/DOI:10.33524/cjar.v14i1.73</u>.
- Young, T.J. 2015. Questionnaires and surveys. *Research methods in intercultural communication: A practical guide*:163-180. Available from:<u>https://doi.org/DOI:10.1002/9781119166283.ch11</u>.
- Yousuf, M.I. 2007. The delphi technique. *Essays in Education*, 20(1):8. Available from: <u>https://openriver.winona.edu/eie/vol20/iss1/8/</u>.
- Zeid, A., Sundaram, S., Moghaddam, M., Kamarthi, S. & Marion, T. 2019. Interoperability in smart manufacturing: Research challenges. *Machines*, 7(2):21. Available from:<u>https://doi.org/DOI:10.3390/MACHINES7020021</u>.
- Zeng, M.L. 2019. Interoperability. *KO Knowledge Organization*, 46(2):122-146. Available from: <a href="https://doi.org/DOI:10.5771/0943-7444-2019-2">https://doi.org/DOI:10.5771/0943-7444-2019-2</a>.
- Zhong, R. 2020. The coronavirus exposes education's digital divide. *The New York Times*, 18:2020. Available from:<u>https://www.nytimes.com/2020/03/17/technology/china-schools-</u> <u>coronavirus.html</u>.
- Zhu, Y., Yang, C., Zhang, J. & Chen, B. 2021. Developing an empathy educational model (EEM) for undergraduate nursing students: A Delphi Technique. *Nurse Education in Practice*, 50:102922. Available from:<u>https://doi.org/DOI:10.1016/j.nepr.2020.102922</u>.

### **APPENDIX A: Ethics clearance certificate**



#### UNISA COLLEGE OF SCIENCE, ENGINEERING AND TECHNOLOGY ETHICS **REVIEW COMMITTEE**

2021-11-02

Dear Ms KJ Maremi

ERC Reference # :2021/CSET/SOC/070 Name : KJ Maremi Student #: 44977883 Staff #:

**Decision: Ethics Approval from** 2021/11/02 for three years

Researcher(s):

Ms KJ Maremi; 44977883@mylife.unisa.ac.za; 0846164296

Prof M Herselman; MHerselman@csir.co.za; 012 841 3081 Supervisor (s): Prof A Botha, Abotha@csir.co.za; 012 841 3265

Working title of research: A School Management System Interoperability Maturity (SMSIM) Model for Private Schools in South Africa

Qualification: MSc Computing

Thank you for the application for research ethics clearance by the Unisa College of Science Engineering and Technology Ethics Review Committee for the above-mentioned research. Ethics approval is granted for 3 years.

The low risk application was reviewed by the College of Science, Engineering and Technology (CSET) Ethics Review Committee on 2021-11-02 in compliance with the Unisa Policy on Research Ethics and the Standard Operating Procedure on Research Ethics Risk Assessment. The decision will be tabled at the next Committee meeting for ratification. The proposed research may now commence with the provisions that:

- 1. The researcher(s) will ensure that the research project adheres to the values and principles expressed in the UNISA Policy on Research Ethics.
- 2. Any adverse circumstance arising in the undertaking of the research project that is relevant to the ethicality of the study should be communicated in writing to the SOC Ethics Committee.
- 3. The researcher(s) will conduct the study according to the methods and procedures set out in the approved application.



University of South Africa Prelier Street, Muckleneuk Ridge, City of Tshwane PO Box 392 UNISA 0003 South Africa Telephone: +27 12 429 3111 Facsimile: +27 12 429 4150

- 4. Any changes that can affect the study-related risks for the research participants, particularly in terms of assurances made with regards to the protection of participants' privacy and the confidentiality of the data, should be reported to the Committee in writing, accompanied by a progress report.
- 5. The researcher will ensure that the research project adheres to any applicable national legislation, professional codes of conduct, institutional guidelines and scientific standards relevant to the specific field of study. Adherence to the following South African legislation is important, if applicable: Protection of Personal Information Act, no 4 of 2013; Children's act no 38 of 2005 and the National Health Act, no 61 of 2003.
- 6. Only de-identified research data may be used for secondary research purposes in future on condition that the research objectives are similar to those of the original research. Secondary use of identifiable human research data requires additional ethics clearance.
- No field work activities may continue after the expiry date (2024-11-02). Submission
  of a completed research ethics progress report will constitute an application for
  renewal of Ethics Research Committee approval.

Note:

The reference number 2021/CSET/SOC/070 should be clearly indicated on all forms of communication with the intended research participants, as well as with the Committee.

Yours sincerely, HALOUIUL

Prof HH Lotriet

Chair of the Department of Information Systems Ethics Review Subcommittee College of Science, Engineering and Technology (CSET) e-mail : <u>lotrihh@unisa.ac.za</u>

Tel : 011 471 2661



Prof E Mnkandla Director : School of Computing College of Science Engineering and Technology (CSET) E-mail : <u>mnkane@unisa.ac.za</u> Tel : (011) 670 9104

Aule pp

Prof BB Mamba Executive Dean College of Science Engineering and Technology (CSET) E-mail : <u>mambabb@unisa.ac.za</u> Tel : (011) 670 9230

URERC 25.04.17 - Decision template (V2) - Approve

University of South Africa Prelier Street, Muckleneuk Ridge, City of Tshwane PO Box 392 UNISA 0003 South Africa Telephone: +27 12 429 3111 Facsimile: +27 12 429 4150 www.unisa.ac.za

# **APPENDIX B: Informed consent form**

### PARTICIPANT INFORMATION SHEET

Ethics clearance reference number: 2021-CSET-SOC-070

30 November 2021

Title: A School Management Systems Interoperability Maturity (SMSIM) Model for Private Schools in South Africa.

**Dear Prospective Participant** 

My name is Keneilwe Maremi and I am doing research with Prof Marlien Herselman, an adjunct Professor in the Department of Science, Engineering and Technology (School of Computing) towards an MSC in Computing at the University of South Africa. We invite you to participate in a study entitled: School Management Systems Interoperability Maturity (SMSIM) Model for Private Schools in South Africa.

### WHAT IS THE PURPOSE OF THE STUDY?

The purpose of the study is to conceptualize a School Management Systems Interoperability Maturity (SMSIM) model for private schools in South Africa. This conceptualized SMSIM model will apply a specific Interoperability Maturity Model to assess the level of interoperability in SMS and provide clear guidance on how SMS can benefit from using interoperability maturity levels to improve their current SMS systems. In addition, this model may enhance the level of interoperability maturity in SMS and highlight the value of such a model for existing SMS in schools.

### WHY AM I BEING INVITED TO PARTICIPATE?

You have been identified as a participant because you have a minimum of 10 years experience in one of the following fields: ICT in Education, Interoperability, Schools Management Systems (SMS), and Information Technology. The study aims to conceptualise a model that will help improve SMS and your input in validating and verifying the model will add value to any school that aims to improve their SMS.

#### WHAT IS THE NATURE OF MY PARTICIPATION IN THIS STUDY?

The study uses the Delphi method which is a process used to arrive at a group opinion or decision by surveying a panel of experts. Interviews will take place in several rounds until a consensus is reached. The advantage of this method is location: Interviews can be conducted without face-to-face contact. The questions that will be asked are about SMS, and challenges faced in schools when submitting data to the Department of Education and daily administrative tasks, also how interoperability can help mitigate these problems. The interview is expected to take 30-45 minutes. You will be expected to anonymously answer these questions and evaluate a copy of the other experts' feedback without knowing the identity of the experts, then based on the first round of feedback from experts, more questions will be developed, and you will also need to participate in the second round, this process will repeat itself over and over until a consensus is reached.

# CAN I WITHDRAW FROM THIS STUDY EVEN AFTER HAVING AGREED TO PARTICIPATE?

Participating in this study is voluntary and you are under no obligation to consent to participation. If you do decide to take part, you will be given this information sheet to keep and be asked to sign a written consent form. You are free to withdraw at any time and without giving a reason however it is encouraged to finish all the rounds until a consensus is reached to ensure the conceptualized model is accurate, valid and effective.

### WHAT ARE THE POTENTIAL BENEFITS OF TAKING PART IN THIS STUDY?

The benefits are if the model is successfully conceptualized and schools in South Africa apply it to improve their SMS. Your role in making this model effective, valid and accurate would have added much value to schools in South Africa and you will be acknowledged for your contribution. In addition, this model may enhance the level of interoperability maturity in SMS and highlight the value of such a model for existing SMS in schools.

# ARE THERE ANY NEGATIVE CONSEQUENCES FOR ME IF I PARTICIPATE IN THE RESEARCH PROJECT?

There are no identified risks.

# WILL THE INFORMATION THAT I CONVEY TO THE RESEARCHER AND MY IDENTITY BE KEPT CONFIDENTIAL?

Your identifiable information will not be collected. Your name will not be recorded anywhere, and no one will be able to connect you to the answers you give. Your answers will be given a code number, or a pseudonym and you will be referred to in this way in the data, any publications, or other research reporting methods such as conference proceedings. Only the supervisor and I will have access to this data.

### HOW WILL THE RESEARCHER(S) PROTECT THE SECURITY OF DATA?

The researcher will store hard copies of your answers for a minimum period of five years in a locked cupboard/filing cabinet at the CSIR for future research or academic purposes; electronic information will be stored on a password-protected computer and authenticated cloud. Hard copies will be shredded, and electronic copies will be permanently deleted from the hard drive of the computer.

# WILL I RECEIVE PAYMENT OR ANY INCENTIVES FOR PARTICIPATING IN THIS STUDY?

No compensation is provided for participating nor is any cost incurred. This research does not force anyone to participate, and participation is voluntary.

### HAS THE STUDY RECEIVED ETHICS APPROVAL

This study has received written approval from the Health Research Ethics Committee of the College of Agriculture and Environmental Sciences, Unisa. A copy of the approval letter can be obtained from the researcher if you so wish.

# HOW WILL I BE INFORMED OF THE FINDINGS/RESULTS OF THE RESEARCH?

If you would like to be informed of the final research findings, please contact Keneilwe Maremi on 0128427262 or <u>KMaremi@csir.co.za</u>. The findings will be published in conferences, journals and the dissertation. If you require further information or want to contact the researcher about this study, please contact Prof Marlien Herselman, 0128413081, <u>MHerselman@csir.co.za</u>.

Should you have concerns about how the research has been conducted, you may contact Prof Marlien Herselman, 0128413081, <u>MHerselman@csir.co.za</u>.

Thank you for taking the time to read this information sheet and for participating in this study.

Thank you.

KMaremi

Keneilwe Maremi

# **APPENDIX C: Delphi round one information**

The informed consent form was accompanied by the School Management systems Interoperability Maturity (SMSIM) Model Round 1 Questionnaire

Purpose: The purpose of the study is to conceptualize a School Management Systems Interoperability Maturity (SMSIM) model for private schools in South Africa. This conceptualized SMSIM model will apply a specific Interoperability Maturity Model to assess the level of interoperability in SMS and provide clear guidance on how SMS can benefit from using interoperability maturity levels to improve their current SMS systems.

Please provide answers to the following questions based on your knowledge as a selected expert:

# **Round 1 Questionnaire**

	Round 1 Questionnaires	1-Strongly	2-Disagree	3-Neither	4-Agree	5-Strongly	Comments
		Disagree		Agree Nor Disagree		Agree	
1	School Management Systems (SMS) are large repositories			Disugree			
	that schools use to manage the day-to-day administrative						
	activities of the school						
2	Multiple users use SMS and these can be configured to meet						
	the needs of each school independently						
3	SMS help to overcome monotonous paperwork in schools, are						
	very easy to use, and can work accurately and smoothly in						
	different scenarios. SMS reduce workload, increases						
	efficiency in school management, and saves time						
4	Many schools in South Africa use a SMS called SA-SAMS						
	(South African School and Administration Management						
	System) as mandated by the South African Department of						
	Basic Education DBE)						
5	To make the administration of schools more effective, the						
	DBE developed SA-SAMS to solve not only management and						
	administration challenges but also to streamline data in						
	schools						
6	Not all schools in South Africa are using SA-SAMS, but their						
	SMS should easily integrate with SA-SAMS to reduce						
	duplicate work						
7	Although SMS offers many advantages to schools, it also come	s with many d	isadvantages s	uch as:			
7.1	Increased workloads for teachers						
7.2	Duplication of work						

	Round 1 Questionnaires	1-Strongly	2-Disagree	3-Neither	4-Agree	5-Strongly	Comments
		Disagree		Agree Nor		Agree	
				Disagree			
7.3	Redundancy and errors						
74	Delays for the submission of reports to the DBE by schools						
/							
7.5	Most schools use a free version of SMS which often lacks						
	features such as discipline, custom reports, registration, and						
	inventory. Some are narrowed by space capacity, the number						
	of learners, and storage.						
7.6	SMS are often installed on one standalone computer, which						
	limits access, which causes substantial tension and stress on						
	the administrator working alone and is the only resource						
	available during peak demand times such as the end of the						
	term when schools are required to process learner reports.						
	This mostly results in annoying delays.						
7.7	Teachers have a small amount of access to the system for						
	performing simple tasks like capturing marks and mostly						
	consume teaching time by manually validating learner's						
	marks on paper that are never-ending, increasing the chances						
	of errors.						
7.8	Non-availability of skilled ICT personnel who can utilise the						
	system. There is a lack of specific and inadequate availability						
	of relevant software, lack of maintenance culture, and lack of						
	data security and impurity on the government's side.						
7.9	SMS can only be accessed within the school's intranet.						

	Round 1 Questionnaires	1-Strongly	2-Disagree	3-Neither	4-Agree	5-Strongly	Comments
		Disagree		Agree Nor		Agree	
				Disagree			
7.10	Limit learners and parents or guardians from accessing the						
	learners' information.						
8	SMS has room for improvement due to its disadvantages						
9. Interoperability has educational benefits that can help address the above-mentioned disadvantages of SMS. As these advantages include:							
9.1	Reducing the burden on staff members of the school to enter						
	data						
9.2	Improving data quality						
9.3	Promoting adaptability and promote data sharing						
9.4	Supporting data-driven decision making						
9.5	Enhancing efficiency and timeliness						
10	These advantages of interoperability can enhance SMS to						
	ensure an improved overall information flow for effective						
	decision making and management of schools						
11	Interoperability is the platform that provides continuous						
	information exchange, for example, exchanging information						
	that other systems can use to improve the performance, create						
	services, control the processing of information systems and						
	operations						
12	There is a need to conceptualise a model that will bring						
	together interoperability in SMS						

# **Appendix D: Delphi round two information**

### SMSIM Model Round 2 Questionnaire

Purpose: The purpose of the study is to conceptualize a School Management Systems Interoperability Maturity (SMSIM) model for private schools in South Africa. This conceptualized SMSIM model will apply a specific Interoperability Maturity Model to assess the level of interoperability in SMS and provide clear guidance on how SMS can benefit from using interoperability maturity levels to improve their current SMS systems. School Management Systems Interoperability Maturity (SMSIM) model

The School Management Systems Interoperability Maturity (SMSIM) model is a feeder model that assesses the level of interoperability in the School Management Systems (SMS) and continuously provides guidelines to improve the current level of interoperability in SMS. This model consists of five phases: the foundation, primary, intermediate, and tertiary phases. SMSIM model uses the tested, verified, and approved Interoperability Maturity Models (IMM), which are a combination of Information System Interoperability Maturity Model (ISIMM) and Organisational Interoperability Maturity Model (OIM) to assess and provide guidelines for bettering and improving SMS. A feeder at each phase goes into each level of ISIMM and OIM to assess interoperability at each level and provide guidelines for the next level.


In order to explain the model, each phase (middle section of the model) will be addressed as this section is supported by both models on either side.

**Foundation phase:** At this level, a feeder is sent into Information Systems Interoperability Maturity Model (ISIMM) and Organisational Interoperability Maturity Model (OIM) manual or independent level parallelly to assess the required level of interoperability, the outcome of the level serves as a guideline for the next level: Systems are not connected in this interoperability level, and data is shared manually through mail posts. Schools are normally unprepared for interoperability to occur, they only communicate via phones among staff members and other schools, there is no interaction with other schools except through communicating via manual means, and there is a limited shared purpose. For schools to get to the adhoc level, they need to have basic data sharing and general guidelines for interoperability and some electronic means of communicating and sharing information. This can be as simple as using email to communicate with other schools and each other.

At this phase, teachers have increased workloads, there is duplication of work as marks and administrative tasks are done manually and then transferred to SMS later, there is a lot of redundance and errors because of fatigue and human error, submission of reports to the DBE are always delayed, and there are no skilled ICT personnel who can utilize the system. There is a lack of specific and inadequate availability of relevant software, maintenance culture, and data security and impurity on the government's side. According to literature and the capabilities of SMS, most schools, especially in the private sector, have passed this stage, specifically when it comes to manual communication and data sharing. Although, some of these disadvantages still exists even with School Management Systems (SMS) put in place in private schools of South Africa and the data analysis of the results of round 1 questionnaire has confirmed this, Interoperability can integrate these systems to ensure seamless information exchange, reduce the burden on staff members of the schools, improve data quality, promote adaptability, and enhance efficiency.

**Primary phase:** At this level, a feeder is sent into **ISIMM and OIM Adhoc level** parallelly to assess the required level of interoperability, the outcome of the level serves as a guideline for the next level: In this level of interoperability, The school has separate reporting lines of responsibility, general guidelines to achieve interoperability, and a shared purpose with other schools, there is

basic data sharing, applications and databases are separate, data is not shared with other schools, and data is exchanged through systems in a point-to-point manner. To get to the collaborative level, schools need to have a shared logical data model, basic collaboration with other schools, and general doctrine to support interoperability and some experience in interoperability.

Schools in this level of interoperability have improved decimally compared to the schools in the foundation phase of interoperability, however the improvement is minimal, and a lot of work has to be implemented to realise the full benefits of interoperability. It is observed that, at this level, school staff members are still burdened with entering data into the systems of the schools as databases and applications are separate and data is shared in a point-to-point manner. This opens room for errors, duplicates, redundancy and compromising the integrity of the data between the back-and-forth communication of the systems. By using a SMSIM model as a guideline model to help improve SMS, the requirements of the next level will be obtained and with slow progress, this type of school will run smoothly.

Intermediary phase: At this level, a feeder is sent into ISIMM and OIM collaborative level parallelly to assess the required level of interoperability, the outcome of the level serves as a guideline for the next level: In the collaborative level of interoperability, there is broader connections to legacy systems although minimal standard functions exist, applications and databases are still separate at this level. There is basic data collaboration with other schools, and data is still not shared, however logical data models are shared with other schools. Schools can share communications and knowledge about specific topics with other schools. General doctrine is put in place to support interoperability and some experience, reporting lines are still separate, and the school shares goals, purpose, and value system with other schools. Schools need to exchange data between independent applications to get to the integrative level. Collaboration must be at an advanced level, share communication and knowledge with other schools, and have a detailed doctrine and experience in interoperability systems.

Unfortunately, legacy systems are outdated computing software and/or hardware that are still being used. These types of systems meet the needs they were originally designed for, but they don't allow for growth. These systems present a major investment and for the most part, they cannot be abandoned. Interoperability provides means of connecting these types of legacy systems with the

new breed. With this advantage of interoperability, information is never lost and there is room for growth, although majority of the disadvantages of SMS are still not entirely solved in this phase as data is still not shared and there are still silos in the schools even though some communication between schools can occur. Schools in this phase must work on getting to the next phase to fully realise the benefits of interoperability. Many schools in South Africa are in this phase of interoperability, they know the benefit of interoperability and to some degree, measures are put in place to support interoperability, but data is still not shared, applications and databases are still separate and unfortunately there is still too much strain on the staff members of the schools to enter data into different systems.

When interoperability is applied correctly in the school, there should be seamless information exchange between the systems, by a click of a button, other departments, such as finance and administrative must be able to retrieve information without the need to reenter what was already entered by another staff member from another department. Paid SMSs can do this to some degree, Ed-admin which was mentioned by one of the experts during round 1 questionnaire of the Delphi study, has 40 different modules which can be accessed from anywhere at any time. To some degree these SMS is interoperable, it is also able to integrate with other systems for ultimate benefits however it unfortunately still works in silos with other schools and other applications not included in the system. Also, there is no evidence of such an SMS integrating with legacy systems. The SMSIM model will assess and provide guidelines to achieve the next level of interoperability.

**Secondary phase:** At this level, a feeder is sent into **ISIMM and OIM Integrative or combined level** parallelly to assess the required level of interoperability, the outcome of the level serves as a guideline for the next level: At this level of interoperability, schools need to have detailed doctrine & experience in using interoperability's systems as well as the ability to communicate, sharing knowledge, and interact with other schools. On a technical level, data is shared and exchanged between independent applications, integration of systems is implemented between organisations. For schools to get to the unified level, they need to have interoperability completely implemented in the school, both on a technical level and the human activity level. Schools need to interoperate with other school on a day-to-day basis completely. Schools that qualify to be on this level have implemented interoperability effectively, which means there is integrations of systems from one organisation to another, this is the ultimate goal with disadvantages of interoperability closed observed and addressed. Schools can communicate and seamlessly share information without compromising security and privacy. Most schools have yet to reach this stage of interoperability; hence disadvantages of SMS are still evident.

To get to the next level of interoperability, schools must be completely interoperable with other schools on a day-to-day basis. The school's data is fully distributed between organisations with a common interpretation on a technical level. Schools are continuously interoperating, data and applications are fully shared, processes are completely automated, and front and back-office systems are fully interoperable.

**Tertiary phase:** At this level, a feeder is sent into **ISIMM and OIM's Unified level** parallelly to assess the required level of interoperability, the outcome of the level serves as a guideline for the next level: At this level, schools have a shared understanding with other schools, completely interoperable with other schools on a day-to-day basis. The school's data is fully distributed between organisations with a common interpretation on a technical level. Schools are continuously interoperating, data and applications are fully shared, processes are completely automated, and front and back-office systems are fully interoperable. This is the last level of interoperability, and schools would have reached the ultimate level of interoperability by getting to this level. This level is normally easy to achieve for schools of the same kind.

The section below presents the round 2 questionnaires that were used to assess the model's utility, validity, effectiveness, and quality.

## **Round 2 Questionnaire**

	Round 2 Questionnaire	1-Strongly	2-	3-Neither	4-Agree	5-Strongly	Comments
		Disagree	Disagree	Agree Nor		Agree	
				Disagree			
1	There are no doctrines put in place						
	that enable SMS to cooperate with						
	other SMS						
2	Schools cannot use SMS to share						
	information and communicate,						
	important data into SA-SAMS is						
	done manually.						
3	Schools do not use SMS to share						
	information and communicate with						
	other schools.						
4	In most schools, SMS is installed						
	on a standalone computer with						
	limited access.						
5	All the other systems and						
	applications that exist in schools are						
	most probably not connected to						
	SMS						
6	The systems in the schools are most						
	probably unable to share data &						
	communicate with each other						
7	Questions about the Model	•	•	-	•	-	•

	Round 2 Questionnaire	1-Strongly	2-	3-Neither	4-Agree	5-Strongly	Comments
		Disagree	Disagree	Agree Nor		Agree	
				Disagree			
7.1	With the current state of SMS, the						
	SMSIM model assess the level of						
	interoperability and shows						
	guidelines for achieving the next						
	level with the ultimate aim of						
	addressing the disadvantages of						
	SMS						
7.2	Is the SMSIM model effective and						
	accurately able to assess the level of						
	interoperability in SMS and provide						
	guidelines on how to get to the next						
	level?						
7.3	Is the SMSIM model a valid model						
	for providing guidelines to address						
	the current SMS problems?						
7.4	SMS will be improved when						
	utilising this model						
1		1			1		

In your view, how can the model be improved?